SPORT CONSUMPTION PATTERNS IN THE EASTERN CAPE: CRICKET SPECTATORS AS SPORTING UNIVORES OR OMNIVORES

A thesis submitted for the fulfilment of the requirements for the degree of

MASTER OF COMMERCE (ECONOMICS)

Department of Economics and Economic History

Rhodes University

Grahamstown

KELCEY BROCK ¹ 608B1506

Supervisors: Prof Gavin Fraser and Mr Ferdi Botha

December 2014

¹ The financial assistance from the ALLAN GRAY Scholarship as part of the Rhodes University Prestigious Scholarships is hereby acknowledged. Research funding from the Rhodes University Research Committee used to fund elements of the research is also acknowledged. The opinions, conclusions and recommendations arrived at, are those of the author and are not attributed to Rhodes University or the respective funders.

DECLARATION

Excepting for the references specifically acknowledged in the text and sources of assistance, insight or information explicitly acknowledged, this thesis is entirely my own work. This thesis has not been submitted to any other University, Technikon or College for degree purposes.

KELCEY BROCK

.....

ABSTRACT

Since its inception, consumption behaviour theory has developed to account for the important social aspect that underpins or at least to some extent can be used to explain consumption behaviour. Modern consumption behaviour theory is anthropocentric in nature, with people and societal influence at the forefront of the theory. To date, empirical studies on consumption behaviour of cultural activities (for example, music and arts), entertainment and sport have used Bourdieu's (1984) omnivore/univore theory to suggest that consumption of leisure activities is bound up in social ties. To date, no such investigation has been conducted in the context of sport in South Africa. The aim of the study therefore is to investigate whether South African cricket spectators are sporting omnivores or univores, thus, essentially investigating whether sports consumption behaviour in South Africa is bound up in social ties.

A number of positive economic and social ramifications could result from gaining a holistic understanding of sports consumption behaviour in South Africa. Given these ramifications, the secondary goal of the research is to identify motives for consumers making specific sport consumption decisions, and determining whether certain characteristics can be attributed to these consumption decisions. Recommendations based on the findings of the research could help various stakeholders understand sports consumption patterns in South Africa, which could in turn lead to the realization of positive economic and social benefits.

The study made use of a questionnaire, administered at four different limited overs cricket matches in the 2012/13 cricket season, to obtain a range of responses reflecting specific types of consumption behaviour as well as motives for consumption decisions of cricket spectators in the Eastern Cape. Using individual binary probit models and post estimation F-tests, the results indicate that consumption behaviour of sport within South Africa predominantly differs on the grounds of education and race. This suggests that there are aspects of social connotations underpinning sports consumption behaviour within South Africa.

ACKNOWLEDGEMENTS

I would like to express my appreciation to everyone who helped, guided, supported and encouraged me throughout my Masters. I owe it all to you. To my supervisor, Professor Gavin Fraser – thank you for your support and guidance throughout this journey. Every meeting and discussion has been valuable and I have learnt so much from you. Thank you for the constructive feedback and the time and effort put into my research. It is appreciated. To my co-supervisor, Mr. Ferdi Botha, thank you for your assistance throughout my research and particularly for your help with the econometrics of the research – your time, effort and feedback is appreciated. A special thank you to Leigh Deyzel and Dave Emslie from the Chevrolet Warriors cricket franchise for allowing me to conduct the research and for providing tickets for the matches for the data collection – your help and interest in my research is greatly appreciated.

To my colleagues in the Department of Economics, thank you for your support. The Friday tea chats and corridor advice has taught me many a valuable lesson. A special thank you to Mrs. Tiana Muller for the constant messages of support, cups of tea and laughter – it has helped make this journey a fun one. To my research assistants, thank you for the time and energy you dedicated to helping me with the data collection. I am so grateful for your assistance, especially given that you had academic pressure of your own – I could not have done this without you.

Finally, to my mom and dad - Marcelle and Billy, brothers - Ryan and Sean, Fiancé Byron, Gran and friends, thank you for your unfailing love and support. You have been my pillar of strength throughout my life, and for this thesis, no different. I am forever grateful!

TABLE OF CONTENTS

	Decl	aration	i	
	Abstract			
	Ack	Acknowledgements		
Table of contents			iv	
	List	List of tables and figures		
	CHA	APTER 1: INTRODUCTION	1	
	1.1	Context of the research	1	
	1.2	Goals of the research	5	
	1.3	Methods, procedures and techniques	6	
	1.4	Ethical considerations	7	
	1.5	Organisation of the study	7	
	CHA	APTER 2: THEORETICAL FRAMEWORK	8	
	2.1	Introduction	8	
	2.2	General consumption theory	8	
	2.3	Consumption theory for intangible goods: social status and leisure consumption	11	
	2.4	Sport as a form of leisure consumption	12	
	2.4.1	The determinants of sports consumption	12	
	2.5	Summary and conclusion	18	
	CHA	APTER 3: LITERATURE REVIEW OF EMPIRICAL STUDIES	19	
	3.1	Introduction	19	
	3.2	The omnivore/univore theory	19	
	3.3	Empirical studies investigating the omnivore/univore theory	21	
	3.4	Developments of the omnivore/univore theory: Binary to multi-dimensional form	23	
	3.5	Developments of the omnivore/univore theory beyond the cultural context	27	
	3.6	Empirical studies investigating the omnivore/univore theory in Sport	29	
	3.7	Motives for attendance: an overview of the entertainment and sports industry	38	
	3.8	Summary and conclusion	45	

CHAPTER 4: METHODS AND DATA

4.1	Introduction	46		
4.2	Defining sporting omnivores and univores	46		
4.3	Method	53		
4.3.1	Model specification	53		
4.3.2	Variable description	53		
4.4	Data	54		
4.4.1	Overview of data and questionnaire composition	54		
4.4.2	Sampling and administration	55		
4.5	Summary statistics	57		
4.5.1	Demographic statistics	57		
4.5.2	Question responses	58		
4.6	Summary and conclusion	60		
СНА	PTER 5: RESULTS, DISSCUSSION AND RECOMMENDATIONS	62		
5.1	Introduction	62		
5.2	Overview of the results for the omnivore/univore classification	63		
5.3	Probit results predicting the probability of an omnivore classification	64		
5.3.1	Probit results for the conventional definitions	65		
5.3.2	Probit results for the alternative definitions 1 to 4	70		
5.3.3	A comparative analysis of socio-demographic findings across definitions	79		
5.4	Motivation for attendance results: A comparison between classifications	82		
5.5	Summary and Conclusion	86		
СНА	PTER 6: SUMMARY AND CONCLUSIONS	87		
6.1	Introduction	87		
6.2	Executive summary of the research	87		
6.3	Conclusions of the research	91		
6.4 L	imitations of the study and suggestions for future research	92		
REF	ERENCES	94		
APP	ENDIX A	106		
APPENDIX B				
APP		111		
		v		

46

LIST OF TABLES

Table 4.1:	Summary of alternate univore/omnivore classifications	52
Table 4.2:	Summary statistics	58
Table 4.3:	Motives for attendance (%)	59
Table 4.4:	Omnivore/Univore classification (%)	60
Table 5.1:	Omnivore/univore classifications	63
Table 5.2:	Probit results predicting the probability of a univore classification under the conventional definitions	65
Table 5.3:	Probit results predicting the probability of a univore classification under the alternate definitions	69
Table 5.4:	Motivation for attendance results: A comparison between classifications	83
Table C1:	Post-estimation F-test results showing significance between explanatory variables	111

LIST OF FIGURES

Figure B1:	Model of Spectator Sport Consumption	110
------------	--------------------------------------	-----

CHAPTER 1: INTRODUCTION

1.1 Context of the research

Consumption behaviour is an aspect of modern economic theory that has developed and grown in importance, and is of particular interest to producers, manufacturers, and marketers. Consumption behaviour is generally understood as the process in which individuals or groups of individuals acquire, use, and dispose of a product, service, idea, or experience (Batra and Kazmi, 2008). It involves the study of an entire consumption decision process, from gathering the information to the actual purchase stage and involves understanding individuals' feelings, thoughts, and actions (Kumar, 2008). The study of consumption behaviour is based on a number of fundamental economic assumptions. Individuals are assumed to be rational beings that aim to maximize their utility, and who are able to anticipate the consequences of their individual choices (Becker and Murphey, 1988). Consumers will thus use the resources at their disposal to attain commodities that satisfy their optimal level of utility. Any purchase and consumption of a commodity is assumed to provide some satisfaction or utility and the principle of diminishing marginal utility is assumed (Deaton and Muellbauer, 1980). Based on the assumptions, it is possible to predict probable consumption behaviour, allowing economic decisions based on general consumption patterns. Traditional consumption behaviour theory generally considered the impact of changes in income and prices on consumption decisions. Successive consumption behaviour theory builds on this theory and considers other factors that may influence consumer behaviour (Lancaster, 1966). Traditionally, consumption behaviour focuses largely on consumer decisions of tangible commodities; however, because consumption behaviour of intangible commodities also contributes significantly to a country's Gross Domestic Product, examination of such behaviour is also needed (Rao, 2012). As described by Blundell (1988), the proximity of interplay between empirical and theoretical considerations of consumption behaviour, and the rapid growth in availability of new data, makes the analysis of consumption behaviour an attractive area of research. Bray (2008) supports the idea that consumption behaviour, since its inception, has become a sustained area of research in economics.

According to Kumar (2008), traditional consumption behaviour considered the consumption of economic goods and services. Although not a traditional perspective, art (Snowball *et al.*, 2009) and sport (Glencoe, 2013) are seen as commodities, specifically classified as intangible

commodities that can be consumed. Much like the consumption of economic goods and services, watching sport, whether in physical attendance or on television, is a form of consumption, where sport is the consumable intangible commodity (Glencoe, 2013). In the market for economic goods and services, many motives and factors (such as economic determinants, human characteristics, and of particular relevance to this research, social influences) have been identified that underpin the purchasing decisions made by consumers (Shah, 2010). For producers and marketers to make effective business decisions it is vital that they base their decisions on the consumers' motives for particular consumption behaviour (Laesso et al., 2013). In researching consumers' decision-making motives, it becomes possible to identify consumption behaviour that is associated with a specific type of consumer. In this way, producers and marketers are able to identify and target a particular market in order to maximise their revenue. Over and above identifying motivating factors for positive consumption behaviour, empirical studies in the arts for example (which include amongst others, Snowball et al., 2009, Peterson, 1992 and Peterson, 2005) have noted the usefulness of stratifying consumers according to their consumption behaviour patterns, classifying them as either univorous or omnivorous consumers. Such classifications have been used to determine the extent to which consumption is bound up in social ties.

Consumption behaviour studies undertaken in the South African context specifically focused on the consumption patterns of the arts. In this context, univores were defined as patrons with narrower cultural tastes (attending only one modern genre or two traditional genres of art) while omnivorous patrons were those that consumed a wider array (more than two types of art genres) of both popular and high cultural goods (Snowball et al., 2009). The main objective of the study was to identify the determining indicators that differentiate univorous from omnivorous cultural consumers. The information obtained from categorizing consumers of the arts into omnivorous and univorous patrons respectively, allows marketers and producers of the various forms of the arts to base future decisions on the specific classification of their respective audiences (Snowball et al., 2009). To date, however, consumption behaviour studies are notably limited in the context of South African sport (Gerber and Terblanche, 2012). More specifically, there appears to be no empirical research on the classification of consumers as univores or omnivores in the context of sports consumption in South Africa. Snowball et al. (2009) argued that studies considering omnivore/univore status are particularly valuable in the case of South Africa given the potential social tolerance ramifications that could stem from such a classification (as was highlighted by Bryson (1996), Coulangeon (2005), Tampubolon (2008) and Van Eijck and Leivens (2008)). This therefore highlights the need for research on an omnivore/univore classification of sports consumption patterns in South Africa.

A number of studies researching consumption behaviour have been conducted outside South Africa. These include comparative research involving omnivorousness (Peterson, 2005), audience segmentation and omnivore/univore classifications in cultures (Peterson, 1992), theories of consumer behaviour (Trail and James, 2010), contemporary perspectives on consumer motives, goals and desires (Ratneshwar *et al.*, 2000) and consumer behaviour in retrospect and prospect (Zaichkowsky, 1991). Similarly, many studies on consumer behaviour with specific application to sport have been conducted internationally. These include, for instance, predictors of attendance at major sporting events (Hall *et al.*, 2009), theoretical models of sport spectator consumption behaviour (Trail *et al.*, 2000; Trail and James, 2010), and a review of sport motivation constructs (Milne and McDonald, 1999).

While many international empirical analyses have explained consumption behaviour in sport, Gerber and Terblanche (2012) have conducted the only such study in the South African sport context. This study considered team performance and sport attendance at Currie Cup and Super 14 rugby, a context, which, according to Asher (2011), has not suffered a decline in match attendance and consumer interest to the same extent as cricket. South African domestic and national cricket match attendance statistics show that there has been a general decline in attendance in all match formats over the past few decades (ESPN, 2013). Former South African limited-over and test cricket player, Fanie De Villiers, argued that support for limited-over cricket is "diabolical" yet states that this could not be attributed to the quality of cricket played (Asher, 2011). The decline in match attendance and whether the disparity between the supply and demand for cricket in South Africa can be attributed to factors of the game (for instance, the presence of certain players, or the quality of cricket played) or consumer influences (for instance, social factors or omnivorous or univorous consumer characteristics).

According to Brember (2009), sport is highly regarded among South Africans, both in terms of active participation and in a passive role as a spectator. More specifically, South Africans place a great deal of recognition on cricket as both an amateur and professional sport (Louw, 2010). This study will therefore focus on the application of consumption behaviour theory in cricket. With South African cricket forming a significant aspect of international cricket

heritage (Morgan, 2013), it becomes important to examine factors that would increase the likelihood of the long-term survival of the game. This could be achieved by investigating the extent to which sporting consumption patterns in South Africa are bound up in social hierarchies by classifying consumers as either sporting univores or omnivores, in conjunction with considering other possible motives for attendance. Sporting univores could be defined as spectators who choose to attend fixtures of a specific sport/s to the exclusion of others (or have a very narrow consumption demand for sport), while sporting omnivores choose to attend a wide range of sporting events (or are general sports followers).

The study attempts to highlight the importance of classifying a market to facilitate effective marketing, such that the correct target market is identified and marketers are better equipped to direct marketing at the factors that motivate specific consumption behaviour. Sport has also been recognised as an important source of revenue and economic benefit, not only for the individual participants, sporting venues and associations but also for cities, regions and a country as a whole (Hall et al., 2009). More specifically, Hall et al. (2009) noted that, owing to a positive correlation between match attendance and revenue, it becomes necessary to determine the internal and external factors that affect attendance with the aim of improving aspects of the sport to increase match attendance and subsequently revenue too. With revenue having such significant and widespread economic importance, such a study will facilitate effective customised marketing to encourage attendance and in so doing facilitate the longterm viability of cricket in South Africa and sustainable sports related revenue for the country. Furthermore, Hall et al. (2009) found that high attendance at sports fixtures in the past were positively correlated to an increase in sponsor revenue. It is therefore important to identify factors associated with positive consumption behaviour that could be used to encourage match attendance.

According to empirical results found by Szymanski and Kuypers (1999) and Gilmore *et al.* (2011), sport forms part of an important industry that significantly affects GDP and, as a result, it is important to ensure that the sports industry be nurtured and maintained. Eschenfelder and Li (2007) supported this and noted that sport is a critical centre of the United States economy. They argued that, for 2005, the sports industry was greater than the automobile and agricultural industry as well as greater than the sum of all the public utilities (Eschenfelder and Li, 2007). In 2005, calculated estimates revealed that the Gross National Sports Product or GNSP (an economic measure used to determine the relative size of the sports industry) was \$168.469 billion (Milano and Chelladurai, 2011). As seen in the United

States, the sports industry poses as a fundamental contributor of the economy's revenue and, as a result, it becomes necessary that a sports industry be developed and sustained to ensure maximum possible revenue for the economy. In the case of South Africa, the sports industry has been noted as a "multibillion rand industry" and contributed to 2 percent of the country's GDP (South African Web, 2013). Statistics South Africa (2012) also highlights the importance of the South African sports industry in generating significant tourism revenue. Thus, aiding the growth of South Africa's GNSP (through the positive identification of factors that are correlated to match attendance) would likely facilitate South African GDP growth. Identifying what motivating factors are attributed to positive match attendance would aid in determining whether spectator consumption behaviour can be specifically classified to enhance marketing strategies and encourage specific consumption behaviour. A final argument grounding the need to increase or sustain a country's GNSP is that generally the global sports industry has a large number of existing employees and has annually been accountable for the creation of a significant number of jobs (Eschenfelder and Li, 2007).

Further importance of this study pertains to one of the goals outlined by the Department of Sport and Recreation and the limited funds at their disposal. The Department aims to "maximise access, development and excellence at all levels of participation in sport and recreation in order to improve social cohesion, nation building and the quality of life of all South Africans" (SRSA, 2012:1). Because the Department is subject to limited funds (SRSA, 2012), the information generated from this study could assist in providing guidance as to where the limited funds would most optimally be utilised.

1.2 Goals of the research

The goals of the study are:

- (1) To investigate whether South African cricket spectators are sporting omnivores or univores.
- (2) To identify motives for consumers making specific sport consumption decisions, and whether certain characteristics can be attributed to these consumption decisions.

1.3 Methods, procedures and techniques

With reference to the definition of consumption behaviour, this study focuses on the consumption behaviour of an experience, in this case, sport. Eschenfelder and Li (2007) define sport spectatorship/sport viewership as encompassing the payment for admission to a sports event, subscribing to televised sports programming as well as the purchase of sports related published materials (such as books and magazines). Within sports spectatorship, this study will focus on the payment for admission to attend a sports match where the fixture and the players are seen as saleable commodities (Stander, 2009). The study focuses solely on a team sport. As highlighted by Simmons (2006), the reason for this choice is owing to the fact that literature on attendance for individual-based sport has been found to be negligible. Furthermore, the study focuses solely on the domestic limited-over cricket played by the Chevrolet Warriors at *Axxess DSL St Georges* in Port Elizabeth.

The research follows a post-positivist research approach. Questionnaires comprised of 25 closed-ended questions, five open-ended questions and five demographic questions were developed and administered. The surveys were based largely on the work of Trail and James (2001) and the Center for Sport Consumer Research based in the United States. A combination of the already existing survey formats was used as a guideline for the composition of a questionnaire specifically tailored for the South African cricket context. The surveys were comprised of a number of questions that can be linked to motives for certain consumption decisions and allow for the classification of consumers as sporting omnivores or univores. The questionnaires were also comprised of repeated questions to eradicate any information bias. Questionnaires took approximately ten minutes to complete and were administered to roughly 500 participants during selected fixtures over the duration of a oneday cricket competition. The questionnaires were administered in an interview manner, with the response form for the close-ended questions in a Likert Scale format. Participants were asked to advise the interviewer as to which answer best describes them, with options ranging from 'strongly agree' to 'strongly disagree'. Open-ended questions were used in conjunction with the close-ended questions to generate responses that will inform the research findings. The participants were selected through purposive sampling (Oliver, 2006) from the spectators attending the fixture with the intention of obtaining a relevant representative spectator sample. Data collection was carried out at fixtures played at the same venue to control for any geographical biases. Upon completion of the data collection, the question responses were captured and coded. Individual probit models were estimated for each of the classifications to examine the predictors of being a sporting omnivore or univore. Finally, the study made use of bivariate cross-tabulations to investigate how the factors that are identified as motives for attendance differ amongst omnivores and univores for a given definition.

1.4 Ethical considerations

The questionnaire was designed in such a way to ensure beneficence and non-maleficence of all participants involved. Participants were made fully aware that their participation in the study was completely voluntary and that they could withdraw from the study at any point. Participants were notified of the purpose of the research and what was required of them prior to the commencement of the questionnaire. Participants were informed that the research is sanctioned by the Chevrolet Warriors cricket franchise (represented by CEO Dave Emslie and marketing and commercial manager Leigh Deyzel). All responses were confidential and the data collected were kept in safe housing upon completion of the study. The data were used for research purposes only and will not be made available for alternate research. Finally, the participants were made aware that they may have access to the overall results of the study (although not to the individual responses), if they so wish, and were informed on how they could go about accessing these results. The questionnaire and research had been approved by the Rhodes University Department of Economics and Economic History's research ethics committee and by the Rhodes University Commerce Higher Degrees Committee.

1.5 Organisation of the study

The study is organised as follows: Chapter 2 presents the theoretical framework upon which the research is based. This chapter includes a general review of consumption theory and then considers the consumption theory for intangible goods. The remainder of the chapter considers the consumption theory underpinning sport as a form of leisure consumption. Chapter 3 considers a review of empirical studies to date. Chapter 4 provides an overview of the methods and data that were employed in this study. Chapter 5 presents the results of the study, a discussion thereof, as well as the recommendations stemming therefrom, while Chapter 6 summarises and concludes the study.

CHAPTER 2: THEORETICAL FRAMEWORK

2.1 Introduction

Although the focus of this study is to investigate whether South African cricket spectators are sporting omnivores or univores, understanding general consumption behaviour is important for a number of purposes. As highlighted in Chapter 1, a number of potential economic and social benefits could result from understanding consumption behaviour. To understand the motives underpinning consumption behaviour, it is important to consider theoretical insights as well as empirical studies that have been completed on the subject. This chapter begins with a review of the theoretical framework that has been used to examine consumption behaviour, in general (Section 2.2), and then focuses particularly on the theoretical underpinning of consumption behaviour of intangible goods, particularly leisure (Section 2.3). Section 2.4 shifts the focus of the review to sport as a form of leisure consumption and considers the determinants of sports consumption. Section 2.5 summarises and concludes the chapter.

2.2 General consumption theory

According to Marx and van der Walt (1989), the theoretical interest in consumption behaviour studies originated from addressing the problem of aligning mass production to mass consumption. Evidence of theoretical insights that were offered to understand consumption behaviour dates back to 1965 (Andeason, 1965). It has been argued that since its inception, "many writers have focused their efforts on this interesting and important subject" (Marx and van der Walt, 1989:72).

Numerous theoretical factors have been identified to underpin and influence consumers' decisions and consumption behaviour. Understanding such factors is argued to be crucial to explaining and predicting human behaviour, particularly of value in a consumer-orientated market (Marx and van der Walt, 1989). Traditionally, consumption behaviour was theoretically understood by prepositions pertaining to rationality and the predominant role that income plays in determining expenditures (Katona, 1968). Katona (1968) summarises the traditional understanding of consumption behaviour into two core ideas: firstly, that consumers will choose the best alternative from the conceivable actions that are available to them and, secondly, that, apart from tastes, the primary determinant of consumers'

expenditure is absolute and relative income (therefore includes prices as a determining factor). Louden and Della Bitta (1993) supported this notion and noted that consumption behaviour was traditionally approached from a purely economic perspective, where the sole focus was on the act of purchasing goods and services. Bray (2008) argued that the most prevalent model from the traditional approach to consumption behaviour was embedded in utility theory. Utility theory assumed that consumers were rational beings and made choices that were based on the anticipated outcome of their consumption decision (Bray, 2008). Schiffman and Kanuk (2007) explained that the assumption of the consumer as a rational being presupposes that the consumer is informed of all consumption possibilities, is able to order the options that are available to him/her and from the ordered alternatives, is able to make a decision as to the optimum consumption option. These assumptions in turn form the premise of the rational model of consumption behaviour (Lunenburg, 2010).

Under the rational model of consumer behaviour, Towler (2010) states that consumers were well informed of all consumption alternatives as well as the outcomes that would emanate from these alternatives. Consumers are also aware of their decision criteria and are able to make and implement the best possible consumption choice (Towler, 2010). Furthermore, the traditional approach to consumer behaviour is based on the theoretical utility maximising rule. Williams (2013) noted that, theoretically, a rational consumer will not only rank consumption options based on his or her preferences, but will also choose the goods/services that yield the highest marginal utility per unit of expenditure until their limited budget is exhausted. Simon (1997), however, argued that adopting this traditional approach to consumer behaviour is problematic and that the assumptions thereof do not give an accurate account of consumer behaviour. Unlike the assumptions that underpin the rational model and traditional approach to understanding consumer behaviour, Simon (1997) argued that, although theoretically feasible, in reality consumers often do not have adequate information to make optimum consumption decisions. He further criticized the assumption of rationality as being very limiting in its application to the real world. He argued that consumption decisions are often shaped by less rational influences such as an individual's social relationships and intrinsic values. As a result of the limitations of the application of the rational decision-making model, consumption theory has since developed to provide a more accurate account as an explanation of consumers' decisions. The development of consumer theory was highlighted by Katona (1968), who acknowledged that the understanding of consumption behaviour has developed from a traditional perspective to a relatively more comprehensive perspective (in which the inclusion of a wider range of facets that underpin consumption behaviour is considered). On a similar note, Sahney (2013:10) described contemporary consumer behaviour theory as "dialectical", emphasizing the broad, complex and encompassing scope of the developed theory of consumer behaviour. This is echoed by Blackwell *et al.* (2001), who noted that the theory of consumer behaviour is evolutionary and has grown to encompass a holistic range of factors that influence a consumer's decision-making.

Simon (1997) proposed that an alternative to the rational model of consumption behaviour was the bounded rationality model. This model proposed that consumers are unlikely to have all the information that they require to make an optimum consumption decision. Simon (1997) argued that because of this, consumers are likely to make satisfactory as opposed to optimal decisions. As noted in his critique of the rational model, Simon (1997) placed significant emphasis on the importance of social factors underpinning consumption behaviour. This emphasis is similar to a relatively more modern and inclusive understanding of the factors that underpin consumption behaviour, as explained in a model of consumer behaviour by Marx and van der Walt (1989). This model is based on the notion that an individual's consumption is not only underpinned by the traditional economic factors but also by a complexity of individual factors (which include, amongst others, motivation, attitude, perception, personality and lifestyle) (Tanner and Raymond, 2012) and group factors (such as the consumer's family, social class, cultural group, reference group and opinion leaders) (Bearden and Etzel, 1982). Frank (1999) agreed and argued that consumption, whether intentional or not, is largely influenced by social factors, particularly an individual's social reference groups. Drawing on the importance of the social influence on consumption behaviour, Samuelson (2004) argued that consumption decisions have become a matter of relative consumption effects. That is, individual consumers are less concerned with their individual utility maximization but rather measure their utility in relation to that of other individuals (Samuelson, 2004). This idea is closely aligned with the proposition by Basmann et al. (1988) that two distinguishable types of utility stem from consumption: the first type of utility is derived directly because of the consumption of a commodity, while the second type of utility is derived from being associated with the consumption of a particular commodity. The latter form of utility reiterates the influence that social factors play in influencing consumption behaviour (Basmann et al., 1988).

2.3 Consumption theory of intangible goods: Social status and leisure consumption

Linked to the idea of the social underpinning of consumption behaviour is the significant role that social status plays in influencing and explaining consumption behaviour. According to Scheetz *et al.* (2014), status consumption pertains to the idea that consumption is a means of projecting one's wealth. This idea of social consumption originated from Thorstein Veblen (1899). Veblen (1899) considered leisure consumption, and argued that social stratification, as opposed to utility, often underpinned consumption. The first is conspicuous consumption, which pertained to the use of financial resources to obtain commodities for consumption for the purpose of portraying wealth; the second (which is of particular relevance to this research) is conspicuous leisure, which pertained to the period of time that an individual devotes to leisure activities that grant them a higher social status (Veblen, 1899).

Veblen (1899) suggested that consumption (and the division of labour) were variants of societal behaviour and generally in accordance with one's social status. His theory proposed that "conspicuous abstention from labour therefore becomes the conventional mark of superior pecuniary achievement and the conventional index of reputability" (Veblen, 1994:25). Thus, individuals of higher social status were not only likely to consume specific commodities/activities to portray wealth but were also likely to devote relatively more time to leisure activities, when compared to those of lower social status.

Since Veblen's (1899) suggestion of leisure consumption being largely determined by ones social status, many modern theorists have investigated the determinants of leisure consumption (Scheetz *et al.*, 2014). Many investigations (which include amongst others, Abel, 1990; Constantinides, 1990; Campbell and Cochrane, 1999; Messinis, 1999; Carrasco *et al.* 2005) argued that leisure consumption is likely to be determined by the habitual nature of the individual. Individuals derive utility from consistency in consumption and therefore their consumption decisions are largely dependent on their past consumption choices. Other investigations (for example, Bernheim, 1994; Corneo and Jeanne, 1997; Hogg and Vaughan, 2002; Nir, 2004) argued that the individual consumption decisions are further embedded in their larger social contexts. This argument suggested that the consumption of leisure activities was dependent on other individuals' consumption decisions. As noted by Corneo and Jeanne (1997), consumption is socially embedded in the sense that often individuals avoid the consumption of popular commodities/activities as a means of differentiating their

consumption behaviour in order to gain higher social status. Thus, it is argued that social comparison plays a significant role as a determining factor of leisure consumption.

2.4 Sport as a form of leisure consumption

According to Best (2010), leisure consumption plays a fundamental role in the broader social system. Best (2010:32) argues that leisure consumption helps "to bridge the gap between the individual and the wider social system". In terms of sport, Trail and James (2001:108) noted that "(s)port spectating also represents a predominant form of leisure behavior in contemporary society". More specifically, Best (2010) argued that sport, as a form of leisure consumption, plays a significant role in the promotion of social integration. In order to promote social integration, it therefore becomes important to understand the determinants of sports consumption in order to promote and sustain the consumption of sport.

2.4.1 The determinants of sports consumption

As noted by Cunningham and Kwon (2003:127), "sport marketers have long sought to better understand the factors that influence people to attend sporting events". According to Funk (2008), sports consumption can broadly be understood as the process in which consumers select, purchase, make use of, and dispose of sport-related commodities to satisfy their needs and derive a form of utility. This process is said to often (but not always) lead to the acquisition of a good or service (Economic Glossary, 2012). The focus of this study, and, therefore, the stance adopted in this review of literature, considers only instances where the good/service was actually acquired; where actual event attendance and not only ticket purchases are classified as consumption of the sport. This specification was included as ticket sales may not be a true reflection of the attendance and thus interviewing ticket purchasers may skew the data. For example, a company may block book tickets for their employees (under the assumption that each employee should be allocated a ticket to watch the game); however, only a fraction of those employees might choose to attend the game and as a result, interviewing the ticket purchaser or the ticket holders prior to the match would not be a true reflection of actual sport consumption. This review will thus focus solely on consumption behaviour theory pertaining to the demand for watching a sport (as opposed to ticket demand) as well as motives that govern the decisions individual consumers and groups of consumers make.

Understanding sports consumption has been sighted as seemingly complex in nature (Westbrook and Oliver, 1991), owing to the broad spectrum of factors that influence sports consumers' behaviour (Hansen and Gauthier, 1989; Simmons, 2006; Funk, 2008). In his study on the demand for spectator sports, Simmons (2006) noted that consumption behaviour of spectators at sporting events is largely influenced by the following determinants:

- the price of attending the event (including costs associated with traveling to the event),
- the real income of the spectators,
- the prices of substitute goods,
- the size of the market (which is usually proxied by the local population),
- the relative importance of the match in terms of competition outcomes, and
- uncertainty of outcome or competitive balance.

The identification of these factors is closely aligned to the work of Feehan (2006), who noted that understanding consumption behaviour is fundamental in understanding the demand-side of the market. Furthermore, he argued that the neoclassical theory of demand is the fundamental basis of analysis irrespective of the sport (Feehan, 2006). As such, the demand for sports attendance was specified as a function of the ticket price, income (included to determine whether sports attendance is a normal or inferior good), the prices of complements and substitute goods, and tastes/preferences for a team or a sport (Feehan, 2006). Pawlowski and Breuer (2011) later supported Simmons (2006) and Feehan (2006) but added that theoretically sports consumption is determined by *both* demand and supply-based opportunities.

Ticket pricing has been identified by a number of researchers (for example, Borland, 1987; Whitney, 1988; Simmons, 1996; Garcia and Rodriguez, 2002) as an important factor that determines the consumption behaviour of sport. According to Borland and MacDonald (2003), the majority of empirical findings show that there is a significant negative relationship between the ticket price and the demand for sport consumption. In investigating this relationship, Simmons (2006) argued that ticket pricing is indeed a major issue influencing demand that needs to be carefully managed. Simmons (2006) noted that if spectators are subject to higher ticket prices at 'big' games and lower ticket prices at less attractive games then spectators may be disinclined to attend either the big or less attractive games.

Simmons (2006:82) noted that attendance demand would be affected by the price of the ticket as well as the traveling costs associated with attending the event. Feehan (2006), however, argued that for an attendance demand function to be properly specified it is necessary to account for more than just the travel costs associated with match attendance. The direct and indirect travel costs should be taken into account (for example, the cost of fuel as well as the opportunity cost of time) and the costs of participation-related goods (such as refreshments or merchandise purchased at the event). Forest et al. (2002), however, noted that making comparisons in terms of ticket prices and demand is problematic as the majority of previous studies have failed to account for travel costs and, as such, previous estimations are considered biased. A closely related topic is price elasticity of demand. According to Fort (2000), inelastic ticket pricing for sporting events has been a recurring finding for almost thirty years. The motives underpinning this finding are a highly contested topic in sports economics debates. Feehan (2006) found that price-elasticity estimates may be biased downwards if capacity-constrained fixtures occur often. A capacity-constraint, he argued, will result in the observed demand curve lying within the effective demand curve and a relatively steeper observed demand curve will be prevalent (Feehan, 2006). In opposition to Forest et al.'s (2002) finding on biased estimations, Feehan (2006) highlighted an alternate view in which highly inelastic estimates of ticket price elasticity were argued to be unbiased. Marburger (1997) and Fort (2000) put forward this argument, stating that if a team's revenue function takes into account complementary sales and television revenue as opposed to just ticket revenue, then "optimising behaviour may correspond to a ticket price on the inelastic portion of demand" (Feehan, 2006:95). In a study of English football by Forest et al. (2002), ticket price elasticity estimates were found to be much closer to unity relative to previous estimates. This, they argued, is because of previous estimations corresponding to a time where complementary sales and television revenue were not important.

The second major determining factor of demand for sport, as identified by Simmons (2006) and Feehan (2006), is the income of the spectators. As noted by Feehan (2006), the inclusion of income in the function for demand is particularly relevant in determining whether a good is a normal or inferior good. Empirical evidence on the classification of sport as a normal or inferior good is not unanimous. Cairns (1990) investigated the nature of sports based on the findings of 14 different studies and concluded that there was evidence that basketball and Australian rules football were normal goods, while evidence suggested that hockey was an inferior good. The results pertaining to baseball and soccer in this study were mixed (Cairns, 1990). According to Eschenfelder and Li (2007), most sports activities are considered normal

goods, with the demand for sport increasing as income increases. They further noted that a study by Noll (1974) suggested that baseball is an inferior good (which partly contradicts the finding by Cairns (1990)) and a study by Walsh (1986) similarly suggests that hiking and backpacking are inferior goods. Owing to the lack of unanimity of empirical findings, it is therefore not surprising that "(i)nvestigators often cannot find any significant impact of income on attendance" (Cairns, 1990:10) and that many economists do not support the assertions of various sports as normal or inferior goods owing to the fact that "evidence is not clear-cut" (Siegfried and Eisenberg, 1980:41).

The third factor identified as a determinant of demand is the price of substitute and complementary goods. According to Buraimo (2006), televised sports may have either a complementary or substitute effect concerning match attendance. In some instances, fans have been found to prefer watching a televised version of a fixture, in which case television is seen as a substitute good for live attendance. Conversely, televising sport is seen as a complementary good to match attendance based on the argument that exposure on television is similar to that of advertising (Buraimo, 2006). Studies investigating the relationship between television and match attendance have been conducted for many sports, which include amongst others, college football (Kaempfer and Pacey, 1986; Fizel and Bennett, 1989), English soccer (Baimbridge *et al.*, 1996), and English rugby league (Carmichael *et al.*, 1999). Although the findings of the studies were not unanimous, what was a common finding for all the studies was that televised sport did not adversely affect match attendance (Buraimo, 2006).

Szymanski and Kuypers (1999) found that one of the factors that contributed to declining match attendance was the increasing availability of alternate forms of entertainment. This suggests that other forms of entertainment can be seen as a substitute for match attendance. Montgomery and Robinson (2008:14) found that "sports do compete with arts events for attendance shares", which similarly suggests that other forms of entertainment could in fact be a substitute for sports consumption. Feehan (2006), however, argued that there is uncertainty in terms of what constitutes a 'substitute good' for live attendance. He stated that while other forms of entertainment and even other sports events in the locality may be classified as a substitute for match attendance, it may not necessarily always hold true. For example, match attendance will have no substitute goods if a devoted sports fan were to attend all the matches regardless of other attractions (Feehan, 2006). According to Simmons (2006), complementary goods for sports events include goods such as a match-day programme, betting opportunities,

entertainment provided before and during the event, catering at the event and the intangible products such as the atmosphere, fan solidarity and fan identity, which unify sports fans globally. Eschenfelder and Li (2007) noted that such complementary goods would be used in conjunction with match attendance and therefore have a positive demand relationship.

Feehan (2006) stated that market size has been found to be a major determinant of match attendance demand; a finding that he acknowledged is common to a range of sports. Feehan (2006) argued that a large market size is associated with high attendance levels when other factors are held constant. This is supported by Eschenfelder and Li (2007), who noted that the size of the market is an important economic factor that influences the demand for goods and services. Studies in a number of sports have investigated the relationship between the size of a market and match attendance and the findings appear unanimous. Examples of such studies include Malaysian Semi-Pro League Soccer (Wilson and Sim, 1995) and Major League Baseball (Schmidt and Berri, 2001). Both studies found that market size had a significant and positive effect on match attendance. These arguments, in conjunction with Simmon's (2006) argument on excess demand, suggest that sports consumption is dependant or at least to some degree influenced by the extent of access that consumers have to the sport. Furthermore, it could be suggested that further research should take into account the extent to which potential consumers would have access to or would be able (in terms of venue capacity) to consume the sport if they so wish.

The tastes or preferences of spectators are factors that have been identified to influence demand for sport. According to Eschenfelder and Li (2007), consumer tastes or preferences are subjective elements that influence the consumption behaviour of individuals. Individuals base their consumption decisions on internal aspects of their being (Rutenberg, 2003). Some of the factors that have been identified include personal preferences, social influence, habits, customs, and beliefs (Tamara, 2011). These factors are explored in more depth in Section 3.7.

Uncertainty of outcome is another major factor recognized to influence demand and has been identified as one of the most important tenets in analysing attendance demand for sport (Feehan, 2006). The general hypothesis that underpins this factor is that outcome uncertainty boosts attendance demand. According to Feehan (2006), there is little empirical evidence that supports this hypothesis. A literature study by Szymansky (2003) reviewed the effect of uncertainty of outcome on attendance demand for a wide range of sport. Of the 22 cases that were reviewed, ten suggested that outcome uncertainty had a positive impact on demand,

seven offered weak support for the relationship between outcome uncertainty and demand, while the remaining five cases found outcome uncertainty not to be an influencing factor on demand. From this, it was concluded that the effect of outcome uncertainty on demand is conflicting (Szymansky, 2003). A related study by Gerber and Terblanche (2012) considered the relationship between team success and spectator attendance (or demand) in the case of provincial rugby in South Africa. Their results revealed that there was a significant positive relationship between the games won and final log positions of a team and spectator attendance. It was concluded that team success was an influencing factor on attendance demand (Gerber and Terblanche, 2012).

Another determinant of consumption behaviour in general (and applicable in the case of sports consumption too) is the availability of resources, specifically, time. According to Sullivan and Katz-Gerro (2007), the resource of time that can be divided between working and leisure has become an accentuated scarce resource. Sullivan and Katz-Gerro (2007:9) argued that "(i)ndividuals with high levels of human, economic, and cultural capital have less time for leisure" but very interestingly found that despite their relative time constraint these individuals "still engage in a greater range of out-of-home leisure activities". Thus, their study suggests that even though these individuals have less time at their disposal, time does not seem to be much of a constraining factor in terms of the variety of leisure activities these individuals choose to consume. Southerton (2003) proposed that a possible reason for this might be that individuals who are faced with relatively greater time constraints are more likely to compress, fragment and compartmentalize their time. This suggests that individuals with less time for leisure activities (including sport) are likely to allocate it in such a way that they are able to fragment their time to consume a wide range of leisure activities as opposed to concentrating their time on one or few leisure activities (Southerton, 2003).

There are a number of other determinants that have been suggested in empirical evidence, which could influence match attendance or the demand for sport. Szymanski and Kuypers (1999) suggested that low attendance figures (or demand) can be attributed to factors such as declining standards of play, declining excitement of the match, unsafe/dilapidated stadiums, the presence of hooliganism at the event and the increasing availability of alternate forms of entertainment (which supports the notion of other forms of entertainment as a substitute for sport, as discussed earlier). Simmons (2006), in turn, suggested factors that could be attributed to high attendance figures, which included a change in tastes towards a sport, the redevelopment of stadiums, more comfortable facilities, and an increase in complimentary

facilities such as restaurants, television screens, and shops. This, in conjunction with the factors proposed by Szymanski and Kuypers (1999), suggests that complements and substitutes (which both constitute other forms of entertainment) have a significant impact in influencing sports consumption.

According to Wakefield (1995), one of the most influential factors determining sports consumption is social influence. Wakefield (1995) argued that, due to their use of aggregate data, previous empirical studies often overlooked the impact that social factors play in influencing the consumption of sport. Choi *et al.* (2009:265) similarly argued that an individual's reference group has a "significant impact" on sports consumption decisions. Zheng (2014) agreed and noted that sports consumption is largely influenced by the specific social environment. Furthermore, Zheng (2014:378) argued that "sports consumption is not only defined as an economic behavior, but also a social behavior, so, it will unavoidably be affected by various non-economic factors in the social life of human beings". This therefore highlights the significant role that social influences play in determining sports consumption behaviour. As in the case of cultural and entertainment leisure activities, the idea that sport may similarly be bound up in social hierarchies has been empirically investigated through the application of the omnivore-univore theory in various sporting contexts and is an important area of future research.

2.5 Summary and conclusion

This chapter presented a review of some of the theoretical underpinnings of consumption behaviour. Since its inception, consumption theory has developed to account for the important social aspect that underpins or at least to some extent can be used to explain consumption behaviour. From the review of the theoretical framework, it is evident that modern consumption theory is anthropocentric in nature, with people and societal influence at the forefront of the theory. The chapter reviewed determining factors of leisure consumption and then considered the factors that affect the demand for sport as a form of leisure consumption. The determinants of sports consumption included ticket prices, costs associated with consumption, spectator income, price and availability of complements and substitutes, market size, tastes and preferences, uncertainty of outcome, availability of resources, and social influences.

CHAPTER 3: LITERATURE REVIEW– EMPIRICAL STUDIES

3.1 Introduction

Theoretical understandings of consumption behaviour acknowledge social influence as a determinant of consumption behaviour (Veblen, 1899; Wakefield, 1995; Cornea and Jeanne, 1997 and Choi *et al.*, 2009). The idea of social comparison and thus social stratification in leisure consumption was initially investigated for the case of cultural activities (for example, Bourdieu, 1984; Wilensky, 1964; Peterson, 1992; Lamont, 1992) and has since developed to include the consumption of sport (for example, Miles and Sullivan, 2010; Lefvre and Ohl, 2011; Warde, 2006; Widdop and Cutts, 2013). The idea of social stratification of the consumption of culture (as a form of leisure consumption) was initially coined by Bourdieu (1984), and has developed into what is known today as the omnivore/univore theory. This chapter starts by presenting an overview of the omnivore/univore theory, where a review of the main arguments is provided. An empirical stance is then adopted.

To date a number of empirical consumer behaviour studies have been conducted in different contexts. Section 3.3 provides a review of cultural consumption behaviour studies, where the omnivore/univore theory was originally investigated. Section 3.4 provides a review of empirical work on the developments of the omnivore/univore theory since it first appeared in a binary form, while Section 3.5 looks at the developments of the omnivore/univore theory beyond the cultural context. Section 3.6 then expands on this and reviews empirical studies on the application of the omnivore/univore theory in sport. The focus of the chapter then shifts to considering the motives for attendance. Section 3.7 provides a brief overview of factors that have been identified as motives for attendance from empirical studies in both entertainment and sports consumption studies. A final section then summarises and concludes the chapter.

3.2 The omnivore/univore theory

Empirical work on consumption behaviour of different forms of culture has established a definite link between cultural consumption patterns and social stratification (Widdop, 2010). According to Chan and Goldthorpe (2007), the idea of linking cultural consumption to lifestyles dates back to 1964, where empirical research by Wilensky (1964) suggested that individuals from a high

educational background rarely had a strong aversion to culture that was associated with the masses. Wilensky (1964) further noted that highly educated individuals in fact often enjoyed 'mass' culture in at least some of its forms. This relationship, which in essence looked at the prevalence of an omnivore/univore classification in arts, was originally theorized in 1984 when Bourdieu (1984) proposed that social status could be used to understand the cultural consumption preferences. According to Peterson (2005), the theory suggested by Bourdieu (1984) was the first of its kind in that it proposed a theoretically grounded means of conceptualizing the relationship between consumers' tastes, status and social class. Bourdieu (1984) suggested that cultural consumption preferences could be classified according to 'high' and 'low' cultural forms. In order to appreciate 'high' cultural forms, he argued that it required one to have high cultural capital, which, inter alia, stems from an individual's upbringing, education and socio-economic background. Bourdieu (1984) noted that these 'high' cultural forms were more likely to be consumed by individuals from high income and high education backgrounds, who were considered to be of a high social class. At the opposite end of the spectrum, 'low' cultural forms, which were referred to as mass or popular culture, were cultural forms that were more likely to be consumed by individuals from lower income and lower education backgrounds and were thus perceived as being of a lower social class. As such, Bourdieu (1984) argued that the appreciation of cultural forms was an indicator of social class and owing to the fact that appreciation of 'high' cultural forms required an individual to have high cultural capital; the "masses" who failed to possess such high cultural capital were essentially excluded from the consumption thereof. Finally, Bourdieu (1984) noted that cultural appreciation, which was essentially indicative of cultural capital, provided a means of judgment of social superiority.

Peterson (1992) contested Bourdieu's (1984) hypothesis that individuals from high income and educational backgrounds would more likely consume 'high' culture forms and individuals from lower income and educational backgrounds would more likely consume 'low' cultural forms. Peterson (1992) investigated patterns of musical preferences and social status in the U.S. The investigation revealed that while elements of Bourdieu's (1984) argument did hold true, his hypothesis did not hold in its entirety. Peterson (1992) argued that while 'high' cultural forms (which in the case of music were classified as the classical and opera genres) were more likely to be preferred by individuals from high income and high education groups, such patrons had also highlighted a preference for a wide range of popular/non-elite musical forms (for example, folk, jazz and rock genres). Peterson's (1992) thinking in this regard was therefore in line with what Wilensky (1964) had originally suggested. Furthermore, Peterson (1992) noted that individuals from lower education and lower income groups were conversely generally found to appreciate only one genre of music (which was indicative of narrower cultural preferences). As a concluding remark, Peterson (1992) suggested that the notion of social superiority or the existence of the taste-exclusive stratification was no longer applicable in explaining cultural consumption and social status. Instead, he proposed that the classification of patrons' tastes in terms of omnivorousness was a more suitable indicator of social status. This finding was in line with the findings of Lamont (1992).

Consistent with Peterson (1992), Lamont (1992) also contested Bourdieu's (1984) hypothesis. Lamont (1992) considered whether the relationship between social status and tastes was the same across geographical boundaries; essentially questioning Bourdieu's (1984) claim that high-brow patterns of taste were directly attributed to social class and therefore should be the same in all advanced capitalist societies. Lamont (1992) focused on Paris (the same focus area as Bourdieu (1984) originally considered), New York, as well as regional cities in France and the U.S respectively. Lamont (1992) found that high-brow snobbery was evident in all four of the cities but was only found to be a predominant pattern in Paris. This therefore brought to light the fact that high-brow tastes were not the same in all advanced capitalist societies as Bourdieu (1984) had suggested.

The contested hypothesis and differences in empirical findings suggested that there is a need for further empirical investigations. This was supported by Scheetz *et al.* (2014:5) who noted that "while many economists have highlighted the role of social comparison in consumer behavior, little work has been done to empirically test this idea."

3.3 Empirical studies investigating the omnivore/univore theory

Since Peterson (1992) and Lamont (1992) contested Bourdieu's (1984) hypothesis, a wide array of studies have considered the application of the univore/omnivore theory in various contexts. In a review by Peterson (2005), he noted that writers had extensively considered the notion of highbrow snob and omnivority since 1992 when Bourdieu's (1984) theory was first contested. Such studies have been conducted in a range of different countries, in Europe, North America and Australia.

Peterson and Kern (1996) and Rossman and Peterson (2005) considered the extent to which omnivorousness had changed over a ten year period. Peterson and Kern (1996) compared the findings of a national survey from 1982 to 1992, while Rossman and Peterson (2005) considered the same, for ten successive years thereafter. The findings of the two studies were consistent, where an increase in the prevalence of omnivorousness in high-status individuals was found for both of the decades that were investigated (Peterson and Kern, 1996; Rossman and Peterson, 2005). Both studies suggested that this finding was attributed to high-status individuals becoming more omnivorous over time and younger, high-status cohorts, whose tastes were more omnivorously inclined, were replacing older cohorts whose tastes were more snobbishly inclined (Peterson and Kern, 1996).

Since 1996, the focus of such studies has shifted to considering what the prevalence of omnivority and univority has implied for social tolerance. In a study by Bryson (1996), which looked at musical tastes, a strong positive relationship between omnivorousness and social tolerance was found. Similarly, Tampubolon (2008) found that univores were more likely to consume one or only a select few cultural genres, which suggested that they had tendencies to exclude other social groups. Omnivores, on the other hand, had relatively more socially inclusive tendencies and were therefore argued to be relatively more socially tolerant when compared to univores. Coulangeon (2005) supported these findings in suggesting that people from high status groups are likely to have relatively more diverse social ties and as a result thereof are likely to be relatively more eclectic than people from a lower social status. Van Eijck and Leivens (2008) proposed a development of these studies and considered the possible reasons underpinning the relationship between social tolerance and omnivore/univore status. They argued that while omnivores were more likely to be socially tolerant than univores (which supported the previous empirical studies), the reason for univores being relatively less socially tolerant was more to do with the fact that, owing to their exclusive cultural tastes, these individuals felt socially isolated. The study also noted that the extent of social tolerance depended on how the individuals chose to combine different cultural genres. Their argument proposed that unlike univores who felt socially isolated, omnivores were orientated towards the welfare of others and as such would not feel socially isolated. Van Eijck and Leivens (2008) further proposed the notion of 'high-brow dominance'. 'High-brow dominance' referred to the idea that while a high-brow univore would be an individual who is likely associated with a closed and therefore isolated attitude, individuals who were more likely to consume high-brow cultural forms were more likely to be omnivores. From this finding, they concluded that there was a positive relationship between high-brow tastes and social inclusion.

In relation to the omnivore/univore theory, if high-brow tastes do require a high degree of cultural capital as proposed by Bourdieu (1984) (which is, in turn, associated with an exclusive taste group and therefore social isolation), then Van Eijck and Leivens' (2008) study similarly contested Bourdieu's (1984) theory. The findings are, however, in line with the proposition by Peterson (1992) in that although 'high' cultural forms were generally more likely to be appreciated by high social status individuals; these high status individuals were also found to appreciate a wide array of cultural forms (including non-elite cultural forms), which in turn was associated with social inclusion (Bryson, 1996; Van Eijck and Leivens, 2008).

3.4 Developments of the omnivore/univore theory: Binary to multi-dimensional form

As noted by Antrobus and Snowball (2010), there have been a number of developments on the omnivore/univore theory since it first originated in its binary form. Such studies (Alderson *et al.*, 2007; Chan and Goldthorpe, 2007; Snowball *et al.*, 2009; Antrobus and Snowball, 2010) have suggested different sub-classifications of omnivores/univores to account for the variations of consumption within an omnivore/univore status.

The derivation of the sub-classifications throughout all the studies was dependent on the consumption decisions of attendees of the various cultural forms. Although all the studies essentially investigated the consumption behaviour of the same core product (cultural forms), the sub-classifications that were derived differed according to their respective contexts and the different consumption patterns that emerged.

A study that was largely in line with the general omnivore/univore theory was conducted by Chan and Goldthorpe (2007). The study looked at applying latent class models to musical consumption in an attempt to identify different types of music consumers. Chan and Goldthorpe (2007), however, chose to derive a sub-category of the traditional omnivore classification to account for different forms of cultural consumption (listening to music and attending musical events respectively). Thus, they made use of the traditional univore classification but differentiated between true omnivores and omnivore-listeners. Alderson *et al.* (2007) similarly identified three different classifications of cultural consumers. For this

study, individuals were either classified as 'omnivores', 'paucivores' or 'inactives'. The classification of omnivores was largely in line with the traditional classification thereof, where individuals who had a high probability of consuming a wide range of cultural products (including both 'high' and 'low' cultural forms) received this status. Paucivores was a unique classification that Alderson et al. (2007) identified. These individuals had neither radically eclectic tastes nor particular tastes, and therefore were deemed to have intermediate levels of cultural consumption. Given this definition, this classification can therefore be argued to fall between the traditional understanding of an omnivore and univore. While paucivores were seen to consume a wide array of cultural products, Alderson et al. (2007) noted that their consumption patterns suggested a bias towards popular culture. The final classification of cultural consumers was the 'inactives'. These individuals had a relatively low probability of consuming any of the cultural products that were incorporated in the study. Similar to both Chan and Goldthorpe (2007) and Alderson et al. (2007), Snowball et al. (2009) and Antrobus and Snowball (2010) derived different categories to classify the consumption patterns of the cultural consumers. Both studies chose to focus on the consumption behaviour of the attendees of live theatre performances at the Grahamstown National Arts Festival. Despite analysing consumption behaviour at the same Arts Festival, the sub-classifications differed. This suggested that consumption classifications could be unique to the consumers that are under investigation and therefore suggests that research undertaken should consider the derivation of sub-classifications based on the consumption patterns that emerge for a specific group of consumers in a given context.

As Chan and Goldthorpe (2007) had done, Snowball *et al.* (2009) chose to make use of the traditional understanding of the univore classification, and similarly chose to differentiate between different types of omnivores. Three types of omnivores were identified. The first category of omnivores was referred to as the 'modern omnivores'. This category was comprised of attendees who chose to consume one of the following forms of entertainment – Comedy, Movies or Popular Music. The second category was that of 'traditional omnivores', where attendees chose to consume more than two of the following forms of entertainment – Classical Music, Art, Dance or Musicals. The final category was 'cultural omnivores', which comprised of individuals that consumed any four or more types of the live theatre performances that were offered at the National Arts Festival. Univores, on the other hand, were attendees that either chose to consume only one kind of live theatre show at the Festival or they were attendees that chose to consume only two types of entertainment from the

traditional category. From a review of these sub-classifications, it becomes evident that the *type* of product being consumed is important in the derivation of sub-classifications.

Antrobus and Snowball (2010) adopted a relatively broader stance and considered the general consumption behaviour of the Arts Festival as an event (as opposed to Snowball *et al.*'s (2009) stance of looking at consumption within the Arts Festival). The study focused on whether festivals facilitated and promoted omnivorous cultural consumption. More particularly, it considered whether the festival would promote and facilitate omnivorous cultural consumption behaviour amongst consumers who would otherwise be classified as univores. Antrobus and Snowball (2010) investigated the possibility of deriving a new classification of cultural consumers, this being the 'festivores'. Antrobus and Snowball (2010:338) defined the festivore as being a "festival patron who consumes the festival as a total event, which includes various art and theatre forms and cultural goods". From the identification of a further sub-classification, the focus of the research (for example on an entire event or consumption patterns within an event) similarly plays an important role in the development of sub-classifications of consumption behaviour patterns.

The derivation of the sub-classifications of omnivores/univores suggests that since the omnivore/univore theory was first proposed, the application thereof has developed from its original binary form to a multi-dimensional form. Although the nature of the studies was different, their results bore similarities both to the traditional/binary omnivore/univore theories (Bourdieu, 1984; Peterson, 1992) and to each other.

In terms of the general findings, Alderson *et al.* (2007), Snowball *et al.* (2009), Antrobus and Snowball (2010) and Chan and Goldthorpe (2007) were all in support of Peterson's (1992) view of the omnivore/univore status and therefore similarly contested Bourdieu's (1984) original proposition. Alderson *et al.* (2007) suggested that high status individuals were more likely to be classified as 'paucivores' than 'inactives'. High status individuals were also more likely to be classified as 'omnivores' than 'paucivores'. Therefore, both findings were consistent with Peterson (1992) in that higher status individuals were found to consume a wide array of cultural forms. Snowball *et al.* (2009) found that the attendees of the Festival were representative of the relatively higher income and higher education groups in South Africa and, as such, the results were consistent with Alderson *et al.* (2007) and similarly in line with the thesis proposed by Peterson (1992). According to Chan and Goldthorpe (2007), relatively higher status individuals were more likely to be true omnivores than univores. High

status individuals were also more likely to be classified as a true omnivore than an omnivorelistener. This finding similarly supports Peterson's (1992) understanding of omnivorous traits in that individuals from high status backgrounds tend to prefer a wide variety of cultural products. While the findings of Antrobus and Snowball (2010) were generally in line with Peterson (1992), elements of Bourdieu's (1984) original preposition were evident. Generally, Antrobus and Snowball (2010) found that omnivores were predominantly from upper socioeconomic backgrounds and were from a relatively higher educational background. This finding was in accordance with the finding of Peterson (1992) in that higher-status individuals preferred a wide array of cultural products. According to Antrobus and Snowball (2010), the findings of the study were, in turn, generally similar to that of Snowball et al. (2009) in that there was no evidence of the Bourdieu-like 'high' cultural consumption patterns. Antrobus and Snowball (2010), however, found that physical theatre and ballet (which are often classed as 'avant garde' and 'elitist' cultural forms respectively) were predominantly consumed by omnivores. Therefore, if omnivores were of high socio-economic and high education backgrounds (as the findings suggest), then the consumption of ballet and physical theatre by omnivores is essentially in line with what was originally proposed by Bourdieu (1984).

While the reviewed empirical studies mostly agreed on their general findings, differences in findings pertaining to demographic variables were evident between the studies. In terms of gender, Antrobus and Snowball (2010) found that generally, women were more likely to be classified as omnivores than men were. This finding was in line with that of Snowball et al. (2009). More particularly, Snowball et al. (2009) noted that there were distinct differences in the preferences of women and men. While men were found to prefer pop music, comedy and (to a lesser extent) classical music, women showed a preference for fine art, dance and movies. Chan and Goldthorpe (2007), on the other hand, found that women were more likely than men to be classified as univores; a finding that contradicts those of both Snowball et al. (2009) and Antrobus and Snowball (2010). In comparison to men, women were less likely to be classified as omnivore-listeners than true-omnivores (Chan and Goldthorpe, 2007). Collins (1992) put forward an interesting proposition as a possible explanation for why women were found to be more likely to be classified as univores than men. Collins (1992) suggested that men are often over-represented in the dominant class positions, while women often occupy the less favourable position in the hierarchy of social class. As such, Collins (1992) argued that women might consume relatively more 'high' culture products than men to gain respect.

Age was the other demographic variable in which inconsistencies between the reviewed studies were found. The findings on age of Snowball et al. (2009) were consistent with those of Tampubolon (2008), as well as Peterson and Kern (1996) and Rossman and Peterson (2005). The results revealed that there was a significant difference between the attendees classified as 'traditional omnivores' and 'modern omnivores' in terms of their age. Snowball et al. (2009) suggested that the likelihood of being classified as a 'modern omnivore' was higher for younger Festival attendees, while the likelihood of receiving a 'traditional omnivore' classification was higher for relatively higher age differentials. This finding was consistent with the findings of both Peterson and Kern (1996) and Rossman and Peterson (2005), where younger, high status cohorts, whose tastes were more omnivorously inclined, were replacing older cohorts whose tastes were more snobbishly inclined. Both Antrobus and Snowball (2010) and Chan and Goldthorpe (2007) opposed this finding, but in turn disagreed on the role that age plays in the omnivore/univore classification. Chan and Goldthorpe (2007) found that younger people were more likely to be univores than older people and similarly older people were found more likely to be classified as true omnivores than omnivorelisteners. Antrobus and Snowball (2010), however, found that Festival omnivores were of a wide variety of age groups. This finding was inconsistent with the argument of other writers (for example, Peterson and Kern, 1996; Rossman and Peterson, 2005; Snowball et al., 2009). Antrobus and Snowball (2010) put forward that age was not a determining factor in the omnivore/univore classification.

3.5 Developments of the omnivore/univore theory beyond the cultural context

While the studies that have been reviewed thus far have all considered different types of *cultural* consumption behaviour, developments of the omnivore/univore hypothesis extend to other forms of entertainment as well. Chan and Goldthorpe (2005), for example, considered the omnivore/univore hypothesis in application to the entertainment industry. Their study focused on consumption behaviour of dance, theatre and cinema attendance. The study investigated whether individuals who attended these types of entertainment could similarly be classified as omnivores and univores and whether the classification of consumers could be socially stratified. Suominen (2007) also considered the application of the omnivore/univore theory across different mediums of entertainment, but focused on whether spectators of performing arts and spectators of movies were the same in terms of their consumption behaviour. Suominen (2007) specified ballet, dance performance and opera as 'high-brow' goods while movies were specified as 'low-brow' goods.

Although the studies were very similar in nature, the findings of these two studies not only revealed differences to each other but definite comparative grounds between consumption patterns of patrons of other forms of entertainment and patrons of cultural forms became evident. Chan and Goldthorpe (2005) found that with some qualification, attendees of dance, theatre and cinema (as alternate forms of entertainment) could be classified as either omnivores or univores. The results also revealed that omnivores were predominantly of a relatively higher social status than univores. This therefore suggested support for Peterson's (1992) omnivore/univore hypothesis in that high status individuals were also found to consume a wide range of entertainment products (or displayed a broad taste spectrum). Omnivores were found to be from relatively higher income and educational backgrounds than univores. Suominen (2007) disagreed with the finding on income and rather argued that income was not found to explain consumption decisions. In terms of education, Chan and Goldthrope's (2005) finding was again contested by Suominen (2007), who found that while education did significantly explain differences in consumption behaviour, higher educated individuals showed a greater probability of consuming performing arts (which in this study was used as a proxy for high-brow culture). In this regard, the Suominen's (2007) findings on education were consistent with what was originally proposed by Bourdieu (1984), in that those belonging to the dominant or bourgeoisie classes (to which higher levels of education were associated) affirm their higher social status by consuming and appreciating high-brow or legitimate cultural forms. In terms of the other demographic determinants, Suominen (2007) found that females were more likely to be classified as omnivores, as they consumed a wide array of both high-brow and low-brow cultural goods. This finding is in line with Snowball et al. (2009) and Antrobus and Snowball (2010). Although they differed in their findings, both studies concluded that consumption behaviour of the entertainment forms in question could indeed be socially stratified (Chan and Goldthorpe, 2005; Suominen, 2007).

As noted by Widdop (2010) and as is evident from this review, empirical evidence to date suggests that consumption behaviour of leisure activities (culture and entertainment) is still largely bound up by social hierarchies. Widdop (2010) further highlighted that empirical investigations on consumption behaviour to date had been encumbered with investigations in the fields of art and music, which he noted have often come to the detriment of other cultural forms, especially sport. To date very little empirical work has been done on the application of the omnivore/univore theory to sport. Warde (2006:108) further stated that sport is a domain of extensive ramifications for a country's economy, society and culture and "is often treated

as a rather trivial topic, and thought to be far less important in understanding the relationship between economy, society and culture than many minority arts and leisure activities". Owing to these and other ramifications (as were highlighted in Section 1.1), and the fact that "theoretically sport consumption remains at large underdeveloped" (Widdop, 2010:2), there exists a need to further research and develop the application of consumption behaviour theory to the sport context. The next section of this review considers sport as a form of leisure consumption, and reviews the determinants of sports consumption as well as existing empirical evidence on the application of the omnivore/univore theory in sport to date.

3.6 Empirical studies investigating the omnivore/univore theory in sport

Widdop and Cutts (2013) investigated whether consumption of sport is related to social status in England. The study aimed to investigate Bourdieu's (1984) theory of consumption behaviour being an indicator of social status. The focus was specifically on individuals' actual participation in different sporting codes as opposed to focusing on self-reported tastes/preferences for different form of consumption (from the consumption forms that are of interest for this study - match attendance and television viewing), it is argued that there is a positive relationship between sports participation and sports consumption (for example, White and Wilson, 1999; Wilson, 2002; Thrane, 2001; Mehus, 2005). Furthermore, according to Karg and McDonald (2011:327), "sport involvement complements traditional sports consumption". Given this line of argument, literature considering the omnivore/univore theory for sports participation will also be reviewed.

According to Widdop and Cutts (2013), cultural studies have stressed the importance of distinguishing between 'mass/popular' and 'high/elite' cultural forms. As such, their study included a wide range of sport that would cut across these existing perceived symbolic boundaries. The study included 29 different sporting codes thought to be representative of sports that were more likely to be associated with the masses (for example soccer) as well as sports that were more likely to be associated with exclusivity (for example water, racket and adventure sports). A number of demographic variables were included in estimating patterns of consumption behaviour. These included age, gender, ethnicity, education, and marital status. The study made use of a Multiple Indicator Multiple Cause (MIMC) model and a Latent Class analysis; and through the estimated regressions was successfully able to classify sports consumers into five different categories. The first consumption category was termed 'classic
omnivores' and referred to individuals whose consumption behaviour cut across the perceived symbolic hierarchies of sport between exclusive/high-brow and popular sporting codes. More specifically, 'classic omnivores' were those individuals who had a high propensity of participating in all types of sport (or displayed a wide range of sport consumption). The second classification was termed the 'fitness class' and represented individuals who were more likely to participate in swimming, cycling, general fitness and adventure and water sports. They similarly were found to distance themselves from male dominated sports such as soccer, cricket and golf. The third classification was termed 'low-brow omnivores'. According to Widdop and Cutts (2013), this group was similar to the 'fitness class' in that they displayed similar omnivorous consumption patterns (in that both classifications displayed a moderate probability of participation in both popular and exclusive sporting codes). The major difference between the 'fitness class and 'low-brow omnivores', however, was that unlike the former, the latter did participate in the male dominated sports. As noted by Widdop and Cutts (2013:118), individuals classified as low-brow omnivores "did not readily consume exclusive sports that are perhaps indicators of privileged lifestyles". The fourth classification was termed 'high-brow' and as the name suggests, these were individuals who chose to part-take in high-brow sports to the exclusion of popular or mass sports. These consumed the high-brow or exclusive sporting activities such as golf, racket sports, adventure sports and water sports. Widdop and Cutts (2013) noted that these individuals were similar to the 'high-brow snobs' as identified by Peterson and Kern (1996) in the cultural context. The final classification was referred to as the 'nonparticipants'. These individuals had a very low probability of participating in any form of sporting activities (regardless of the nature of the sport). While these individuals may have consumed sport in the form of watching sport on television or attending sporting events as a spectator, Widdop and Cutts (2013) chose to narrow the focus to individuals who physically participated in a sport.

Widdop and Cutts (2013) found that the 'classic omnivore' group, which represented 9% of the overall sample, was predominantly composed of individuals from higher social strata and higher educational background. This supports Peterson's (1992) view of omnivores in acknowledging that consumers from high social strata are likely to consume a wide range of (in his case cultural) goods. Individuals in the 'classic omnivore' group were found to be predominantly White males and generally fell between the ages of 25 and 44. The 'fitness class' group, which represented 27% of the overall sample, was found to be of a mixed social class and education background. This classification bares similarities to what Alderson *et al.* (2007) identified as 'paucivores' in the cultural context. These sports consumers were

similarly individuals who had neither radically eclectic tastes nor particular tastes, and were similarly deemed to have intermediate levels of sports consumption (Alderson *et al.*, 2007). The 'fitness class' group was representative of individuals from higher and lower social classes and educational backgrounds. This group posed a number of similarities to the 'classic omnivore', but a significant difference in the gender explanatory variable was evident. The likelihood of being classified under the 'fitness class' was significantly higher for women than for men. Owing to the fact that 'fitness class' was comprised of consumption of quite a broad range of sport, this finding is in line with Snowball *et al.*, (2009) and Antrobus and Snowball (2010) where women are more likely to be classified as omnivores (in their case represented by a broader taste spectrum than men).

In terms of the 'high-brow' group, which represented 13% of the sample, Widdop and Cutts (2013) found that this group comprised of individuals from high and medium social strata and displayed a broad range of educational backgrounds. Widdop and Cutts (2013) noted that similar to the 'classic omnivore' group, the 'high-brow' group was predominantly made up of White males but these two groups were found to be significantly different in that the individuals from older cohorts were more likely to be classified as 'high-brows' than individuals from younger cohorts. This finding is consistent with the findings of Peterson and Kern (1996), where older cohorts are more likely to display snobbish taste preferences. The fourth classification, 'low-brow omnivores', represented 13% of the total sample. The 'lowbrow omnivores' were found to be from lower social strata as well as from lower educational backgrounds and predominantly fell between the ages of 16 and 24. This finding is similarly argued to be in line with Peterson and Kern (1996), in that younger cohorts are more likely to display omnivorous taste preferences than are older cohorts. The remaining 41% of the sample were classified as non-participants and were found to generally be older women from lower educational backgrounds and lower social strata. In conclusion, Widdop and Cutts (2013:107) noted "the findings largely corroborate research in other cultural and leisure fields" both in terms of the similarity between sporting and cultural lifestyle types and in terms of the different classifications' socio-economic compositions.

Lefevre and Ohl (2011) investigated whether omnivority was prevalent in upper social groups evident in their disagreement of choice with regard to their consumption of physical and sporting activities in France in 2000. The study was primarily concerned with the extent to which sporting fields and social positions were related. The upper social class was the focus group of the study where the upper social class included "managing directors of industry and business, executives, intellectuals and those in similar professions" (Lefevre and Ohl, 2011:47). The study focused on participants of sport between the ages of 15 and 75 and comprised of 6526 participants. The results showed that 90% of the sporting activities identified by participants were sports that they both participated in and liked. They argued that participation in sport provided an apt proxy for tastes (Lefevre and Ohl, 2011). The study initially made use of 279 different sporting and physical activity codes, but later filtered out any sport that did not represent the participation of at least 30 upper class individuals. The study then focused on the remaining 70 sporting codes, which were argued to be representative of the upper class sports consumption. The results showed that of these 70 sporting codes, 14 of them appeared to be socially discriminating. According to Lefevre and Ohl (2011:48), "the social ranking of sports is generally not very marked". They further found that activities that dominate are generally widely diffused between the social groups that participate in them, and that the most distinct differences in terms of sports consumption appear to be in terms of age and gender. Univority was found to increase with age (a finding that was supported by Peterson and Kern (1996) and Rossman and Peterson (2005) in the cultural context) and was found to be more prevalent in women than in men (in contrast to the findings of Snowball et al., 2009, Antrobus and Snowball, 2010 and Widdop and Cutts, 2013). As noted by Lefevre and Ohl (2011:48), their study suggested that "distinction is not dominant among the upper class usages of sport and that the relation of homology between social positions and fields seems to be too simple to understand the complexity of usages of a mass culture such as sport". They further noted that "in a society in which information, styles, fashion and trends circulate faster than previously; homology is not the dominant characteristic of the relation between the sporting field and social positions" (Lefvre and Ohl, 2011:48).

In order to make inferences about the sample in terms of their omnivority, and in recognising the diversity of the forms of sports consumption, the study considered the extent to which individuals' consumption behaviour of sport was marked by multi-activity. Lefevre and Ohl (2011) identified three different classifications of sports consumption, where individuals were differentiated in accordance with the number of sporting activities they chose to partake in. Individuals were termed 'univores' if they participated in one sporting activity to the exclusion of others. Individuals were similarly classified to be of low omnivority if they participated in two or three sporting activities, while participation in four or more sporting activities meant that individuals were 'high omnivores'. Lefevre and Ohl (2011) found that 83% of the sampled individuals participated in at least one sport (while the remaining 17%).

did not participate in any sport). The results revealed that, on average, 2.96 sporting activities were consumed per individual, therefore indicating a clear presence of omnivority in the sample. Of the sampled sports participants, 28% were classified as univores, 41% were classified as 'low omnivores' while the remainder of the sample were classified as 'high omnivores'. In terms of different social levels, the results suggested that there was no statistically significant difference in terms of low omnivority. However, there were differences in social class when univores and high omnivores were compared. Univores were predominantly from the lower social classes, while the relatively higher social status individuals were predominantly classified as high omnivores. This finding therefore supports the proposal by Peterson (1992) in that high status individuals are likely to consume a wide range of (in this case sports) commodities and people of lower social status are conversely likely to be of a narrower taste spectrum. With statistically significant differences only found between the extreme opposite sides of the taste spectrum, this suggests that it may be valuable for further research to consider fewer sub-classifications of an omnivoreity).

Warde (2006) conducted a similar study to Lefevre and Ohl (2011), in which the relative importance of different social characteristics in predicting individuals' sports participation was investigated. One significant difference between the two studies was that Warde (2006) considered consumption both in the form of participating in sport and in the form of sports spectatorship. Initially, the study investigated the time-use surveys in Britain to obtain a general overview of reported sports participation. Warde (2006) suggested that sports participation differed significantly in terms of gender, age, occupational class and educational background. On average, men were found to participate in more sporting activities than women. Warde (2006:115) further noted that women who are of higher working classes are more likely to participate in sporting activities than women of lower working classes; suggesting a "significant interactive effect between gender and class". As would be expected, sports participation decreased monotonically with age. The survey results also suggested that higher levels of education were predominantly associated with higher levels of sports participation. In terms of class, the findings showed that the higher the class, the greater the participation in sports. In comparison to other empirical results, the findings on education and class are both in line with those of, amongst others, Peterson (1992) in the cultural context, and Lefevre and Ohl (2011) in the sporting context. Warde (2006) further investigated the relationship between the choice of sport and social status by considering the nature (or general classification) of sports that individuals reported to partake in. Warde (2006) acknowledged that owing to the large number of sports that individuals reported as participating in, and the relatively limited sample size, complex statistical analyses on consumers' preferences for all sports was not possible. As such, Warde (2006) chose to do frequency and cross-tabulation analyses of class, gender, education, age and ethnicity on preferences for the more popular sporting activities. It was found that individuals of professional and managerial classes were not only more likely to participate in sport but were also more likely to participate in a wider range of sporting codes.

The findings also suggested that while some sporting activities did reveal a class bias, sporting activities such as soccer, walking, jogging and general fitness, were found to be in no way correlated to a specific class. Conversely, sports activities such as dancing and bowls were more likely to be preferred by relatively lower class individuals, while cycling and squash were more likely to be preferred by the middle class, and upper class individuals predominantly preferred golf. Furthermore, sailing, cricket and basketball were sports that were less frequently reported to be participated in by lower working classes. Unlike the level of participation (as reviewed earlier), the type of sporting code was not influenced by the level of education. On the other hand, choice of sport significantly differed between ethnic groups. Warde (2006) found that all race groups other than 'White' were less likely to participate in golf and walking, but more likely to participate in cricket. Overall, gender was argued to be the "most powerful discriminating variable" (Warde, 2006:115). Men were not only found to be more likely to participate in sport, they also participated in a wider range of sporting activities than women. If omnivority is defined in terms of the breadth of consumption, this finding was in line with the findings of Lefevre and Ohl (2011) in the sporting context and similarly in contrast to the findings of Snowball et al. (2009) and Antrobus and Snowball (2010) in the cultural context. Warde (2006) concluded that in terms of participation in sport there are some distinctions between class' choice of sports consumption, particularly evident between individuals at the top and bottom of the class hierarchy in terms of their occupation. Therefore, Warde (2006:115) noted, "there are thus some elements which would support a Bourdieusian analysis of the role of cultural capital in sport". Furthermore, Warde (2006) found that younger individuals and individuals from higher class and higher educational backgrounds were more likely to participate in sport. According to Warde (2006), this finding is somewhat concerning in that, if sport is a means in which individuals are able to maintain their health and well-being, then the sports industry can be cultivating significant social inequalities (Warde, 2006).

In terms of sports spectatorship, Warde (2006) noted that social meaning can be obtained through investigating consumption behaviour, both in the form of live attendance and in the form of broadcast sports coverage. Warde (2006) considered which sport would be most popular as a spectator sport. The results indicated that soccer was the most popular spectator sport, with tennis nominated as the next most popular. Snooker, rugby union and Formula 1 racing were the next highest ranked spectator sports respectively. In terms of social characteristics, education, gender and class were determining factors of sports spectatorship (although more particularly in terms of specific sports as opposed to an overarching conclusion). Education did not seem to be a determining factor of soccer spectatorship; however, lower levels of education were more likely to be associated with the consumption of motor sport and snooker, while individuals with relatively higher levels of education were more likely to be spectators of rugby union and tennis. Gender was a very important determining factor of sports consumption. Of the sample, 24% of men indicated that sport is their favourite television genre while only 2% of women felt the same way. Conversely, 26% of women reported sport as their least favourite television genre compared to only 6% of men. These findings collectively suggested that there is indeed a strong gender bias in sports consumption, both in the form of spectatorship and in the form of participation (as reviewed earlier).

In terms of class, Warde (2006:119) found that "older unqualified working-class men particularly like snooker, boxing and wrestling, while large employers and senior managers are five times more likely than semi-routine manual workers to like watching golf most." Although Warde (2006) did not put forward any further comments on this finding; if it is considered in conjunction with the classification of sports by Widdop (2010) and Widdop and Cutts (2013) (where golf was classified as an exclusive sport); it could be argued that this finding supports the notion that individuals from a relatively higher occupational class hierarchy are more likely to consume exclusive sport. This, in turn, is similar to Widdop and Cutts' (2013) classification of 'high-brow' sports consumers, Peterson and Kern's (1996) classification of 'high-brow snob' cultural consumers and in line with Bourdieu's (1984) hypothesis that the elite usually defined themselves in relation to an exclusive set of cultural tastes and practices. From this finding, it could conversely be suggested that individuals from lower educational and occupational class hierarchy are more likely to consume sports that are associated with the masses (which could also be compared to Widdop and Cutts' (2013) classification of 'low-brow omnivores' who were found to be from lower social strata and lower educational backgrounds).

Warde (2006:119) argued, "there are certainly some hints that different sports carry connotations of social position, limitedly in relation to spectating, a little more in relation to participation". Warde (2006) further noted that a particularly pertinent finding was that sport creates greater divides between men and women; a finding that he argues requires greater appreciation than initially given by Bourdieu (1984). Sports consumption, both in the form of participation and spectatorship, indicated a strong gender bias. Warde (2006:120) argued that with women consuming a relatively narrow spectrum of sporting activities, their consumption behaviours suggested "some degree of exclusion". If this finding is looked at in terms of univority (when defined in terms of a narrow taste spectrum), it could imply that women in this case are more likely to display univorous sports consumption tendencies. This finding would then be in line with the previous sports consumption studies reviewed thus far and inconsistent with the gender findings in the cultural context as highlighted by Snowball *et al.* (2009) and Antrobus and Snowball (2010).

Miles and Sullivan (2010) adopted a broader perspective and investigated the relationship between taste and value in culture and sport. While this study looked at sport and cultural activities as a whole, it still proved a very useful study to review in terms of integration of tastes across sports and culture. The study made use of both an analysis of the Taking Part survey (which contained data on participation in sport, leisure and cultural activities in England) and a multiple-correspondence analysis (a method that had previously been used to analyse cultural participation and age). The results of the study were largely in line with other empirical analysts. As was found by Warde (2006), Miles and Sullivan (2010) noted that high levels of education were predominantly associated with participation in the activities, while lower levels of education were more likely to be associated with inactivity. The study also found that relatively higher age groups were more likely to be associated with 'high-brow' pursuits; a finding that confirmed those of Peterson and Kern (1996) in the cultural context and Widdop and Cutts (2013) in the sporting context, where older cohorts are more likely to display snobbish taste preferences. Consistent with Warde (2006) in the context of sport, Miles and Sullivan (2010) similarly found that inactivity was more prevalent amongst the older cohorts. In terms of television spectatorship, the results suggested that relatively older age groups were more likely to watch activities associated with 'high-brow' tastes while younger cohorts were more likely to consume 'low-brow' activities.

Miles and Sullivan (2010) investigated the extent to which adult encouragement affects the consumption behaviour of children. Parental encouragement and participation were strongly related and parents who encouraged children to participate in activities such as reading were also found to encourage participation in sport; possibly suggesting the important parental role in the development of a broad range of activity participation. Miles and Sullivan (2010:14) found that level of education is often linked to the level of encouragement, where "people with no qualifications are less likely to have been encouraged, not just in reading and the arts, but also in sport". Furthermore, they suggested that children who receive high levels of encouragement generally go on to have high levels of educational attainment (Miles and Sullivan, 2010). Another aspect of tastes that the study considered was the relationship between participation in cultural and sporting activities and well-being. Non-participation in activities was closely related to adverse aspects of well-being such as poor health and unhappiness. Conversely, participation in cultural and sporting activities was associated with positive aspects of well-being such as good health, happiness and non-smoking. These findings could therefore suggest the importance of encouraging, developing and enhancing the consumption of cultural and sporting activities to improve the well-being of a country's citizens.

In terms of the high-brow/low-brow dimension of the study, Miles and Sullivan (2010) found no link to the aspects of well-being. The findings of the study further highlighted the importance of "family formation and relationship status structure" as an influencing factor of participation (Miles and Sullivan, 2010:22). For example, it was found that prior to motherhood, some women were actively involved in a wide range of cultural and sporting activities, but since they had had children, their participation in activities had changed. Therefore, Miles and Sullivan (2010:22) suggested, "in almost every case, having children refocuses and privatises interests, reducing time and energy for independent or new pursuits". This could in turn suggest why women (in at least the sports studies reviewed thus far) are more likely to be classified as univores (in terms of having narrow consumption patterns) than omnivores. Miles and Sullivan (2010) further undertook interviews to gain a more in-depth understanding of cultural and sports participation. From the interviews, Miles and Sullivan (2010:20) noted "the archetypal 'low' or popular activity – TV viewing – can provide a means of access to legitimate culture". This highlights the importance of empirical research to include television spectatorship in order to gain a more accurate understanding of consumption behaviour. Miles and Sullivan (2010:21) further noted the importance of accounting for passing participation, where "(people) may actually visit museums, galleries or

the theatre occasionally but as incidental, uncommitted consumers looking for general entertainment, a distraction for children at weekends and holidays, or just 'something to do'." Therefore, Miles and Sullivan (2010) recognized the importance of consumption behaviour studies considering the motives underpinning consumers' attendance (to be able to determine whether the consumers' decisions form part of their usual consumption patterns or are as a result of an external influence). Amongst others, Snowball *et al.* (2009) highlight the importance of considering motives for attendance and furthermore suggested that motivation for attendance might be productively linked to the omnivore/univore theory. While Chapter 2 has already highlighted a number of possible factors that influence or underpin consumption behaviour, the next subsection provides a brief overview of some of the other motives for attendance that have been identified in existing empirical studies.

3.7 Motives for attendance: an overview of the entertainment and sports industry

According to Murray (1964:7), motivation can be understood as "an internal factor that arouses, directs, and integrates a person's behaviour". Therefore, understanding an individual's motivation can provide an indication of an individual's consumption behaviour. Furthermore, according to Laesso *et al.* (2013), understanding consumers' motives for particular consumption behaviour can facilitate producers and marketers in making effective business decisions, in that they are better able to base their production and marketing decisions respectively on the consumers' motives for particular consumption behaviour.

Swanson *et al.* (2008) studied the psychological motivation for arts attendance. Data for the study was obtained from a Midwestern United States performing arts centre. The sample of patrons attending the arts was obtained from three different performing arts events. Swanson *et al.* (2008) posed questions to the attendees pertaining to their motives for attending the arts, which ranged across six different motivational categories. These categories included recreation, social inclusion, escapism, self-esteem enhancement (which was taken as a proxy for status), and aesthetics. The reported motives for attending the arts were assessed according to the demographics of the sample. These included income, education, age and gender. In terms of income, the results suggested that individuals from relatively higher income levels were relatively more likely to be motivated by aesthetics, education and recreation, than lower income groups; the highest income group was more motivated by social interaction than any other income group. In terms of education, the results suggested that aesthetics, recreation and educational motives were significantly greater with higher levels of education. Individuals

from lower educational backgrounds were relatively more motivated by escapism than consumers with relatively higher levels of education. A particularly interesting finding in this regard was that "participants with the highest and lowest educational levels are more likely to be motivated by self-esteem than customers of moderate educational levels" (Swanson *et al.*, 2008:315). The study also found that older cohorts were more likely to report aesthetics, education, and recreation as their motives for attendance, while self-esteem was significantly higher only for individuals exceeding 50 years of age. Finally, motives for attendance differed across gender, with females being more likely to be motivated by aesthetics, education and recreation than are male respondents.

Swanson *et al.* (2008) concluded that motives for attendance generally differed between groups of different demographic characteristics, which Swanson *et al.* (2008) noted had implications for the marketing of arts events. With significant relationships being found between demographic variables and the educational, aesthetics and recreational motives, Swanson *et al.* (2008:319) proposed "marketing managers should pay close attention to artistic, educational, and recreational aspects of performing arts when developing services". Similarly, with a significant relationship found between education level and escapism, Swanson *et al.* (2008:319) suggested "managers should put significant resources into developing and delivering programs that are attractive to patrons with a low educational level".

Perkins (2013) considered the motives for popular music concert attendance in Australia. The sample was comprised of individuals between the ages of 18 and 32. The data was collected by means of an interview. In the first instance, participants were asked to describe what motivates them to attend popular music concerts (where the responses were completely unaided by the interviewer). The participants were then provided with a list of different motives and asked to provide insight into the applicability of these motives to their attendance of popular music concerts. The study identified eleven common motives for attendance, four of which were particularly pertinent in terms of underpinning attendance of popular music concerts allowed them to relive old memories to which music was associated, and that music often "took them back to their childhood" (Perkins, 2013:4). The results also identified another dominant theme; that the destination or venue of hosting the music concert highly influenced individuals' attendance. Another prevailing motivational factor underpinning music concert

attendance was status enhancement. For example, individuals reported that they enjoyed the "bragging rights" associated with concert attendance and that you were considered more of a fan if you attended a music concert than if you engaged in other forms of musical consumption (Perkins, 2013:5). The final motive for attendance of popular music concerts was the "opportunity to experience new music" or the opportunity to listen to music that would only be performed at music concerts (Perkins, 2013:4); for example, if artists played cover songs of another artist that one would otherwise not be able to enjoy. Perkins (2013:5) suggested "it is possible that popular music concert organisers may be able to employ different marketing tactics to broaden the appeal of specific concerts, beyond the promotion of genre and artist". This therefore suggests that an understanding of motives of attendance can be beneficial to producers and marketers and even result in the enhancement of the product that the consumers are offered.

In the South African context, Snowball et al. (2009) considered the factors that motivate attendance in the cultural context. The study tested for the following motives: escapism, entertainment, socializing, artistic (referred to as aesthetics in the Swanson et al. (2008) study) and status motive. These motives were regressed against socio-demographic data, which included household income, age, gender, nationality and home language (which was used as a proxy for race/cultural differences). The study showed that the entertainment motive was the most commonly reported motive underpinning cultural consumption. Artistic merit and escapism were found to be the second and third (respectively) most likely motives underpinning cultural consumption. Finally, education and status were found to generally be equally weighted. The results further indicated that the motivation for attendance differed with age, occupation and education (which Snowball et al. (2009) highlight is in accordance with Swanson et al. (2008)). People of higher level occupations were more motivated by social and status reasons than people of lower occupations. Individuals from relatively lower educational backgrounds were more likely to be motivated by educational factors, while people from relatively higher education were more likely to attend because of artistic motives. In terms of home language, the results suggested that individuals who indicated that their home language was one of European origin were relatively more likely to be motivated by education, status, and escapism, whereas individuals whose home language was of an African decent were relatively more motivated by entertainment (Snowball et al., 2009). Snowball et al. (2009) further considered how motivation differed between omnivores and univores. These results revealed only incremental changes between the two groups. Omnivores, for example, were generally more motived by entertainment than univores; univores were relatively more motivated by status and social reasons than omnivores. In concluding the research, Snowball *et al.* (2009:482) noted "motivations were found to be as important in determining consumption patterns as socio-demographic variables" and further noted that "this is a potentially fruitful area for further research".

In terms of the sporting context, James and Ridinger (2002) did a comparative analysis of sports consumption motives between male and female sports fans. The study focused specifically on determining whether the motives for being a fan of women's college basketball differed from the motives of being a fan of men's college basketball. The study found that men and women were not equally as likely to be sports fans. More specifically, men and women were not equally as likely to be general sports fans or specific sports fans, which suggested that gender biases were prevalent in terms of sports consumption. The overall results indicated that men's rating of fans of sport, in general, and fans of specific sports were significantly higher than that of female participants. Females, on the other hand, were found more likely to be a fan of a specific sport as opposed to a fan of sport in general. Although this was not the focus of James and Ridinger's (2002) study, one could argue that in terms of the spectrum of sports events, this finding could support the findings of Lefevre and Ohl (2011) and Warde (2006) in the sporting context (where females were generally found to have a relatively more narrow taste spectrum than men) and contrast the findings of Snowball et al. (2009) and Antrobus and Snowball (2010) in the cultural context (where the opposite holds true). In terms of the motivating factors underpinning attendance, James and Ridinger (2002) identified that the aesthetic motive differed between fans of men and women's basketball; where fans of women's basketball reported a greater appreciation for the aesthetic value than was found true for men's basketball. Finally, the following motives were also found to be higher for men than for women: social identity, empathy and achievement (i.e. sharing the feelings of disappointment or happiness of a team in terms of their performance results) and knowledge (where males want to know about the technical aspects of the game). As noted by James and Ridinger (2002), this therefore suggests that sport provides an important identity for males that is not shared for females. James and Ridinger (2002) interestingly suggested that it is importance for future research to be conducted on specific teams, in order to gain better insight into the motives that influence individual team support and the consumption behaviour of sports fans.

Cunningham and Kwon (2003) made use of a theory of planned behaviour to understand better the consumer's intensions to attend a specific sporting event (in their case, hockey).

Data were collected from 140 students from a midwestern university in the United States. The participants were individuals who took part in physical activity. The data collection was in the form of questionnaires, which analysed motives for attendance at the particular hockey games. A different questionnaire then considered the previous behaviour, attitudes, subjective norms, perceived behavioural control and the intensions of the sample. 'Previous behaviour' included an analysis of how many hockey matches the spectators had attended in the previous and current season, while attitudes measured how valuable the spectators perceived attending the matches to be. Subjective norms were included to determine the influence that significant others had on match attendance, whereas perceived behavioural control was included to consider time and money as the salient detriments of hockey match attendance. The study found that, in general, friends and family (although to a lesser extent) positively influenced hockey match attendance. Time and money were conversely found to generally constrain hockey match attendance (a finding, which Cunningham and Kwon (2003) noted is not surprising given that the sample was comprised of university students). The results also indicated that previous behaviour and attitude were positively related to the intentions of attending a hockey match. More specifically, previous behaviour significantly contributed to the understandings of the intentions of match attendance; with attitudes and subjective norms being relatively more salient predictors of intensions to attend than perceived behavioural control. Finally, Cunningham and Kwon (2003:140) suggested that these findings have "several practical implications for sport marketers". Given the positive finding of subjective norms, sport marketers should emphasize the social aspect of a sporting event in order to attract people to attend the fixtures. Based on their findings, they further proposed that sports marketers should draw on spectators' attitudes. Sutton et al. (1997) proposed that, an example of how this could be done, was through increasing player accessibility to fans. Cunningham and Kwon (2003), however, did acknowledge that a limitation of the study was that it considered only reported intensions and not actual consumption behaviour and therefore it was noted that there is scope for further research to consider actual consumption behaviour as well as to develop theories specific to the sports domain. Trail and James (2011:1) support this shortcoming where they noted, "there is a disconcerting lack of holistic understanding about the components and processes of spectator sports consumption".

To address this shortcoming, Trail and James (2011) proposed a comprehensive model to understand sports consumption behaviour. While it is acknowledged that other studies have previously proposed models to explain sports consumption, as argued by Trail and James (2011) these studies have been lacking in terms of extensiveness, in that they fail to include a variety of factors that have been known to influence consumption behaviour. The model proposed by Trail and James (2011) was derived from drawing on existing theory and empirical research and thus provided a comprehensive means of understanding spectator sports consumption. Trail and James (2011) acknowledged the important role that culture plays in influencing consumption behaviour of sport. As such, the model was illustrated with culture as an overarching influencer (an illustration of the model is provided in Appendix B). In the development of their model, Trail and James (2011) adopted the understanding of 'culture' from the definition proposed by de Mooij (2004:26) where culture is defined as "shared beliefs, attitudes, norms, roles, and values found among speakers of a particular language who live during the same historical period of a specific geographic region". Trail and James (2011) suggested that the sports consumption process is composed of four main aspects, namely motivation, activation, external constraints and post-consumption reaction (where, as mentioned earlier, culture is seen to influence motivation and activation directly, which in turn affects the other components of the consumption process).

Trail and James (2011) described motivation as an individual's values, goals, personality and needs. They noted that culture influenced motivational factors as well as influenced (both directly and indirectly) how individuals perceived the spectator sport product. In terms of the motivation stage of the consumption process, personality was understood as the "unique pattern of enduring thoughts, feeling and actions that characterize a person" (Bernstein et al., 2006:540). 'Needs', on the other hand, were understood in terms of Miller and Gordon's (1975:8) definition as a "definite state of an organism that recurs periodically and that has a specific requirement for its satisfaction". The model stated that an interaction between personality and needs influenced personal values and goals. The model adopted the understanding of 'value' as defined by Rokeach (1973:5), where value is defined as "an enduring belief that a specific mode of conduct or end-state of existence is personally or socially preferable to an opposite or converse mode of conduct or end-state of existence". The model states that values influenced personal goals, where personal goals were understood in the model as cognitions that correspond to an individual's objectives (Jolibert and Baumgartner, 1997). The model then showed that personal goals together with values influence the activation process.

Trail and James (2011) described 'activation' (also referred to as the decision-making stage) in terms of an individual's perception of the product. According to James and Trail (2001), the activation process begins with perception, which was understood as the process during

which the consumer is able to make sense of his/her environment. Perception was said to underpin the activation stage, which was comprised of awareness of the product, interest in the product and evaluation of the product. The model suggested that perception of messages from social media interacted with how individuals learnt which, in turn, determined an individual's awareness of the product. Once the individual becomes aware of the product, the model suggested that individuals begin to evaluate the relevance of the product. In conjunction with internal motivators, the evaluation of the relevance influences an individual's attitude towards the product. Trail and James (2011) argued that 'attitude' was the primary determinant of whether individuals chose to consume the product or not. More particularly, they hypothesized that attitudes towards sports products lead to intensions to consume them but that external factors could constrain both intensions and actual consumption.

External constraints were understood in the model as factors that restrict consumption behaviour. As noted by Trail and James (2011), previous empirical results (which include Kim and Trail (2010, 2011) and Trail, Robinson and Kim (2008)) have found that external constraints do affect sports attendance. The final stage of the consumption that was proposed in the model was the post-consumption reaction, which involved factors that either confirmed or refuted an individual's expectations (Trail and James, 2011). The model suggested that if an individual is satisfied (dissatisfied), then they were more (less) likely to repeat their consumption behaviour in future. The final aspect of the model, which was closely linked to satisfaction, was that of self-esteem. The model adopted Rosenburg's (1979) understanding of self-esteem, where self-esteem referred to a positive outlook from a self-evaluation of oneself. Trail and James (2011), as supported by previous empirical studies, noted the inclusion of self-esteem in the model as a factor used to explain consumption behaviour. For example, Trail et al. (2005) found that 49% of sport spectator behavioural intentions were explained by self-esteem behaviours. Trail and James (2011) finally noted that the derived model proposed a holistic understanding of the processes and components involved in spectator sports consumption.

In concluding this section of the review, it is evident that the consumption process is a complex process and is influenced and underpinned by a wide range of internal and external factors. It can therefore be argued that, owing to the subjective nature and individuality of the decisions governing consumption behaviour, it is necessary for each individual context to undertake research on motives for attendance for a specific group of individuals in order to

comprehensively understand their consumption decisions as well as pattern of consumption behaviour.

3.8 Summary and conclusion

This chapter adopted an empirical stance. The main propositions of the omnivore/univore theory by Bourdieu (1984) were reviewed, as well as the contested stance on consumption behaviour as proposed by Peterson (1992). Since the origin of the omnivore/univore theory, writers have proposed extensions and developments to the theory, which were then reviewed. It became evident from this, that it is valuable for studies to consider definitions other than the original or conventional definitions of omnivore and univore to move from a binary understanding of consumption behaviour, towards a comprehensive and multidimensional understanding of the omnivore/univore theory for a given context. Overall, it was concluded that consumption behaviour of leisure is largely still bound up by social hierarchies. The final section of this chapter reviewed empirical findings on the motivation for attendance of leisure activities. As noted by Snowball *et al.* (2009), this is a topic closely aligned to the omnivore/univore hypothesis. Numerous factors were identified as motives for attendance. Overall, it was concluded that "motivations were found to be as important in determining consumption patterns as socio-demographic variables" (Snowball *et al.*, 2009:482).

With sport being a domain of extensive ramifications for a country's economy, society and culture, it was concluded that further research and development of the application of consumption behaviour theory to the sport context is necessary.

CHAPTER 4: METHODS AND DATA

4.1 Introduction

Drawing from previous studies reviewed in Chapter 3, this chapter provides insight into the methods employed and data collected in the study. Section 4.2 provides a discussion of the various definitions and classifications of sporting univores and omnivores to be used in this study. Section 4.3 discusses the econometric methods that were adopted in this study, which includes a model specification and a description of the variables. Section 4.4 provides an overview of the data used in this study. A discussion of the design of the questionnaire is presented, in which an explicit rationale for the choice of the included questions is given. Included in this section is a discussion of the sampling strategies that were adopted for the study and an overview of the administration of the questionnaire. Section 4.5 then provides an overview and discussion of the summary statistics. Finally, Section 4.6 summarises the chapter.

4.2. Defining sporting omnivores and univores

As noted by Widdop (2010), consumption studies to date have considered the classification of consumers on the ground of their social hierarchies. Bourdieu (1984) highlighted that sport is an object of struggles between social classes. This study investigates the classification of consumers' behaviour in terms of social hierarchy in accordance with what was proposed by Widdop (2010:3) where he states that:

"Certain types of sporting activities such as the golf, racket sports, water and winter sports (classified as highbrow or legitimate culture) would be consecrated among those in the higher classes, whilst others, such as football, tenpin bowling and weightlifting would be avoided through the association with the masses".

This is used in conjunction with Warde (2006), where exclusive sports included racket sports, water sports and adventure sports in the adaptation of the conventional definition in this context. To account for the two dominant forms of sports consumption, two separate conventional definitions were derived. According to Widdop and Cutts (2013), it is important to look at reported tastes as well as actual spectatorship behaviour in order to get a comprehensive overview of the cultural genre (or in this case sporting code). Definitions were derived for consumption in the form of attendance and consumption in the form of television

viewing respectively. In order to investigate the consumption of sport in relation to social hierarchies, a suitable definition² of omnivore and univore was developed, drawing from the work of both Widdop (2010) and Warde (2006).

The first of these conventional definitions specifically focused on consumption of sport in terms of watching the sport in attendance. For this definition³, consumers were classified as univores if they had exclusive sports consumption patterns. The classification of exclusivity in consumption was twofold. Firstly, consumers were assumed to have exclusive consumption behaviour if they attended high-brow/legitimate culture sports (otherwise referred to as elitist or exclusive sports) to the exclusion of other sports that were associated with the masses. Secondly, those consumers who chose to consume only sports that were associated with the masses (to the exclusion of the high-brow/legitimate sports) were also classified as having exclusive tastes (and therefore deemed univores). Conversely, consumers were classified as omnivores if they were found to straddle both consumption classifications. In other words, consumers were classified as omnivores if they displayed a wide array of consumption behaviour, or if they attended both high-brow/legitimate sports and sports that are associated with the masses (otherwise referred to as non-elitist sports). This definition was largely underpinned by Bourdieu's (1984) theory of the appreciation of high and low cultural forms; where high cultural forms were more likely to be appreciated by individuals from high income and high education groups, while low cultural forms (or those that were associated with the masses) were likely to be appreciated by individuals from lower education and income backgrounds. Drawing from both Widdop (2010) and Warde (2006) and by taking careful consideration of the extent to which consumers would have access to consumption (e.g. the extent to which the sport is broadcast on television or the extent to which the venues are able to accommodate large numbers of spectators); the following sports⁴ were classified as high-brow/legitimate culture sports: golf, tennis, equestrianism, swimming, motor-sport, rowing, cycling, water polo, and squash. With the exception of equestrianism and golf, each of these sports fell under either a water, racket or adventure sport. Golf was specifically classified by Widdop (2010) as a high-brow/legitimate sport and was therefore classified as such in the derivation of this definition. For this study, equestrianism was classified as a highbrow/legitimate culture sport. This classification is supported by The Economist (2014:1), in that "equestrianism is considered one of the more elitist sports".

² Hereafter referred to as the conventional definition

³ Hereafter referred to as 'Conventional definition-live'

⁴ Of the sports that the spectators identified as the sports that they choose to watch in attendance

Furthermore, because other sports have not been empirically classified as highbrow/lowbrow, a common ground of classification was needed for the derivation of these classifications. This study made use of structural constraints as a common ground of classification. According to Kim and Trail (2010:192), structural constraints are "factors that interfered between leisure preferences and participation, that is, physical or environmental factors that prevented an individual from leisure participation". For the purpose of this study, if the venue at which the sport event is generally hosted did not allow for/accommodate large amounts of spectators then the sport was said to exclude the masses and subsequently classified as elitist in nature⁵. Following this line of argument, the remainder of the spectator sports that were identified were classified as lowbrow/associated with the masses, and these included soccer, rugby, hockey, and netball.

The second conventional definition⁶ was derived using the same principles as the first conventional definition (i.e. informed by empirical classifications, and where none are evident the remainder of the sports were classified on common grounds). This definition differed from the first definition in that it looked only at consumption in the form of watching television. As with the previous definition, this definition was derived by classifying the various sports (that spectators had identified as the ones that they watch on television) into high-brow/legitimate sports and sports associated with the masses respectively. In this case, the following sports were classified as high-brow/legitimate culture sports: golf, tennis, equestrianism, swimming, multi-sport events (particularly The Winter Olympic Games), extreme sport, cycling, motorsport, badminton, squash, shooting, and fishing/surfing. The same argument as applied in the Conventional definition-live was used to support the classification of golf and equestrianism as high-brow/legitimate culture sports. Similarly, tennis, badminton and squash were considered high-brow/legitimate culture sports owing to their previous empirical classification as racket sports (Widdop, 2010). Extreme sport and motor sport were classified as what Warde (2006) refers to as 'adventure sports', while swimming, fishing and surfing were classified as watersports. The multi-sport event (in this study particularly referring to The Winter Olympic Games) is comprised of an array of different winter sports (Olympic.org, 2013), and as such, watching multi-sport events was classified as a high-brow/legitimate culture sport in accordance with the classification as identified by Widdop (2010). In terms of

⁵ For example, a sport such as soccer is generally hosted at large stadia (therefore having the capacity to accommodate masses of spectators), whereas a sport like shooting is generally hosted at relatively smaller facilities where the capacity/structural constraint of the venue would not allow for masses to attend.

⁶ Hereafter referred to as 'Conventional definition-TV'.

cycling, Beckett (2014:1) argued that "TV coverage is weak, sporadic and uneven in its global reach". Based on this, and in conjunction with Kim and Trail's (2010) structural constraints argument, cycling was classified as a highbrow/legitimate sport.

In order to classify the sports that have not been previously empirically classified, structural constraints (Kim and Trail, 2010) were again used as the basis of classification. In this case, sports were classified based on the extent to which they are broadcast on television in South Africa. Based on the broadcasting schedules available to television viewers in South Africa (Supersport, 2014; SABC Sport, 2014; ETV, 2014) the following sports (excluding sports that have been previously empirically classified) are broadcast: rugby, cricket, soccer, basketball, volleyball, netball, athletics, hockey and wrestling/boxing/mixed martial arts. These sports were therefore classified as sports associated with the masses, as they were available to television viewers. The remaining sports were classified as highbrow/legitimate sports on the grounds of structural constraints.

Similarly, under this definition, consumers were classified as univores if they had exclusive television consumption patterns. In other words, consumers received a univore classification if they watched only high-brow/legitimate culture sports on television (otherwise referred to as elitist or exclusive sports) to the exclusion of the sports that were associated with the masses, or if they watched only sport that was associated with the masses to the exclusion of the high-brow/legitimate culture sports. Conversely, consumers were classified as omnivores if they watched sport on television that was associated with the masses and high-brow/legitimate culture sports (i.e. if they were found to be straddling both consumption classifications).

As highlighted in Chapter 3, sports consumption, particularly in relation to empirical quantitative research, remains a largely underdeveloped field of research (Widdop, 2010). As such, there is need for researchers to explore aspects of research beyond the existing empirical work. As highlighted by empirical research to date (for example, Alderson *et al.*, 2007; Chan and Goldthorpe, 2007; Snowball *et al.*, 2009; Antrobus and Snowball, 2010) there is a need for research to explore new/sub classifications of the omnivore/univore classification, to provide a representational account for variations in consumption behaviour. Thus, in addition to the conventional definitions (as discussed above), this study considers other possible definitions/classifications of omnivore/univore that can be derived for the sporting context.

This study proposes four possible alternative indicators of sporting omnivores/univores. The first of these alternate definitions⁷ pertains to the level at which the sport is played. The development of this definition is largely informed by Montgomery and Robinson (2008) where they noted the importance of differentiating between professional and amateur sport in examining consumption behaviour of art and non-art events. For this definition, spectators are classified as omnivores if they watch both professional and amateur cricket. However, if they only watch professional cricket, such spectators would be classified as univores. This definition pertains to Question 24 (see Appendix A) where spectators were asked whether or not they watch both professional and amateur cricket. This question was recoded so that strongly agree and agree were taken as a proxy for 'yes', and strongly disagree and disagree were taken as a proxy for 'no'. Spectators who responded with 'yes' for this question were classified as omnivores, whereas a response of 'no' meant that the consumers were univores.

The second alternate definition of omnivore and univore⁸ considers whether spectators watch only cricket live (i.e. in attendance) or choose to watch cricket and other sports live. This definition was based on Peterson's (1992) definition of omnivores and univores, where participants' consumption was classified in accordance to the number of forms of art they consumed. The derivation of this definition was also largely informed by the work of Snowball *et al.* $(2009)^9$, who defined omnivores and univores in the South African arts context. In addition to this insight, and particularly in the sporting context, the second definition was further informed by Lefevre and Ohl (2011:49)¹⁰, who considered omnivority based on the principle of multi-activity in sports participation, but adapted their understanding in accordance with the consumption form. Therefore, the definition derived for this sporting context defined univores as spectators who choose to only watch cricket live to the exclusion of other sport (in other words, these individuals were characterised as having a very narrow consumption demand for sport), while omnivores were those who choose to attend a wide range of sporting events (or those individuals that were considered to be general sports followers). This definition pertained to Question 2 and Question 6 (of Appendix A) where spectators were asked whether cricket is the only sport that they choose to watch live and whether or not they attend other professional sports matches. Spectators who indicated that

⁷ Hereafter referred to as Definition 1.

⁸ Hereafter referred to as Definition 2.

⁹ Snowball *et al.* (2009) defined univores as patrons with narrower cultural tastes (attending only one modern genre or two traditional genres of art) while omnivorous patrons were those that consumed a wider array (four or more types of art genres) of both popular and high cultural goods.

¹⁰ Lefevre and Ohl (2011) classified individuals according to number of sporting activities that they participated in; where participation in two or three activities was indicative of low omnivority, whereas participation in four or more activities was indicative of high omnivority.

they only watch cricket live and who indicated that they do not choose to attend other professional sports matches (in other words suggesting exclusive consumption behaviour) were considered to be univores. Conversely, spectators who indicated that cricket is not the only sport that they choose to watch live and who indicated that they do attend other professional sports matches were considered to be omnivores (conversely, indicative of a wide/inclusive consumption behaviour). As with the previous definition, 'strongly agree' and 'agree' were taken as a proxy for 'yes', and 'strongly disagree' and 'disagree' were taken as a proxy for 'no'. Spectators who responded with 'yes' to Question 2 and 'no' to Question 6 were classified as univores, whereas a response of 'no' to Question 2 and 'yes' to Question 6 meant that the consumers were omnivores.

The third alternate classification of omnivores and univores¹¹ is very similarly to Definition 2. but considers the consumption of sport in the form of watching television as opposed to live/in attendance. The derivation of this definition was similarly based on the notion of narrow/exclusive and wider/inclusive tastes as used by Peterson (1992) and Snowball et al. (2009). For this definition, participants were classified as univores if cricket was the only sport that they watched on television, to the exclusion of other sports. Furthermore, participants who chose to consume only other sports (to the exclusion of cricket) were classified as univores owing to the exclusivity of their taste spectrum. Omnivores, on the other hand, were those participants that chose to watch a wide range of sport on television (indicating that they are general sports fans or had a wide taste spectrum). Definition 3 pertained to Question 25 and Question b (Appendix A). In Question 25, spectators were asked whether cricket is the only sport that they choose to watch on television. For this question, responses of 'strongly agree' and 'agree' were taken as a proxy for 'yes' while 'strongly disagree' and 'disagree' were taken as a proxy for 'no'. Question b (in which the response form was 'yes' or 'no') asked spectators whether they watch cricket on television. Spectators who responded with 'yes' to Question 25 (in other words those spectators who stated that cricket was the only sport that they watch on television) and yes to Question b were classified as univores; or 'agree' or 'strongly agree' to Question 15 (where participants were asked whether they watched sports channels on television) and 'no' to Question b were too classified as univores. Conversely, spectators who answered 'no' to Question 25 and either 'yes' or 'no' to Question b (indicating that they either watch cricket and other sports on television or they just watch other sports on television) were classified as omnivores.

¹¹ Hereafter referred to as Definition 3.

The final proposed classification of omnivore and univore¹² was again based on the spectrum of spectators' tastes, but considered whether the spectators classified themselves as general sports fans (displaying a relatively wide taste spectrum) or only cricket fans (displaying a relatively narrow taste spectrum). This classification pertained to Question 9 (Appendix A) where spectators were asked whether they consider themselves a sports fan as opposed to solely a cricket fan. This definition encompassed all forms of sports consumption. As was done for definition 1 to 3, 'strongly agree' and 'agree' were taken as a proxy for 'yes' while 'strongly disagree' and 'disagree' were taken as a proxy for 'no'. A response of 'yes' implied that spectators considered themselves to be general sports fans. This meant that spectators were characterized as having a relatively wide spectrum of tastes and as a result were classified as omnivores. Spectators who responded 'no' to Question 9 considered themselves to be only cricket fans, suggesting a relatively narrow/exclusive spectrum of tastes and as such were classified as univores.

The above four alternate classifications of omnivore and univore are summarized in Table 4.1.

Response to Question	Classification	Definition
If 'yes' to Question 24	Univore	1
If 'no' to Question 24	Omnivore	1
If 'yes' to Question 2 and 'no' to Question 6	Univore	2
If 'no' to Question 2 and 'yes' to Question 6	Omnivore	2
If 'yes' to Question 25 and 'yes' to Question b	Univore	3
If 'yes' to Question 15 and 'no' to Question b	Univore	3
If 'no' to Question 25 and either 'yes' or 'no' to Question b	Omnivore	3
If 'no' to Question 9	Univore	4
If 'yes' to Question 9	Omnivore	4

Table 4.1: Summary of alternate univore/omnivore classifications

¹² Hereafter referred to as Definition 4

4.3 Method

This section presents the model specification as well as a description of the variables included.

4.3.1 Model specification

Since the dependent variables are all in binary form, binary probit models were estimated for the conventional definitions as well as for all four alternate omnivore/univore classifications. This was done to determine the probability of someone being an omnivore or univore, given specific individual characteristics.

For each definition *i*, it is assumed that individuals are either classified as univores (*definition*_{*i*}=1) or omnivores (*definition*_{*i*}=0). The model is specified as follows:

 $Pr(definition_i = 1 | Xi) = \Phi(X'\beta)$

where Φ is the cumulative normal distribution function, β is a vector of parameters, and the probability of being classified as a univore depends on a vector of explanatory variables X'.

The general model that is estimated takes the form:

$$\Pr(definition_i = 1 \mid X_i) = \beta_0 + \beta_1 race_i + \beta_2 gender_i + \beta_3 age_i + \beta_4 education_i + \beta_5 income_i + u_i$$

Where u_i is the error term, and all other variables are discussed in the following section. All regressions were estimated using White heteroscedasticity-robust standard errors.

4.3.2 Variable description

Based on existing literature (for example, Snowball *et al.*, 2009; Antrobus and Snowball, 2010; Chan and Goldthorpe, 2007; Widdop and Cutts, 2013) the variables included in the study were age, gender, race, highest level of education and personal monthly net income. The variable 'age' was measured in years, which consisted of the categories of 11-20, 21-30, 31-40, 41-50, and 51 and above. The gender variable was coded as 1 if a spectator was male

and 0 for a female spectator. The race variable included 'Black', 'White', 'Coloured¹³' and 'Indian'. The education variable denotes the highest level of education obtained and consisted of 'primary education', 'secondary education (but no matric)', 'matric', 'and post-secondary education'. Finally, personal monthly net income was measured in South African Rands and represented the spectators' personal monthly income after tax and other statutory deductions. The categories for personal monthly net income were $R0 - R20\ 000$, $R20\ 001 - R40\ 000$, and $R40\ 001+$.

4.4 Data

This sub-section provides insight into the data that was obtained for this study. The section considers how the questionnaire that was used to obtain the data was developed, and discusses various aspects pertaining to the administration of the questionnaire.

4.4.1 Overview of the data and questionnaire composition

As noted by Lefevre and Ohl (2011:46), "(a) questionnaire survey provides a convenient tool for studying the social distribution of sports practices". The study made use of a questionnaire to obtain a range of responses reflecting specific types of consumption behaviour as well as motives for consumption decisions of cricket spectators in the Eastern Cape. The questionnaire was comprised of 25 closed-ended questions with a Likert scale response format. Response options for these questions ranged from 'strongly agree' to 'strongly disagree'. The questionnaire was further comprised of five open-ended questions that were designed to generate qualitative data pertaining to spectators' perceptions of their consumption behaviour. Finally, five demographic questions were designed and included in the questionnaire, providing a basis upon which consumers could be classified.

The questionnaire was developed to elicit responses pertaining to the two elements of the study. To elicit responses regarding motives for specific consumption decisions, questions were developed that were largely informed by the work of Trail and James (2001) and the Centre for Sport Consumer Research based in the U.S. This aspect of the study was also largely underpinned by the work done by Snowball *et al.* (2009) on motivation for attendance. To generate data pertaining to this aspect of the study, a combination of both closed and open-ended questions were developed. Referring to the questionnaire in Appendix A, questions 1,

¹³ The term 'Coloured' in South Africa pertains to people of mixed race origin (SAHO, 2014).

3, 4, 5, 8, 10, 11, 13, 14 16, 18, 19, 21, 22 and 23 as well as the open-ended question referring to the influencing factors of watching a cricket match live or on television were developed to determine the factors underpinning certain consumption behaviour decisions (or motives for attendance) that cricket spectators make.

To generate data pertaining to the classification of cricket spectators as sporting omnivores or univores, the questionnaire included questions that were positively and directly associated with consumption behaviour that would typically be characteristic of either a sporting omnivore or univore respectively. For the two conventional definitions, questions marked (d) and (e) of Appendix A were used to obtain data pertaining to the types of sport that consumers chose to consume live and on television respectively. Questions marked (g), (h), (i) and (j) of Appendix A were incorporated in order to obtain demographic data, to be able to make inferences about the type of consumer that would typically be associated with particular consumption behaviour.

The development of the alternate definitions was largely informed by the empirical study by Snowball *et al.* (2009), which similarly needed to construct relevant definitions/subclassifications of omnivore and univore for their given research context. Referring to the questionnaire, questions 2, 6, 9, 15, 24, and 25 of the closed-ended questions, as well as the question asking whether spectators watch cricket on television or not, were all structured to elicit responses pertaining to the alternative omnivore/univore classifications.

4.4.2 Sampling and administration

Data collection took place by means of a cross-sectional survey. The questionnaire was administered at four different first class limited-over cricket matches played by the Chevrolet Warriors at *Axxess DSL St Georges* in Port Elizabeth during the 2012/2013 cricket season¹⁴.

Purposive sampling was chosen as the sampling strategy to select participants for this study. As highlighted by Battaglia (2008), purposive sampling is a means of obtaining a sample, which is logically assumed to be representative of an entire population. In this instance, the selected sample was logically assumed to be representative of the cricket spectators in the

¹⁴ The matches were played on 7 November 2012, 18 November 2012, 23 November 2012 and 22 February 2013.

Eastern Cape that watch the limited-overs format of the game¹⁵. The motive for the choice of the sampling technique was underpinned by the Laerd dissertation (2013) where it was noted that purposive sampling is an efficient means of sample selection when the population size is relatively small and the intention of the study is to make accurate statistical inferences from a sample to a population of interest. Within purposive sampling, the study made use of a combination of maximum variation/heterogeneous sampling and critical case sampling techniques. The maximum variation/heterogeneous sampling technique is a manner in which a sample is selected, where the aim is to capture a diverse range of perspectives pertaining to the topic of interest (Laerd, 2013). This technique has been argued to exhibit a wide range of participant attributes, behaviours and experiences and was deemed appropriate for this study. Elements of critical case sampling techniques were also used to inform the sample selection. This technique was deemed appropriate for this study in that it has been noted to be particularly useful in research where the number of cases studied is relatively small (in this study four cases/matches were considered). It has further been noted that critical case sampling is beneficial in that, decisive explanations of phenomena and logical generalisations can be made from the chosen sample (Laerd, 2013) and thus proved a valuable and appropriate technique to use in this study.

In order to purposefully generate the sample required, each of the four interviewers were advised to specifically only interview individuals that possessed certain characteristics. For example, 'Interviewer 1' was instructed to interview only Black males and females, 'Interviewer 2' was advised to only interview White males and females, and so on. Owing to the limited number of spectators that attended these matches, the interviewers were able to conduct interviews with the majority (if not all) of the population that were willing to participate in the study.

Interviewers were briefed and trained prior to the fixtures on how best to illicit accurate spectator perspectives. The interviewers were specifically briefed on how they were to respectively select their respondents (with regards to the different dependent variables that were included in the study) as well as how to make a value judgement as to the reliability of the data collected. The questionnaire was developed in a manner in which the interviewers were guided and prompted throughout the interviews (this is explained in the prelude to Appendix A). Upon selection, the spectator was approached by the interviewer and asked

¹⁵ This specification was made owing to Date's (2012) comment that limited overs and test matches differ so significantly that they should essentially be viewed as separate sporting codes.

whether they would be willing to be interviewed for the study. All participants were explained the purpose of the study and were given a document (found as the first page of Appendix A) explaining the nature of the study and interview. Only after the spectator had agreed to be a participant in the study, did the interview commence. The duration of the administration of the interviews was approximately ten minutes per respondent (although this did vary in accordance with the lengths of discussions of topics presented in certain questions, but on average interviews ranged from between 5 and 15 minutes each).

The data collected from the four matches were then pooled to obtain the final data set, with a final sample size (excluding any interviews where data was deemed unreliable) of 438 respondents.

4.5 Summary statistics

An overview of the summary statistics of the demographics of the sample and the question responses are presented in the following sub-section.

4.5.1 Demographic statistics

Table 4.2 presents some basic summary statistics. Roughly, 71.69% of the sample was male, while the majority of the sample was White. The largest proportion of spectators was between the ages of 21 and 30. The table also shows that 47.26% of the spectators had a post-secondary education. This, if considered in conjunction with the age composition, could suggest that the majority of the sampled spectators were young professionals. Finally, the statistics revealed that the majority of the sampled spectators earned a monthly net income between R0 and R20 000.

Table 4.2: Summary	statistics
--------------------	------------

	N	%
Gender:		, u
Male	314	71 69
Female	124	28.31
Race:		
Black	91	20.78
White	2.2.9	52.28
Coloured	98	22.37
Indian	20	4.57
Age:		
No Response	3	0.68
11-20	92	21.00
21-30	145	33.11
31-40	62	14.16
41-50	67	15.30
51+	69	15.75
Highest level of education:		
No Response	4	0.91
Primary education	11	2.51
Secondary education (no matric)	69	15.76
Matric	147	33.56
Post-secondary	207	47.26
Personal Monthly Net Income:		
No Response	41	9.36
R0-R20 000	314	71.69
R20 001-R40 000	59	13.47
R40 001+	24	5.48

4.5.2 Question responses

_

Table 4.3 provides an overview of the responses to the questions pertaining to the motives for attendance and the omnivore/univore classification respectively. The results illustrate that 94.97% consider themselves fans of South African cricket. About 91.94% of the sample indicated that watching South African cricket makes them feel proudly South African. From these two indicators, the majority of spectators are likely to be intrinsically rather than extrinsically motivated to attend the fixtures. Questions 4 and 10 both indicate that the majority of the spectators are motivated to attend the cricket matches because of the sport in general as opposed to watching because a particular player or team is playing. It is also evident that the majority of the spectators consider uncertainty of outcome or good competitive balance as important in motivating their attendance. Finally, 90.62% indicated that their love for sport developed because of participating in sport at school; suggesting that there is a strong developmental aspect in sport.

Table 4.3	: Motives	for	attendance	(%)
-----------	-----------	-----	------------	-----

Ansatz DisagreeDisagree DisagreeDisagree DisagreeDisagree DisagreeDisagree DisagreeQ1: I consider myself a fan of South African cricket31.11[105][16][6][438]Q3: I play/have played cricket competitively before37.76 [165]25.86 [113]18.08 [79]18.30 [80]100.00 [437]Q4: I consider myself a fan of cricket as a whole and not just of a particular team55.15 [241]33.41 [146]9.84 [43]1.60 [177]100.00 [437]Q5: I attend cricket matches mainly because my friends or family are attending18.76 [82]25.17 [110]38.44 [168]17.63 [168]100.00 [437]Q8: The social experience is just as important as the cricket match game entertainment6.68 [196]18.43 [29]52.07 [160]22.82 [161]100.00 [434]Q10: I attend cricket matches because I game entertainment6.68 [29]18.43 [29]52.07 [20]22.82 [160]100.00 [434]Q13: I prefer close competition games where the outcome is uncertain.61.93 [270]26.61 [116]7.56 [33]3.90 [434]Q14: I attend the cricket matches as a way of meeting new people13.93 [143]31.28 [145]42.47 [137]12.33 [436]Q14: I attend the cricket allows me to feel part of the community13.93 [145]31.28 [103]23.67 [155]7.37 [32]100.00 [435]Q14: I attend the cricket allows me to feel part of the community11.61 [145]
Q1: I consider myself a fan of South African cricket 71.00 [311] 23.97 [105] 3.65 [16] 1.37 [6] 100.00 [438]Q3: I play/have played cricket competitively before 37.76 [165] 25.86 [113] 18.08 [79] 18.30 [80] 100.00 [437]Q4: I consider myself a fan of cricket as a whole and not just of a particular team (241] 55.15 [146] 33.41 [443] 9.84 [71] 1.60 [437]Q5: I attend cricket matches mainly because my friends or family are attending important as the cricket match matches match 18.76 [196] 25.17 [148] 38.44 [77] 17.63 [437] 100.00 [437]Q8: The social experience is just as important as the cricket matche players 18.76 [29] 25.07 [148] 22.82 [16] 100.00 [434]Q10: I attend cricket matches because I man specifically a fan of one/some of the players 18.20 [29] 35.02 [20] 37.79 [99] 8.99 [434]Q11: I enjoy attending the before and after me entertainment 18.20 [270] 35.02 [116] 37.79 [33] 8.99 [17] 100.00 [434]Q13: I prefer close competition games where the outcome is uncertain. 61.93 [270] 26.61 [116] 7.56 [33] 3.90 [17] 100.00 [436]Q14: I attend the cricket matches as a way of meeting new people 33.33 [161] 23.67 [137] 35.63 [136] 7.37 [133] 100.00 [435]Q14: I attend the cricket allows me to feel part of the comm
African cricket[105][105][105][105][106][1438]Q3: I play/have played cricket37.7625.8618.0818.30100.00competitively before[165][113][79][80][437]Q4: I consider myself a fan of cricket as a whole and not just of a particular team55.1533.419.841.60100.00Q5: I attend cricket matches mainly because my friends or family are attending18.7625.1738.4417.63100.00g8: The social experience is just as important as the cricket match44.9533.9417.433.68100.00[106][148][76][16][436][436]Q10: I attend cricket matche[196][148][776][16][434]players[29][80][226][99][434]Q11: I enjoy attending the before and after game entertainment[79][152]37.798.99100.00[210][211][16][33][17][436]Q13: I prefer close competition games where the outcome is uncertain.[270][16][33][17][436]Q14: I attend the cricket matches as a way of meeting new people[61][137][186][54][438]Q14: I attend the cricket atlows me to feel art of the community[161][137][163][132][435]Q18: Watching cricket allows me to feel part of the community[161][172][72][10][435]Q19: Watching South Af
Q3: I play/have played cricket $[1,17]$ $[1,06]$ $[1,07]$ $[1,06]$ $[1,06]$ Q3: I play/have played cricket 37.76 25.86 18.08 18.30 100.00 competitively before $[165]$ $[113]$ $[79]$ $[80]$ $[437]$ Q4: I consider myself a fan of cricket as a whole and not just of a particular team 55.15 33.41 9.84 1.60 100.00 Q5: I attend cricket matches mainly because my friends or family are attending 18.76 25.17 38.44 17.63 100.00 Q8: The social experience is just as important as the cricket match 44.95 33.94 17.43 3.68 100.00 Q10: I attend cricket matches because I an specifically a fan of one/some of the players 6.68 18.43 52.07 22.82 100.00 Q11: I enjoy attending the before and after game entertainment 18.20 35.02 37.79 8.99 100.00 Q13: I prefer close competition games of meeting new people 61.93 26.61 7.56 3.90 100.00 Q14: I attend the cricket matches as a way watch live 33.33 23.67 35.63 7.37 100.00 Q14: I attend the cricket allows me to feel art of the community 11.61 39.54 16.55 2.30 100.00 Q14: Statend the cricket allows me to feel art of the community 11.61 39.54 16.55 2.30 100.00 Q14: I attend the cricket allows me to feel art of the community 11.61 39.54 $16.$
Q3: I play/have played cricket competitively before37.76 [165]25.86 [113]18.08 [79]18.30 [80]100.00 [437]Q4: I consider myself a fan of cricket as a whole and not just of a particular team55.15 [241]33.41 [146]9.84 [43]1.60 [71]100.00 [437]Q5: I attend cricket matches mainly because my friends or family are attending18.76 [82]25.17 [110]38.44 [168]17.63 [77]100.00 [437]Q8: The social experience is just as important as the cricket match44.95 [196]33.94 [148]17.43 [76]3.68 [161]100.00 [436]Q10: I attend cricket matchs because I am specifically a fan of one/some of the players6.68 [29]18.43 [80]52.07 [226]22.82 [99]100.00 [434]Q11: I enjoy attending the before and after game entertainment18.20 [79]35.02 [161]37.79 [164]8.99 [39]100.00 [434]Q13: I prefer close competition games where the outcome is uncertain.61.93 [270]26.61 [116]7.56 [33] [131]3.00 [1436]Q14: I attend the cricket matches as a way watch live13.93 [145]31.28 [132]42.47 [145]12.33 [133]100.00 [435]Q14: I whole the tricket allows me to feel part of the community11.61 [145]39.54 [162]16.55 [32]2.30 [435]Q18: Watching cricket allows me to feel part of the community11.61 [181]39.54 [172]16.55 [72]2.30 [100.00 [435]
Competitively before[165][113][79][80][437]Q4: I consider myself a fan of cricket as a whole and not just of a particular team[241][146][43][7][437]Q5: I attend cricket matches mainly because my friends or family are attending18.7625.17 38.44 17.63100.00Q8: The social experience is just as important as the cricket match[196][148][76][16][437]Q8: The social experience is just as important as the cricket match44.95 33.94 17.43 3.68 100.00Q10: I attend cricket matches because I players6.6818.4352.0722.82100.00Q11: I enjoy attending the before and after game entertainment18.20 35.02 37.79 8.99 100.00Q13: I prefer close competition games where the outcome is uncertain. 61.93 26.61 7.56 3.90 100.00Q14: I attend the cricket matches as a way watch live 13.93 31.28 42.47 12.33 100.00Q14: I attend the cricket matches as a way watch live 13.93 23.67 35.63 7.37 100.00Q14: I attend the cricket allows me to feel part of the community $(161]$ $(172]$ $[163]$ $[143]$ Q16: Cricket is my favourite sport to part of the community 33.33 23.67 35.63 7.37 100.00 Q18: Watching cricket allows me to feel part of the community $[181]$ $[172]$ $[103]$ $[162]$ $[1435]$ Q19: Watch
Q4: I consider myself a fan of cricket as a whole and not just of a particular team $[245]$ $[115]$ $[75]$ $[60]$ $[100]$ Q5: I attend cricket matches mainly because my friends or family are attending18.76 [82]25.17 [110]38.44 [168]17.63 [77]100.00 [437]Q8: The social experience is just as important as the cricket match44.95 [196]33.94 [148]17.43 [76]3.68 [16]100.00 [436]Q10: I attend cricket match[196][148] [29][76] [148][16] [436][436]Q10: I attend cricket match6.68 [29]18.43 [20]52.07 [22.82]22.82 [100.00 [434]Q11: I enjoy attending the before and after game entertainment18.20 [79]35.02 [152]37.79 [164]8.99 [39]100.00 [434]Q13: I prefer close competition games where the outcome is uncertain.61.93 [270]26.61 [116]7.56 [33]3.90 [177]100.00 [436]Q14: I attend the cricket matches as a way of meeting new people33.33 [217]23.67 [139]35.63 [321]7.37 [438]Q16: Cricket is my favourite sport to part of the community33.33 [145]23.67 [155]35.63 [32]7.37 [435]Q18: Watching cricket allows me to feel part of the community[181] [172]16.55 [72]2.30 [100]100.00 [435]Q19: Watching South African cricket for the for the formative64.52 [727]27.42 [72]5.99 [70]2.07 [70]
Q4: I consider myself a fan of cricket as a whole and not just of a particular team 55.15 $[241]$ 33.41 $[146]$ 9.84 $[43]$ 1.60 $[7]$ 100.00 $[437]$ Q5: I attend cricket matches mainly because my friends or family are attending 18.76 $[82]$ 25.17 $[110]$ 38.44 $[168]$ 17.63 $[77]$ 100.00 $[437]$ Q8: The social experience is just as important as the cricket match 44.95 $[196]$ 33.94 $[148]$ 17.43 $[76]$ 3.68 $[16]$ 100.00 $[436]$ Q10: I attend cricket matches because I am specifically a fan of one/some of the players 6.68 $[29]$ 18.43 $[80]$ 52.07 $[226]$ 22.82 $[99]$ 100.00 $[434]$ Q11: I enjoy attending the before and after game entertainment 18.20 $[79]$ 35.02 $[152]$ 37.79 $[164]$ 8.99 $[39]$ 100.00 $[434]$ Q13: I prefer close competition games where the outcome is uncertain. 61.93 $[270]$ 26.61 $[116]$ 7.56 $[33]$ 3.90 $[17]$ 100.00 $[436]$ Q14: I attend the cricket matches as a way of meeting new people 33.33 $[61]$ 23.67 $[137]$ 35.63 $[133]$ 7.37 $[100.00$ $[1435]$ Q16: Cricket is my favourite sport to watch live 33.33 $[145]$ 23.67 $[103]$ 35.63 $[155]$ 7.37 $[32]$ 100.00 $[435]$ Q18: Watching cricket allows me to feel part of the community 41.61 $[181]$ 39.54 $[172]$ 16.55 $[20]$ 2.30 $[100]$
whole and not just of a particular team[241][146][43][7][437]Q5: I attend cricket matches mainly because my friends or family are attending 18.76 25.17 38.44 17.63 100.00 Q8: The social experience is just as important as the cricket match 44.95 33.94 17.43 3.68 100.00 Q10: I attend cricket match $[196]$ $[148]$ $[76]$ $[16]$ $[143]$ Q10: I attend cricket match 6.68 18.43 52.07 22.82 100.00 am specifically a fan of one/some of the players 6.68 18.43 52.07 22.82 100.00 Q11: I enjoy attending the before and after game entertainment 18.20 35.02 37.79 8.99 100.00 Q13: I prefer close competition games where the outcome is uncertain. 61.93 26.61 7.56 3.90 100.00 Q14: I attend the cricket matches as a way of meeting new people 33.33 23.67 35.63 7.37 100.00 Q16: Cricket is my favourite sport to watch live 33.33 23.67 35.63 7.37 100.00 Q18: Watching cricket allows me to feel part of the community 11.61 39.54 16.55 2.30 100.00 Q19: Watching South African cricket (64.52) 64.52 27.42 5.99 2.07 100.00
Where the use for the particular term $[211]$ $[110]$ $[110]$ $[110]$ $[111]$ $[111]$ $[110]$ Q5: I attend cricket matches mainly because my friends or family are attending18.7625.1738.4417.63100.00Q8: The social experience is just as important as the cricket match44.9533.9417.433.68100.00Q10: I attend cricket match[196][148][76][16][436]Q10: I attend cricket matches because I am specifically a fan of one/some of the game entertainment6.6818.4352.0722.82100.00Q11: I enjoy attending the before and after game entertainment18.2035.0237.798.99100.00Q13: I prefer close competition games where the outcome is uncertain.61.9326.617.563.90100.00Q14: I attend the cricket matches as a way of meeting new people13.9331.2842.4712.33100.00Q14: I attend the cricket matches as a way watch live13.9323.6735.637.37100.00Q16: Cricket is my favourite sport to watch live33.3323.6735.637.37100.00Q18: Watching cricket allows me to feel part of the community11.61172][72][10][435]Q19: Watching South African cricket for an to the community64.5227.425.992.07100.00
Q5: I attend cricket matches mainly because my friends or family are attending18.76 [82]25.17 [110]38.44 [168]17.63 [77]100.00 [437]Q8: The social experience is just as important as the cricket match44.95 [196]33.94 [148]17.43 [76]3.68 [16]100.00 [436]Q10: I attend cricket match6.68 [29]18.43 [29]52.07 [80]22.82 [99]100.00 [434]Q11: I enjoy attending the before and after game entertainment18.20 [79]35.02 [152]37.79 [164]8.99 [39]100.00 [434]Q13: I prefer close competition games where the outcome is uncertain.61.93 [270]26.61 [116]7.56 [33]3.90 [171]100.00 [436]Q14: I attend the cricket matches as a way of meeting new people13.93 [61]31.28 [137]42.47 [186]12.33 [438]Q16: Cricket is my favourite sport to watch live33.33 [145]23.67 [155]35.63 [32]7.37 [100.00 [435]Q18: Watching cricket allows me to feel part of the community41.61 [181]39.54 [172]16.55 [27]2.30 [10]100.00 [435]Q19: Watching South African cricket for the community64.52 [27,42]27.42 [59]5.99 [2.07]2.07 [00.00 [435]
because my friends or family are attending[82][110][168][77][437]Q8: The social experience is just as important as the cricket match 44.95 [196] 33.94 [148] 17.43 [76] 3.68 [16] 100.00 [436]Q10: I attend cricket matches because I am specifically a fan of one/some of the players 6.68 [29] 18.43 [20] 52.07 [226] 22.82 [99] 100.00 [434]Q11: I enjoy attending the before and after game entertainment 18.20 [79] 35.02 [152] 37.79 [164] 8.99 [39] 100.00 [434]Q13: I prefer close competition games where the outcome is uncertain. 61.93 [270] 26.61 [116] 7.56 [33] 3.90 [171] 100.00 [436]Q14: I attend the cricket matches as a way of meeting new people 13.93 [61] 31.28 [137] 42.47 [186] 12.33 [54] 100.00 [438]Q16: Cricket is my favourite sport to watch live 33.33 [145] 23.67 [145] 35.63 [103] 7.37 [122] 100.00 [435]Q18: Watching cricket allows me to feel part of the community 11.61 [181] 39.54 [172] 16.55 [230] 2.30 [100]Q19: Watching South African cricket for the community 64.52 [181] 27.42 [192] 5.99 [207] 2.07 [100]
Q8: The social experience is just as important as the cricket match 44.95 [196] 33.94 [148] 17.43 [76] 3.68 [16] 100.00 [436]Q10: I attend cricket matches because I am specifically a fan of one/some of the players 6.68 [29] 18.43 [29] 52.07 [226] 22.82 [99] 100.00 [434]Q11: I enjoy attending the before and after game entertainment 18.20 [79] 35.02 [152] 37.79 [164] 8.99 [39] 100.00 [434]Q13: I prefer close competition games where the outcome is uncertain. 61.93 [270] 26.61 [116] 7.56 [33] 3.90 [171] 100.00 [436]Q14: I attend the cricket matches as a way of meeting new people 13.93 [61] 31.28 [137] 42.47 [186] 12.33 [54] 100.00 [438]Q16: Cricket is my favourite sport to watch live 33.33 [145] 23.67 [103] 35.63 [155] 7.37 [32] 100.00 [435]Q18: Watching cricket allows me to feel part of the community 41.61 [181] 39.54 [172] 16.55 [72] 2.30 [100] 100.00 [435]Q19: Watching South African cricket bl out of the community 64.52 [27.42] 5.99 [20] 2.07 [10] 100.00 [435]
Q8: The social experience is just as important as the cricket match 44.95 [196] 33.94 [148] 17.43 [76] 3.68 [16] 100.00 [436]Q10: I attend cricket matches because I am specifically a fan of one/some of the players 6.68 [29] 18.43 [80] 52.07 [226] 22.82 [99] 100.00 [434]Q11: I enjoy attending the before and after game entertainment 18.20 [79] 35.02 [152] 37.79 [164] 8.99 [39] 100.00 [434]Q13: I prefer close competition games where the outcome is uncertain. 61.93 [270] 26.61 [116] 7.56 [33] 3.90 [171] 100.00 [436]Q14: I attend the cricket matches as a way of meeting new people 13.93 [61] 31.28 [137] 42.47 [186] 12.33 [54] 100.00 [438]Q16: Cricket is my favourite sport to watch live 33.33 [145] 23.67 [103] 35.63 [155] 7.37 [32] 100.00 [435]Q18: Watching cricket allows me to feel part of the community 41.61 [181] 39.54 [172] 16.55 [72] 2.30 [10] 100.00 [435]Q19: Watching South African cricket be use for be well 0.64 bit form (16.20) 64.52 (27.42) 5.99 (20) 2.07 (100.00)
Construct of the relation of the second state of the construction of the communityInterval of the communityInterval of the communityInterval of the communityimportant as the cricket match $[196]$ $[148]$ $[76]$ $[16]$ $[1436]$ Q10: I attend cricket matches because I am specifically a fan of one/some of the players 6.68 18.43 52.07 22.82 100.00 Q11: I enjoy attending the before and after players $[29]$ $[80]$ $[226]$ $[99]$ $[434]$ Q13: I prefer close competition games where the outcome is uncertain. 61.93 26.61 7.56 3.90 100.00 Q14: I attend the cricket matches as a way of meeting new people $[61]$ $[137]$ $[186]$ $[54]$ $[438]$ Q16: Cricket is my favourite sport to watch live 33.33 23.67 35.63 7.37 100.00 Q18: Watching cricket allows me to feel 41.61 39.54 16.55 2.30 100.00 Q19: Watching South African cricket 64.52 27.42 5.99 2.07 100.00
Important as the effect intermination $[150]$ $[150]$ $[160]$ $[160]$ $[160]$ $[160]$ $[160]$ Q10: I attend cricket matches because I am specifically a fan of one/some of the players 6.68 18.43 52.07 22.82 100.00 Q11: I enjoy attending the before and after game entertainment 18.20 35.02 37.79 8.99 100.00 Q13: I prefer close competition games where the outcome is uncertain. 61.93 26.61 7.56 3.90 100.00 Q14: I attend the cricket matches as a way of meeting new people 13.93 31.28 42.47 12.33 100.00 Q14: I attend the cricket matches as a way of meeting new people 33.33 23.67 35.63 7.37 100.00 Q16: Cricket is my favourite sport to watch live 33.33 23.67 35.63 7.37 100.00 Q18: Watching cricket allows me to feel part of the community 41.61 39.54 16.55 2.30 100.00 Q19: Watching South African cricket part of the community 64.52 27.42 5.99 2.07 100.00
Q10: I attend cricket matches because I am specifically a fan of one/some of the players 6.68 $[29]$ 18.43 $[80]$ 52.07 $[226]$ 22.82 $[99]$ 100.00 $[434]$ Q11: I enjoy attending the before and after game entertainment 18.20 $[79]$ 35.02 $[152]$ 37.79 $[164]$ 8.99 $[39]$ 100.00 $[434]$ Q13: I prefer close competition games where the outcome is uncertain. 61.93 $[270]$ 26.61 $[116]$ 7.56 $[33]$ 3.90 $[17]$ 100.00 $[436]$ Q14: I attend the cricket matches as a way of meeting new people 13.93 $[61]$ 31.28 $[137]$ 42.47 $[186]$ 12.33 $[54]$ 100.00 $[438]$ Q16: Cricket is my favourite sport to watch live 33.33 $[145]$ 23.67 $[103]$ 35.63 $[155]$ 7.37 $[32]$ 100.00 $[435]$ Q18: Watching cricket allows me to feel part of the community 41.61 $[181]$ 39.54 $[172]$ 16.55 $[72]$ 2.30 $[100]$ 100.00 $[435]$ Q19: Watching South African cricket part of the community 64.52 $[27.42]$ 5.99 $[2.07]$ 2.07 $[100.00]$
am specifically a fan of one/some of the players[29][80][226][99][434]Q11: I enjoy attending the before and after game entertainment18.20 [79]35.02 [152]37.79 [164]8.99 [39]100.00 [434]Q13: I prefer close competition games where the outcome is uncertain. 61.93 [270]26.61 [116]7.56 [33]3.90 [17]100.00 [436]Q14: I attend the cricket matches as a way of meeting new people 61.93 [61]26.61 [137]7.56 [168]3.90 [17]100.00 [438]Q14: I attend the cricket matches as a way of meeting new people 13.93 [61] 31.28 [137] 42.47 [186] 12.33 [54]100.00 [438]Q16: Cricket is my favourite sport to watch live 33.33 [145] 23.67 [103] 35.63 [155] 7.37 [32]100.00 [435]Q18: Watching cricket allows me to feel part of the community 41.61 [181] 39.54 [172] 16.55 [72] 2.30 [10]100.00 [435]Q19: Watching South African cricket be use for lower
an operation of the orbit o
Q11: I enjoy attending the before and after game entertainment18.20 [79]35.02 [152]37.79 [164]8.99 [39]100.00 [434]Q13: I prefer close competition games where the outcome is uncertain.61.93 [270]26.61 [116]7.56 [33]3.90 [17]100.00 [436]Q14: I attend the cricket matches as a way of meeting new people13.93 [61]31.28 [137]42.47 [186]12.33 [54]100.00 [438]Q16: Cricket is my favourite sport to watch live33.33 [145]23.67 [155]35.63 [32]7.37 [435]Q18: Watching cricket allows me to feel part of the community41.61 [181]39.54 [172]16.55 [72]2.30 [10]100.00 [435]Q19: Watching South African cricket part of the community64.52 [27.42]27.42 [59]5.99 [2.07]2.07 [100.00
Q11: I enjoy attending the before and after game entertainment18.20 [79]35.02 [152]37.79 [164]8.99 [39]100.00 [434]Q13: I prefer close competition games where the outcome is uncertain.61.93 [270]26.61 [116]7.56 [33]3.90 [17]100.00 [436]Q14: I attend the cricket matches as a way of meeting new people13.93 [61]31.28 [137]42.47 [186]12.33 [54]100.00 [438]Q16: Cricket is my favourite sport to watch live33.33 [145]23.67 [155]35.63 [155]7.37 [32]100.00 [435]Q18: Watching cricket allows me to feel part of the community41.61 [181]39.54 [172]16.55 [72]2.30 [10]100.00 [435]Q19: Watching South African cricket part of the community64.52 [27.42]27.42 [110]5.99 [207]2.07 [100.00
Q11: Forly of attending into order and after 1020 5032 5032 5032 5033 10000 game entertainment $[79]$ $[152]$ $[164]$ $[39]$ $[434]$ Q13: I prefer close competition games 61.93 26.61 7.56 3.90 100.00 where the outcome is uncertain. $[270]$ $[116]$ $[33]$ $[17]$ $[436]$ Q14: I attend the cricket matches as a way of meeting new people 13.93 31.28 42.47 12.33 100.00 Q16: Cricket is my favourite sport to watch live 33.33 23.67 35.63 7.37 100.00 Q18: Watching cricket allows me to feel part of the community 41.61 39.54 16.55 2.30 100.00 Q19: Watching South African cricket or black of black
game entertainment $[101]$ $[101]$ $[101]$ $[101]$ $[101]$ $[101]$ Q13: I prefer close competition games where the outcome is uncertain. $[270]$ $[116]$ $[33]$ $[17]$ $[436]$ Q14: I attend the cricket matches as a way of meeting new people $[3.93]$ $[31.28]$ 42.47 12.33 100.00 Q16: Cricket is my favourite sport to watch live 33.33 23.67 35.63 7.37 100.00 Q18: Watching cricket allows me to feel part of the community 41.61 39.54 16.55 2.30 100.00 Q19: Watching South African cricket met of lange of the specific lange of t
Q13: I prefer close competition games where the outcome is uncertain. 61.93 $[270]$ 26.61 $[116]$ 7.56 $[33]$ 3.90 $[17]$ 100.00 $[436]$ Q14: I attend the cricket matches as a way of meeting new people 13.93 $[61]$ 31.28 $[137]$ 42.47 $[186]$ 12.33 $[54]$ 100.00 $[438]$ Q16: Cricket is my favourite sport to watch live 33.33 $[145]$ 23.67 $[103]$ 35.63 $[155]$ 7.37 $[32]$ 100.00 $[435]$ Q18: Watching cricket allows me to feel part of the community 41.61 $[181]$ 39.54 $[172]$ 16.55 $[72]$ 2.30 $[10]$ 100.00 $[435]$ Q19: Watching South African cricket means for long with African cricket $[64.52]$ 27.42 $[27.42]$ 5.99 $[207]$ 2.07 $[100.00]$
where the outcome is uncertain. $[270]$ $[116]$ $[33]$ $[17]$ $[436]$ Q14: I attend the cricket matches as a way of meeting new people 13.93 31.28 42.47 12.33 100.00 Q16: Cricket is my favourite sport to watch live 33.33 $[145]$ 23.67 35.63 $[103]$ 7.37 100.00 Q18: Watching cricket allows me to feel part of the community 41.61 $[181]$ 39.54 $[172]$ 16.55 $[72]$ 2.30 $[100]$ 100.00 $[435]$ Q19: Watching South African cricket means for loss of l
Q14: I attend the cricket matches as a way of meeting new people13.93 [61]31.28 [137]42.47 [186]12.33 [54]100.00 [438]Q16: Cricket is my favourite sport to watch live33.33 [145]23.67 [103]35.63 [155]7.37 [32]100.00 [435]Q18: Watching cricket allows me to feel part of the community41.61 [181]39.54 [172]16.55 [72]2.30 [10]100.00 [435]Q19: Watching South African cricket part of the community64.52 [182]27.42 [192]5.99 [207]2.07 [100.00 [123]
Q14: I attend the cricket matches as a way of meeting new people 13.93 [61] 31.28 [137] 42.47 [12.33 [100.00] Q16: Cricket is my favourite sport to watch live 33.33 [145] 23.67 [186] 155] [32] [438] Q18: Watching cricket allows me to feel part of the community 41.61 [172] 172] [10] 100.00 [435] Q19: Watching South African cricket 64.52 [172] 27.42 [5.99 [2.07 100.00] 100.00 [435]
of meeting new people [61] [137] [186] [54] [438] Q16: Cricket is my favourite sport to watch live 33.33 23.67 35.63 7.37 100.00 Q18: Watching cricket allows me to feel part of the community 41.61 39.54 16.55 2.30 100.00 Q19: Watching South African cricket 64.52 27.42 5.99 2.07 100.00
Q16: Cricket is my favourite sport to watch live 33.33 [145] 23.67 [103] 35.63 [155] 7.37 [32] 100.00 [435]Q18: Watching cricket allows me to feel
Q16: Cricket is my favourite sport to watch live 33.33 [145] 23.67 [103] 35.63 [155] 7.37 [32] 100.00 [435]Q18: Watching cricket allows me to feel part of the community 41.61 [181] 39.54 [172] 16.55 [72] 2.30 [10] 100.00 [435]Q19: Watching South African cricket me for large state
watch live [145] [103] [155] [32] [435] Q18: Watching cricket allows me to feel 41.61 39.54 16.55 2.30 100.00 part of the community [181] [172] [72] [10] [435] Q19: Watching South African cricket 64.52 27.42 5.99 2.07 100.00 watch and the first set of the set
Q18: Watching cricket allows me to feel 41.61 39.54 16.55 2.30 100.00 part of the community [181] [172] [72] [10] [435] Q19: Watching South African cricket 64.52 27.42 5.99 2.07 100.00 mathematical cricket 64.52 27.42 5.99 2.07 100.00
Q18: Watching cricket allows me to feel 41.61 39.54 16.55 2.30 100.00 part of the community [181] [172] [72] [10] [435] Q19: Watching South African cricket 64.52 27.42 5.99 2.07 100.00 watching South African cricket 64.52 27.42 5.99 2.07 100.00
part of the community [181] [172] [72] [10] [435] Q19: Watching South African cricket 64.52 27.42 5.99 2.07 100.00 The second secon
Q19: Watching South African cricket 64.52 27.42 5.99 2.07 100.00 Comparison of the second state of the
Q19: Watching South African cricket 64.52 27.42 5.99 2.07 100.00
makes me reer proudiv South African [280] [119] [26] [9] [434]
O21: I believe playing sport at school 59.50 31.12 7.09 2.29 100.00
developed my love for sport [260] [136] [31] [10] [437]
O22: If a certain player/some players 4.82 7.80 48.62 38.76 100.00
was/were not playing I would not attend [21] [34] [212] [169] [436]
Q23: Some of my family members 29.49 29.72 25.12 15.67 100.00
play/have played cricket competitively [128] [129] [109] [68] [434]

<u>Note</u>: Values in square brackets denote the sample size.

Table 4.4 presents an overview of the sample statistics that pertain to the alternate definitions/sub-classifications of omnivore and univore. These definitions as well as the conventional definition of omnivore and univore are discussed in Section 4.2.1.

Table 4.4. Omnivore/Onivore classification (76)					
	<u>Strongly</u>	<u>Agree</u>	Disagree	<u>Strongly</u>	<u>Total</u>
	<u>Agree</u>			<u>Disagree</u>	
Q24: I watch both professional and	34.40	43.58	15.14	6.88	100.00
amateur cricket	[150]	[190]	[66]	[30]	[436]
O2: Cricket is the only sport I choose to	9.64	10.55	44.72	35.09	100.00
watch live (in attendance)	[42]	[46]	[195]	[153]	[436]
Of From time to time I go to other	53 10	33.64	0.38	3.99	100.00
Q0. From time to time 1 go to other	55.10	55.0 4	9.30	5.00 [17]	[427]
professional sports matches	[232]	[14/]	[41]	[1/]	[437]
Q25: Cricket is the only sport I choose	3.44	4.14	43.22	49.20	100.00
to watch on TV	[15]	[18]	[188]	[214]	[435]
015: I watch sports channels on TV	59 59	32 42	5.02	2 74	100.00
Q13. 1 water sports chamiers on 1 v	[261]	[142]	[22]	[12]	[438]
	[201]	[1,2]	[]	[12]	[150]
Q9: I would consider myself as a sports	63.39	27.92	7.32	1.37	100.00
fan rather than just a cricket fan	[277]	[122]	[32]	[6]	[437]
-					

 Table 4.4: Omnivore/Univore classification (%)
 (%)

Note: Values in square brackets denote the sample size.

Table 4.4 indicates that 78% of the sample watched both professional and amateur cricket. The majority of the sample considered themselves sports fans as opposed to only a cricket fan, and indicated that cricket is not the only sport that they choose to watch on television and live respectively. Overall, the table suggests that across the alternate classifications, consumers display inclusive as opposed to exclusive consumption behaviour.

4.6 Summary and conclusion

This chapter described the methods and data that were used for the empirical analysis of this study. The chapter described the different ways in which cricket spectators were classified into omnivores and univores. The chapter also included an overview of the questionnaire, sampling and administration, and summary statistics. An analysis of how omnivores and univores differ according to socio-demographic characteristics could be used to determine whether consumption behaviour of cricket spectators supports the omnivore/univore hypothesis claims made by Bourdieu (1984) or whether the findings are in line with the counter-argument proposed by Peterson (1992). In conjunction with this, an analysis of how

the motives of attendance differ between omnivores and univores for a given definition may provide substantial insight into understanding sports consumption behaviour as well as prove useful in identifying aspects that could be drawn on by marketers to improve consumption demand for sport. Thus, the next chapter presents the empirical results that were obtained for each of the omnivore/univore definitions as well as the results pertaining to how the omnivores and univores differed in terms of their motives for attendance.

CHAPTER 5: RESULTS, DISCUSSION AND RECOMMENDATIONS

5.1 Introduction

This chapter provides the empirical results and interpretation thereof from the conventional and alternate omnivore/univore classifications as well as the results pertaining to the motivation for attendance. The results derived from the probit models for each definition are presented and interpreted. The chapter then presents the bivariate cross-tabulation results that suggest how omnivores and univores differ in terms of their motives for attendance; here a comparative approach for the classifications is adopted. Throughout the chapter, a discussion of the findings and recommendations stemming therefrom are presented.

The chapter is comprised of the following sections: Section 5.2 provides an overview of the results and interpretation for the omnivore/univore classification. Section 5.3 presents the probit regression results predicting the probability of a univore classification relative to an omnivore classification. This section is divided into two subsections. Section 5.3.1 provides the probit results for the conventional omnivore/univore definitions, while Section 5.3.2 provides the probit results for the four alternative definitions proposed in the study. Section 5.3.3 then goes on to present a comparative analysis of socio-demographic findings across both the conventional and the proposed alternative definitions. The results for the motivation for attendance are provided in Section 5.4. Finally, Section 5.5 summarises and concludes the chapter.

5.2 Omnivore/univore classifications

Table 5.1 indicates the proportion of the sample classified as univores and omnivores respectively for the different definitions.

	-	
	<u>N</u>	<u>%</u>
Conventional definition- Live		
Univore	30	6.85
Omnivore	408	93.15
<u>Conventional definition-TV</u>		
Univore	12	2.74
Omnivore	426	97.26
<u>Definition 1</u>		
Univore	98	22.37
Omnivore	340	77.63
Definition 2		
Univore	85	19.41
Omnivore	353	80.59
Definition 3		
Univore	37	8.45
Omnivore	401	91.55
Definition 4		
Univore	39	8.90
Omnivore	199	91.10

Table 5.1: Omnivore/univore classifications

From the omnivore/univore classification results, it appears that spectators were far more likely to be classified as omnivores than univores for all definitions. In terms of the overall results of the specific definitions, the vast majority of the sample was omnivores when spectators were classified according to the two conventional definitions. This finding suggests that for both forms of sports consumption (television and live), the majority of the spectators are more likely to have an inclusive sport taste spectrum as opposed to an exclusive taste spectrum. That is, spectators are more likely to consume sport across classifications (elitist sports and sport associated with the masses), as opposed to consuming sport from only one classification. In order to comment on whether or not sports consumption is tied up in social hierarchies, it is necessary to disaggregate the probability of a univore classification by all the explanatory variables that are employed in the study. These results are presented and discussed in Section 5.3.

The results that pertained to the alternate four definitions similarly revealed that the majority of the sample was omnivores as opposed to univores. A relatively greater proportion of the sample was classified as omnivores under the two conventional definitions than the four alternative definitions. Similarly, a much smaller proportion of the sample was classified as univores under the conventional definitions than under the alternative definitions. This suggests that spectators are more likely to be classified as a univore under any of the alternate definitions than under the conventional definitions. This could alternatively be interpreted as: the probability of being classified as a univore is significantly lower for the conventional definitions relative to the alternate proposed definitions. These results suggest that generally it appears easier to classify omnivores and univores in terms of their taste spectrum as opposed to classifying them according to the 'eliteness/exclusivity' of the sports consumed.

5.3 Probit results predicting the probability of a univore classification

In order to make inferences about the types of people that were classified as omnivores and univores under the respective definitions, regressions were estimated to determine what characteristics were more likely associated with each classification for the six different definitions. The probit regression results for the overall sample, for both the conventional and the alternative definitions are presented in Tables 5.2 and 5.3 respectively.

Coefficient Marginal Coefficient Marginal eff	ects
Effects	
White 2063 0166 - 1791 - 0067	
(2608) (4396)	
Coloured 1301 0097 - 5909 - 0313	
(3160) (4272)	
(.5107) (.5272)	
(4689) (5941)	
Male 9562*** 1128 1033 0040	
(2372) (2994)	
Ασε	
21-30 0725 0056 - 6617 - 0312	
(3459) (4525)	
$(1.515)^{-1}$ $(1.525)^{-1}$ $(1.5$	
(3918) (4695)	
41-50 - 3222 - 0313 - 4823 - 0270	
(3214) (5067)	
51+ - 3056 - 0296 - 3327 - 0164	
(.3361) (.5877)	
Education	
Secondary education -3.9051***9889 -3.7960***9438	
(no matric)	
(.2496) (.5619)	
Matric -2.9673***8193 -3.1523***6432	
(.3169) (.4407)	
Post-secondary -3.0876***6792 -3.4970***6765	
(.3419) (.3916)	
Income	
R20 0001-R40 000 .1006 .0075	
(.3382)	
R40 001+13740122	
(.4317)	
Constant 4.1642*** 6.0867***	
(.4047) (.8063)	
N 396 313	
Wald y^2 816 23*** 346 88***	
Pseudo R^2 0.1378 0.0796	
% Correctly predicted 93.18 97.12	

Table 5.2: Probit results predicting the probability of a univore classification under the conventional definitions

Note: Values in brackets denote the robust standard errors. The omitted group for race is "Black", for gender is "female", for age is "11-20", for education is "primary education" and for income is "R0-R20 000"

P=0.01***, p=0.05**, p=0.10*. Worth noting is that 'Income' could not be included in the regression for the Conventional definition-TV. This is because only 12 individuals were classified as univores, and after a cross-tabulation, it became evident that of these 12 individuals, none fell into the top two income categories; hence, no comparison groups were available for the omnivore status.

5.3.1 Probit results for the conventional definitions

The Pseudo R² ranged between roughly 7.9% and 13.8% (Table 5.2). The Wald χ^2 statistics show that for both conventional definitions, the explanatory variables jointly explain the variation in the probability of being classified as a univore (with both P<0.01).

a) Conventional definition - live

Looking at the full model (Table 5.2) and post-estimation F-tests (Appendix C), statistically significant differences in univore status were found between males and females, and univore status differed significantly across different levels of education. No significant differences
were found in univore status across racial, age and income groups. The gender bias suggested that males were 11.28% more likely to be univores when compared to females. Stated in terms of their consumption behaviour, males are more likely to display exclusivity in their choice of live sports consumption. Although this finding is in contrast to existing research in the sport context (for example, James and Ridinger, 2002; Warde, 2006; Lefevre and Ohl, 2011), it is in line with previous empirical evidence in the South African context (for example, Snowball *et al.*, 2009; Antrobus and Snowball, 2010). The gender bias similarities between this study and the existing South African research on consumption behaviour is not surprising, given Bourdieu's (1984) argument that tastes of consumers in societies of similar nature were likely to display great similarities.

In terms of education, a strong significant relationship with univore/omnivore status was evident. Individuals in all the education categories were significantly less likely to be classified as univores when compared to individuals with primary education, although the results indicate that the probability of being classified as a univore decreases as education increases. Univores (or spectators who displayed exclusivity in terms of the types of sporting codes they consumed) therefore seem to be more likely to come from the lowest education level (in this case primary education) as opposed to any other education group. This result opposes what Bourdieu (1984) originally proposed for cultural consumption (in that 'high' cultural forms are more likely to be consumed by individuals from high educational backgrounds). The decreasing probability of being a univore, however, does suggest support for Bourdieu's (1984) proposition. It is similarly in line with other existing empirical studies (which include amongst others, Sullivan and Katz-Gerro, 2007; van Eijck and Leivens, 2008; Antrobus and Snowball, 2010) that argue that education promotes or at least to some extent facilitates the movement away from univorous consumption behaviour towards omnivority.

The results pertaining to education for this definition seem to present an anomaly. That is, it appears that the extremes on the spectrum of education level (in this case those with primary education and those with post-secondary education respectively) are both more likely than the middle education groups to be classified as a univore. A possible explanation for this could lie within the way in which the sports have been classified into elitist sports and sports associated with the masses. Furthermore, it may be attributed to the fact that the individuals from the two extremes engage in different types of univorous consumption (for example, one group may consume only elite sports and the other only sports that are associated with the masses – yet

both are classified as univores owing to the exclusivity in their taste spectrum). Further research could empirically explain this anomaly.

b) Conventional definition - TV

The results for the Conventional definition–TV are presented in Table 5.2. Similar to the Conventional definition-live case, education was found to be a statistically significant variable in explaining univore status. Across all the education categories, a significant negative relationship was evident which means that, individuals in all the education categories were found to be significantly less likely to be classified as univores than individuals with a primary education. Interestingly, unlike live sports consumption, the television consumption behaviour results for this definition showed no gender bias. This could potentially be explained in that men may be relatively more likely to watch sport live when compared to women, whereas no such gender differences in the viewing of sport on television are evident. This explanation is supported by previous empirical studies (for example, Mallon, 2012; James and Ridinger, 2002) where it was noted that relatively more men watch live sporting events when compared to women yet the difference is not as distinct in terms of watching sport on television (Muller, 2014; McGinnis *et al.*, 2003).

For both the conventional definitions, the results for age were not found to be statistically significant. This finding is somewhat surprising, given the existing empirical stance that the most distinct differences in terms of sports consumption appear to be in terms age and gender (Lefvre and Ohl, 2011; Warde, 2006). Similarly, for both conventional definitions, income was not found to be statistically significant in explaining omnivore/univore status. With previous empirical research (for example, Lefvre and Ohl, 2011; Peterson, 1992) generally referring to education and income combined as a proxy for social class, this finding does not support any of the empirical research findings in their entirety.

As a general overview, from the conventional definitions it appears that the results do not conclusively support a single stance taken in previous empirical studies with regards to sociodemographic variables. This lack of is support is warranted by Warde (2006) where he noted that, while some sports carry connotations of social position, this is limited in relation to sports spectating. It could possibly further be explained by the preposition by Lefvre and Ohl (2011:48) who noted that "homology is not the dominant characteristic of the relation between the sporting field and social positions". This general finding again emphasizes the importance of taking into account the results from alternative classifications of univores and omnivores to be able to gauge whether sports consumption is to any extent held up in social ties. As such, the next subsection considers the results from the four alternative definitions of omnivores and univores.

Variable	Definition 1 ¹⁶		Definition 2 ¹⁷			Definition 3 ¹⁸	Definition 4 ¹⁹		
	Coefficient	Marginal Effects	Coefficient	Marginal effects	Coefficient	Marginal Effects	Coefficient	Marginal Effects	
Race		e		e		6		e	
White	.7118***	.2053	5731**	1474	0876	0113	-1.1523***	1128	
	(.2125)		(.1881)		(.2256)		(.2494)		
Coloured	.4629**	.1488	1951	0470	.0056	.0007	4972*	0333	
	(.2579)		(.2270)		(.2844)		(.2636)		
Indian	2763	0733	1566	0371	.4456	.0763	.0137	.0012	
	(.4065)		(.3234)		(.4250)		(.3660)		
Male	3357	1040	5539***	1555	.0461	.0058	6298**	0681	
	(.1639)		(.1688)		(.2100)		(.2139)		
Age									
21-30	0446	0130	0173	0044	.4018	.0576	.0564	.0048	
	(.2289)		(.2747)		(.3632)		(.2873)		
31-40	0615	0178	1535	0370	.2791	.0416	1725	0130	
	(.2786)		(.3245)		(.4306)		(.3686)		
41-50	3318	0888	.0370	.0095	.3523	.0542	3911	0262	
	(.2809)		(.2858)		(.4207)		(.3360)		
51+	3519	0938	1817	0435	.5973	.1030	6875*	0396	
	(.2646)		(.3038)		(.3762)		(.3831)		
Education									
Secondary education (no matric)	.4226	.1374	.2347	.0641	1013	0124	4.5012***	.9942	
	(.5820)		(.4635)		(.5563)		(.3169)		
Matric	.5260	.1636	3834	0920	1230	-0.154	4.0846***	.9467	
	(.5717)		(.4815)		(.5830)		(.3316)		
Post-secondary	.6920	.2055	1998	0505	.0150	.0019	3.5956***	.7633	
	(.5780)		(.4971)		(.5789)		(.3701)		
Income									
R20 001-R40 000	.3363	.1076	0874	0216	9700*	0770	.3068	.0312	
	(.2160)		(.2465)		(.4037)		(.2982)		
R40 001+	.0650	.0196	.0368	0.009	5105	0466	.7800	.1161	
	(.3163)		(.3317)		(.4920)		(.4851)		
Constant	-1.4975**		.0471		-1.612**		-5.377***		
	(.5923)		(.4871)		(.5652)		(.3010)		
Ν	396	396			396		396		
Wald χ^2	25.73**	30.35	**		13.41**		386.66***		
Pseudo R ²	0.0633	0.075	3		0.0541		0.2156		
% Correctly predicted	76.52	81.82			92.17		91.16		

1 0 0 0 0 0 0 0 0 0 0	Table 5.3: Probit res	ults predicting	g the probability	v of a univore	classification ur	nder the alternate	e definitions
---	-----------------------	-----------------	-------------------	----------------	-------------------	--------------------	---------------

Note: Values in brackets denote the robust standard errors. The omitted group for race is "Black", for gender is "female", for age is "11-20", for education is "primary education" and for income is "R0-R20 000". P<0.01***, p<0.05**, p<0.10*.

¹⁹ Participants were classified as univores if they considered themselves to be only cricket fans, or omnivores if they considered themselves general sports fans.

 ¹⁶ Participants were classified as univores if they watched only professional cricket, or omnivores if they watched professional and amateur cricket.
 ¹⁷ Participants were classified as univores if they watched only cricket live, or omnivores if they watched cricket and other sport live.
 ¹⁸ Participants were classified as univores if they watched only cricket, or only other sports (excluding cricket) on television, or omnivores if they watched cricket and other sport on television.

5.3.2 Probit results for the alternative definitions 1 to 4

The Pseudo R² ranged between roughly 5.3% and 25.4% (Table 5.3). The Wald χ^2 statistics show that for all four alternative definitions, the explanatory variables jointly explain the variation in the probability of being classified as a univore (with all four statistics being significant at the 5% level).

a) Definition 1

Definition 1 pertained to the consumption of varying levels of cricket played. For this definition, race was found to be statistically significant predictor. Apart from race, none of the other explanatory variables significantly explain univore status (although education appears to be a relatively important explanatory variable, it was not statistically significant). The results show that White participants are on average 20.53% more likely to be classified as univores when compared to Black participants. Coloured participants were on average also more likely to be classified as univores than were Black participants. This suggested that Black participants were relatively more inclined to watch professional and amateur cricket fixtures (thus displaying omnivorous consumption behaviour), while White and Coloured participants were relatively more likely to watch only professional cricket (which is associated with univorous consumption behaviour).

While previous empirical studies have made no comment on the likelihood of a univore/omnivore classification with regards to race specifically, given South Africa's history of racial divisions and the social tolerance implications that are associated with univore status (Bryson, 1996), it is valuable to consider the possible reasons underpinning differences in consumption behaviour with regards to race. The origin of the omnivore/univore theory in conjunction with considering South Africa's past of racial segregation could explain these findings. In Bourdieu's (1984) study of cultural consumption, he noted that consumption behaviour essentially stems from, *inter alia*, one's upbringing, education and socio-economic background. More specifically, he noted that univores (or consumers with high cultural capital) were likely to be from higher income and higher education *backgrounds*, while omnivores (consumers with low cultural capital) were more likely to be from higher income and higher education *backgrounds*. Given the nature of the Apartheid regime, it could be argued that the large majority of the Black South African population may be from lower

income and lower educational backgrounds (even if on average their income and education status has since progressed to higher levels respectively), which, if considered in conjunction with Bourdieu's (1984) preposition, could explain why Black participants were more likely to display omnivorous sports consumption relative to White participants. Another possible reason for this finding may similarly be attributed to South Africa's history. As noted by Nixon (1992:70), during the Apartheid years the "South African 'national' (sports) teams were racially exclusive" and non-White sportsmen were banned from professional sport participation. To redress the inequalities of the past, "sporting authorities have reintroduced racial quotas for SA's teams in a bid to encourage more Black sportspeople to emerge through the ranks" (Njanji, 2013:1). If one takes into account the empirical findings of Miles and Sullivan $(2010)^{20}$, it could be suggested that Black parents (who were previously excluded from sports participation in the Apartheid era) may now be encouraging and supporting/spectating their children (who are not sanctioned by the exclusions of Apartheid) to get involved in a number of sports. This could explain why Black participants were on average less likely to be classified as univores. Furthermore, Brand South Africa (2012) argued that, "it is football - or soccer, as it is universally called here - that has won the hearts of South Africa's black majority". With South Africa's Black majority consuming more soccer (irrespective of the level at which it is played) it could explain why Black participants were found to be relatively more omnivorously inclined.

Bourdieu (1984) and Jensen (2009) underpin a further argument that could support the racial differences found in consumption behaviour. Bourdieu (1984) noted that cultural appreciation provided a means of judgment of social superiority – suggesting that univores were likely to have relatively higher grounds of social superiority when compared to omnivores. While this would have been the case of White consumers under the Apartheid regime, it is somewhat concerning that such consumption behaviour divisions still prevail a decade after the end of Apartheid. Jensen (2009:1), however, argues that "the brutality of apartheid ended in 1994 with free elections, but the white-supremacist ideas that had animated apartheid and the racialized distribution of wealth it was designed to justify didn't magically evaporate". This could further imply that, although South Africa is in many ways trying to redress the effects of apartheid, there are still characteristics of apartheid that prevail. Such prevailing characteristics could therefore suggest why consumption behaviour of sport is still to some extent reflective of racial and supremacy divisions. This is concerning if Bryson's (1996) and Tampubolon's (2008) findings are taken into account. According to Bryson (1996),

²⁰ A positive relationship between consumption behaviour and adult encouragement was found

omnivorousness was found to be positively related to social tolerance. Similarly, Tampubolon (2008) noted that univores had tendencies to exclude other social groups. With South Africa being in a phase of rectifying and reconciling racial divisions of the Apartheid era, it is concerning that consumption behaviour was found to differ significantly along racial lines under this definition. In light of this finding, it would therefore be suggested that it is necessary to encourage the consumption of amateur sport, particularly between White and Coloured South Africans. An increase in the consumption of amateur sport could result in a movement away from univorousness and towards omnivorousness, which could in turn suggest a movement towards higher degrees of social tolerance (Bryson, 1996).

b) Definition 2

Definition 2 pertained to whether participants chose to watch only cricket live, or chose to watch other sport live too. For this definition, the emphasis of the consumption classification was again on the breadth of the participants' consumption. For this definition, results for race and gender were found to be statistically significant. White participants were on average 14.74% less likely than Black participants were to be classified as univores under this definition. This means that Black participants were on average more likely to watch only cricket live, whereas White participants were on average more likely to attend a range of different sports live. Stated in terms of their univore status, it was found that Black participants were on average relatively more likely to be classified as a univore than any other race group. A possible explanation for this finding could relate to the racial notion of the 'Black Diamond'. According to Chevalier (2014:4), the racial term 'Black Diamonds' refers to "members of the new Black middle class - well-educated, professional and affluent". Such individuals' consumption behaviour has been described as "conspicuous consumption" (Sentleste, 2010:1) and has been argued to be a response to "historical deprivation by groups who wish to affirm their new social standing" (Chevalier, 2014:6). With this in mind, it could explain why Black South Africans are likely to be more univorous and in a sense display 'exclusive tastes' by choosing to consume only cricket live as a means of affirming their postapartheid social standing. The post-estimation F-test results (Appendix C) also revealed a similar finding between White and Coloured participants. A statistically significant difference was found, which suggested that Coloured participants were more likely to be classified as univores when compared to White participants. A plausible explanation for this may similarly be that (although to a lesser extent than were Black participants), Coloured participants display exclusive tastes as a means of affirming their post-apartheid social standing. In light of these finding and, given the social tolerance implications that are attributed to univore status, this finding is somewhat concerning given that South Africa is aiming to redress the inequalities and racial divisions of the Apartheid era and encourage social tolerance (National Development Plan, 2014). Therefore, it is necessary to promote the live consumption of a wide range of sports, particularly amongst Black and Coloured participants. A possible way in which this can be done is through issuing complimentary tickets for a variety of sporting codes for instance (perhaps specifically targeting previously disadvantaged individuals). This could promote a wide range of sports consumption and in turn encourage omnivority. This may also allow individuals who were previously subject to narrow consumption options (because of the Apartheid regime constraints) the option to consume a wider variety of sports codes. Encouraging a breadth of sports consumption could in turn result in an increase in social tolerance in South Africa.

Gender differences were found in the classifications of univores under this definition. Men were on average 15.55% less likely to be classified as univores under Definition 2 when compared to women. In terms of exclusivity of tastes, this finding is consistent with previous empirical research in the sporting context (Peterson, 1992; Lefevre and Ohl, 2011; Warde, 2006; James and Ridinger, 2002); however, inconsistent with what was shown for the Conventional definition-live. The gender difference that was found is, however, surprisingly, inconsistent with previous empirical research conducted in South Africa (Snowball et al., 2009; Antrobus and Snowball, 2010). The inconsistency of empirical results of the South African studies (all of which were conducted within the same geographical context, i.e. Eastern Cape, South Africa) is in contrast to previous empirical prepositions. Bourdieu (1984), for instance, noted that tastes of consumers in societies of similar nature were likely to display great similarities. The results, however, suggest that sport and cultural tastes of the same geographical context did not display such similarities. A likely explanation for why sports consumption taste-patterns do not closely resemble those of the cultural context within the same society could be attributed to a change in tastes over time. As noted by Simmons (2006), shifts in consumption behaviour can be attributed to a change in tastes towards sport, which could possibly explain the differences in tastes between cultural and sports consumers in the Eastern Cape.

c) Definition 3

Definition 3 considered the breadth of consumption with regards to watching sport on television. Owing to the similarity of the definitions, and with the only differentiating factor between the definitions being the nature of consumption, the a priori expectation (underpinned by the argument of Karg and McDonald, $(2011)^{21}$) was that the findings for Definition 3 would be largely in line with Definition 2. In other words, it was expected that consumption behaviour would be similar regardless of the nature of the consumption (watching in attendance as opposed to on television). Interestingly, the results for the different forms of consumption were found to be vastly different. As shown in Table 5.1, relatively less people were classified as univores under Definition 3 when compared to Definition 2. This is surprising given that Miles and Sullivan (2010) noted that television viewing should in fact provide consumers with a means of accessing legitimate culture (which could in turn suggest that television encourages univorous consumption behaviour). Miles and Sullivan's (2010) argument could, however, be used to explain why relatively more participants were omnivores under Definition 3 than Definition 2. In line with their argument, television can allow consumers access to sports consumption that they may not necessarily have had access to in terms of live consumption. Individuals, for instance, may not be able to afford admission tickets, or may be subjected to geographical restrictions that prevent them from having access to watch live sport; but television would allow such restrictions to be overcome. Television can therefore afford consumers the opportunity to consume a wider range of sport than what is accessible to them in terms of watching sport in attendance. The access to a wider array of sporting codes could in turn suggest support for the relatively greater amount of participants classified as omnivores under Definition 3 as opposed to Definition 2.

The probit regression results for Definition 3 show that, unlike what was found for live sports consumption under Definition 2, gender and race were not significant predictors of univore status. The post-estimation F-tests²² indicate that no significant difference was found between any of the other explanatory variables, except between individuals in the R0 – R20 000 and R20 001 – R40000 income categories (P<0.10). This finding suggests that the likelihood of middle income (R20 001 - R40 000) earners being classified as a univore is 7.70% less than it is for lower income (R0 – R20 000) earners. This finding is inconsistent with Bourdieu's

²¹ It was argued that complements/similarities in consumption exist across different forms of sports consumptions

²² The post estimation F-test results that test for significant differences between the explanatory variables (other than the assigned base groups) are found in Table C1 in Appendix C.

(1984) proposition, yet consistent with more recent empirical research (for example, Peterson, 1992; Snowball et al., 2009; Chan and Goldthope, 2005). A plausible explanation for the relatively greater probability of being a univore among lower income earners could be attributed to the availability of leisure time. According to Devine et al. (2006) and VW Staff (2012), low-income earners spend a larger portion of their time working (often having to work more than one job to support their families - particularly prevalent in a third world country like South Africa). With relatively less leisure time at their disposal, low-income earners may not spend as much time watching television²³ in comparison to relatively higher income earners. Given this, it is probable that higher income groups are likely to consume a relatively greater range of sporting codes on television, which could in turn suggest why they are likely to be more omnivorous when compared to lower income earners. To overcome the restriction of limited time at the disposal of many lower-income earners, a possible recommendation would be to make televised sport accessible at areas of employment. Employers, for example, may provide a communal work television that employees may have access to during their mandatory work breaks (such as scheduled lunch and tea breaks). This would afford employees the opportunity to watch televised sport. This opportunity could encourage the consumption of a wider range of televised sport that would have otherwise not been possible.

Reid (2012) highlights another argument that could explain the differences in univore status among different income earners. According to Reid (2012), many South Africans cannot afford television subscriptions to broadcasting networks such as DStv and TopTV. Given that DStv, for instance, prides itself as having "an unbelievable range of [...] sport" (DSTV, 2014:1), individuals who are able to afford such networks are more likely to have access to a wider range of sport, relative to those who cannot afford such networks. This could therefore suggest why lower income earners were on average found more likely to be univores than relatively higher income earners. In the interest of promoting social tolerance, through the promotion of omnivority (Bryson, 1996), there is a need to promote a wide range of sports television consumption particularly amongst lower income earners. A possible recommendation to promote this would be to establish events where a range of sport is broadcast to the public at an open access venue. This would allow consumers who cannot afford to own a television, access to watch a range of sporting activities. As argued by Banks (2013), sport, particularly in conjunction with technology, has been recognized as having

²³ Worth noting is that, although it is an encompassing generalisation, according to Frey and Benesch (2008) television consumption is regarded as a dominant leisure activity in most individuals' lives.

unifying properties amongst citizens, even in trying circumstances. Furthermore, Banks (2013:1) noted that sport, through the development of a universal language, promotes better communication amongst a country's citizens and allows communities to "face up to, and solve, many of the challenges where they (we) live". This supports the need for creating ways of allowing greater public access to televised sport, and in so doing, promoting omnivority through access to a broader range of televised sport.

d) Definition 4

The final definition looked at whether sports consumers considered themselves to be general sports fans or solely cricket fans. The results showed statistically significant differences within race, gender, age, and education in determining the consumption classification. The *a priori* expectation for the findings under this definition was that they should be largely similar to the findings of Definition 2 and 3. This *a priori* expectation is grounded in the reasoning that, if an individual considers himself/herself to be a fan of cricket only, he or she is likely to choose to consume only cricket live; whereas a general sports fan would be more likely to choose to consume a range of sporting codes.

In contrast to Definition 1, White and Coloured participants were on average found less likely to be classified as a univore than were Black participants. This meant that White and Coloured participants were more likely to be general sports fans as opposed to only cricket fans, while the opposite was found to be true for Black participants. A possible explanation for this may again be attributed to the notion of the 'Black Diamond'. It is therefore recommended that there is a need to promote a broader sports consumption taste spectrum, particularly amongst Black South Africans, in order to promote social tolerance. This finding is largely similar to what was found under Definition 2 and therefore confirms (in part) the *a priori* expectations. The finding could suggest that, when compared to Black participants, White and Coloured participants generally engage in similar consumption behaviour trends across different forms of consumption. Furthermore, the similarities that were found between Definition 2 and Definition 4 could suggest that trends of live sports consumption could be an indication of the likelihood of a participant classifying themselves as a fan of cricket only or a general sports fan²⁴. In contrast to the *a priori* expectation, the results for Definition 4 appeared vastly different to Definition 3. This could suggest that there is a stronger

²⁴ This differentiation is hereafter referred to as 'fan classification'.

relationship between live consumption of sport and fan classification than there is between television consumption of sport and the same classification.

Similar to Definition 2, the results showed that there was a slight gender bias for Definition 4. Men were on average found to be 6.81% less likely to be classified as univores than women. As was the case for Definition 2, this finding was consistent with previous empirical research in the sports context, but inconsistent with consumption behaviour studies that were conducted within the same society. In terms of age, only very marginal differences were found, with the only statistically significant difference being found between the youngest and oldest age categories. Individuals from the oldest age bracket (51+) were 3.96% less likely to be classified as a univore when compared to individuals in the 11-20 years age bracket. Although not statistically significant, worth noting is that the size of the coefficients do suggest that older participants were on average less likely to be classified as univores than were relatively younger cohorts. This suggests that on average, younger cohorts are more likely to be classified as cricket fans only, as opposed to general sports fans. The finding that younger cohorts are on average relatively more likely to display univorous characteristics than older cohorts is consistent with existing research (Collins, 1992; Antrobus and Snowball, 2010). The finding is, however, inconsistent with other empirical studies (Peterson and Kern, 1996; Rossman and Peterson, 2005; Warde, 2006; Tampubolon, 2008; Snowball et al., 2009; Widdop and Cutts, 2013). A plausible explanation for the exclusivity of younger cohorts' consumption may be underpinned by a characteristic of their life stage. According to Roberts (2012), it is during the younger/childhood ages when people gain their basic leisure tastes. This suggests that leisure tastes start off as basic and as individuals get older, their leisure tastes may develop. This could in turn possibly explain why younger cohorts were found more likely to be only cricket fans (as this was in their early stages of leisure taste attrition), while older cohorts were more likely to be general sports fans (given that their leisure tastes have developed with age). Another explanation for the difference found between the oldest and youngest age groups could be attributed to the time that is at their disposal to allocate to leisure consumption. As highlighted in The World Youth Report (2003:213), "the quality and quantity of young people's discretionary hours are often diminished by strict curfews. When test scores drop or family incomes dip, opportunities to participate in voluntary activities are often restricted, as the hours required for work or study are increased". Thus, with less time at their discretion, this could explain why younger cohorts were relatively more likely to be associated with univorous consumption behaviour. Furthermore, it is plausible that individuals who fell under the oldest age category may be individuals that are no longer part of the workforce and therefore could have more time at their disposal to allocate to the consumption of leisure activities. This could therefore explain why individuals in the oldest age category were found to be significantly more likely than the youngest (P<0.10) and second youngest (P<0.05 as evident in the post-estimation F-test) age categories to be general sports fans.

Given the social tolerance implications that are attributed to univorous consumption behaviour, it would therefore be recommended that the Department of Sport and Recreation in South Africa consider trying to encourage young South Africans to become fans of a wider array of sporting codes. In line with the arguments of White and Wilson (1999), Wilson (2002), Thrane (2001) and Mehus (2005)²⁵, this could be done by encouraging young people to participate in sport. The Department, for example, could invest in development programs of a range of sports (other than mainly cricket, rugby and soccer) for young people to partake in. Based on the argument that sports participation is positively related to other forms of sports consumption, this may encourage young people to become general sports fans and therefore a movement away from univority. Another manner in which omnivourous consumption behaviour can be encouraged amongst young and older (in this case 51+) individuals, is for sports franchises/stadia to provide some sort of incentive for attendance for these age groups. For example, all sports events could give scholars and pensioners discounted rates of admission, which could in turn encourage consumption.

Unlike the findings of both conventional definitions, under Definition 4, participants with either secondary education (with no matric), matric, or post-secondary levels of education were all on average found a lot *more* likely to be classified as univores than participants with primary education. This could otherwise be interpreted as compared to individuals with primary education, individuals from relatively higher levels of education were on average more likely to be general sports fans as opposed to only cricket fans. Furthermore, the probability of being a univore under this definition decreases as the level of education increase. Thus, this finding presents a similar anomaly to that found for education under the Conventional definition-live. That is, that compared to all the other education groups, individuals with primary education are least likely to be univores. Underpinned by Bourdieu's (1984) line of argument, a possible explanation for this finding could be that individuals from the lowest level of education are associated with the least amounts of social capital (in this case specifically sporting knowledge) which is needed to appreciate the technical aspects of a

²⁵ These researchers noted that sports participation is positively related to other forms of sports consumption.

sport. Thus, individuals with primary education generally do not possess high degrees of social capital and are thus not likely to get more involved (i.e. a deeper approach/appreciation) in one sport but rather be involved with a relatively more surface approach in a range of sports. Although this may explain part of the finding, the same line of argument does not hold for higher levels of education. The results indicate that the probability of being a univore decreases as education increases, which essentially supports Peterson's (1992) hypothesis. Peterson (1992) argued that individuals from higher education groups are likely to prefer a wider range of consumption. This could be used to explain why the probability of being classified as an omnivore (or being classified as a general sports fan and therefore having a wide sports consumption) increased as education levels increased. The anomaly in this instance exists in that this argument would not hold true for the lowest level of education. Why the anomaly exists is not clear. In this instance the prevalence of the anomaly cannot be attributed to the way in which the sports were classified (as was proposed for the Conventional definition-live) as this definition did not account for the nature/eliteness of sporting codes. Thus, as was suggested for the Conventional definition-live, the explanation for the existence of the anomaly calls for further research.

5.3.3 A comparative analysis of socio-demographic findings across definitions

This section presents a general overview of the demographic variables that were found to be significant determinants of univore status under each definition. Possible explanations for the differences in determinants across definitions are suggested and a conclusion on the extent to which sports consumption is bound up in social ties is drawn.

In comparing the demographic determinants of univore status in the conventional definitions, it is evident that education was a strong determinant of univore status. Levels of education were therefore suggested to be significant in influencing types of sport consumed. For both conventional definitions, income was not statistically significant. With income and education generally considered proxies for social status, it is therefore suggested that this finding does not warrant a single stance in empirical research to date. In line with the argument by Warde (2006), it appears that when assigning univore status on the basic of the nature/eliteness of sports consumed, consumption behaviour may carry connotations of social position, but to a limited extent in relation to sports spectating.

An *a priori* expectation of the research was that there would be consistency in the determinants of the different definitions that pertain to the same form of consumption behaviour. That is, it was expected that similarities in the Conventional definition-live and Definition 2 (which pertained to live sports consumption) would be evident, and, similarly, similarities between determinants under the Conventional definition-TV and Definition 3 would emerge. In terms of live consumption, the only similarity that emerged was that, in both instances, univore status was found to differ significantly between males and females. Although gender was a common determinant in explaining univore status under both definitions, the nature of the gender bias was different. When compared to females, males were more likely to be classified as univores under the Conventional definition-live and less likely to be classified as univores under Definition 2. This difference could stem from the difference in the nature of the classification. For example, the Conventional definition-live takes into account the nature/eliteness of the sporting codes whereas Definition 2 considers only the breadth of individuals' live sports consumption spectrum. This highlights the value of including more than just the conventional understanding of omnivores/univores in order to provide a multidimensional understanding of the extent to which consumption behaviour of sport is bound up in social ties. This value is reiterated in the fact that there were no similarities found in the determinants of consumption of sport on television.

Across the alternative definitions, education was only found to be statistically significant under Definition 4. In line with the conventional definitions, this finding could be indicative of connotations of social positioning within the breadth of consumption. Owing to the differences in the nature of the relationship between education and univore status across the definitions, it is not possible to attribute the similarities found to the amount of social capital needed to appreciate an inclusive taste spectrum. The similarities between Definition 4 and the conventional definitions could, however, be explained in terms of similarities in the breadth of consumption behaviour. That is, it could be that individuals who displayed exclusivity in their consumption behaviour under the conventional definitions, may similarly display exclusivity in that they consider themselves only cricket fans as opposed to general sports fans. Another plausible explanation could be underpinned by the fact that each of the definitions is tested exclusively. Therefore, it may be that the same individuals were assigned as univores under more than one definition, and therefore, the similarities in the findings of the determinants should be attributed to the individuals rather than the variable itself (i.e. it may be that the univores under Definition 4 and the conventional definitions were largely the same group of individuals). Future research would be needed to decipher whether the group of individuals that were classified as univores under one definition were the same group of individuals that were assigned as univores under a different definition.

Statistical significant differences between race groups were evident in explaining univore status under Definition 1, 2 and 4. Given that Definition 3 pertains to television consumption, this could imply that consumption behaviour between races is different for different forms of consumption; and that watching sport on television narrows the existence of racial differences in sports consumption behaviour. It can thus be suggested that, given the social tolerance ramifications that are attributed to univore status, and given the history of South African racial divides, South Africa should value sports broadcasting and perhaps undertake initiatives to enhance greater accessibility to televised sport (for example, giving broadcast rights to the non-pay television channels).

Univore status was only found to differ significantly based on age in one instance. As was explained in Section 5.3.2, this could be attributed to the time at the disposal of the individuals in the oldest age category. The lack of significance of the age variables under all the other definitions suggests that, unlike what was found in previous empirical studies (for example, Snowball *et al.*, 2009; Tampubolon, 2008; Peterson and Kern, 1996; Rossman and Peterson, 2005), consumption behaviour patterns (in this case of sport) do not significantly differ with age. One possible reason for this could be that the nature of watching sport in South Africa is family-orientated and as a result, members of a family across different age groups, for example, would watch sport together. As highlighted in the Green Paper for Families (2012:40), the South African Department of Sport and Recreation prides itself on "making sport and recreation accessible to South African families". Therefore, it could be that in South Africa, an individual's sports consumption is underpinned by the sports consumption of other members in their family. This reiterates the need to consider the motives for attendance to obtain a holistic understanding of sports consumption behaviour patterns.

From an overall comparative analysis of the socio-demographic findings across all definitions, it is concluded that consumption behaviour of sport displays elements of social connotations. It appears that to a large extent sports consumption behaviour is tied to educational hierarchies, yet the same cannot be said for income groups. Given that previous empirical studies use both education and income as a proxy for social position, this finding therefore does not support any previous empirical studies in its entirety. It is suggested from this, that consumption behaviour of sport for this context is unique, in that not only do the

81

findings differ from sports consumption in other countries (for example, Widdop and Cutts, 2013; Lefevre and Ohl, 2011; Warde, 2006) but they also differ from the findings of other forms of luxury activity consumption behaviour within South Africa (for example, Snowball et al., 2009; Antrobus and Snowball, 2010). Finally, and particularly pertinent to South Africa, was the role that race played in underpinning consumption behaviour. From this is can be suggested that there are elements of racial differences in sports consumption behaviour in South Africa (in this case, particularly amongst Eastern Cape cricket spectators). As discussed in Section 5.3.2, given South Africa's history of racial divides and the social tolerance implications that could stem from exclusive consumption behaviour, it is strongly suggested that the Department of Sport and Recreation and sports marketers consider strategies to eradicate, or at least minimise, these racial differences in consumption behaviour of sport. Possible ways in which this could be done might be through targeted marketing strategies with the aim of 'speaking to' all racial groups (for example, using professional players of different races in marketing media or advertising in different South African languages). Based on the arguments of Bryson (1996) and Van Eijck and Leivens (2008), this could encourage relatively more inclusive taste spectrums and in turn lead to higher levels of social tolerance levels in South Africa, particularly amongst race groups.

5.4 Motivation for attendance: A comparison between classifications

As reviewed in Section 2.4.1, existing literature (both theoretical and empirical) has proposed numerous factors/motives that could explain consumption behaviour of sport. According to Crompton and McKay (1997), understanding motives for attendance of an event is vital for the provision of better service, in that motives are fundamental in the consumption decision-making process, and subsequently, understanding motives could lead to better attendance. Snowball *et al.* (2009) argued that motives for attendance might be productively linked to omnivore/univore status and therefore an analysis of how motivation differs across univore status is valuable in understanding sports consumption behaviour.

Table 5.4 presents the findings generated from the bivariate cross-tabulations, showing differences in motivation for attendance between omnivores and univores across the conventional and alternative definitions.

Question	uestion Definition 1		Definition 2		Definition 3		Definition 4		Conventional		Conventional	
									definition-live		definition-TV	
	Univore	Omnivore	Univore	Omnivore	Univore	Omnivore	Univore	Omnivore	Univore	Omnivore	Univore	Omnivore
1^{26}	20.67	79.33	19.95	80.05***	8.41	91.59	8.17	91.83*	93.27	6.73*	97.36	2.64
3^{27}	19.06	80.94	18.71	81.29**	8.99	91.01	7.55	92.45**	94.96	5.04	97.48	2.52
4^{28}	22.22	77.78	18.60	81.40*	8.27	91.73	7.75	92.25	94.32	5.68	97.16	2.84*
5 ²⁹	22.92	77.08	18.23	81.77	11.46	88.54	9.38	90.63	91.67	8.33	97.40	2.60
8^{30}	21.80	78.20**	19.19	80.81**	8.72	91.28	9.59	90.41	94.48	5.52	97.09	2.91
10^{31}	18.35	81.65	19.27	80.73	10.09	89.91	9.17	90.83	94.50	5.50	99.08	0.92
11^{32}	20.35	79.65	16.88	83.12**	10.39	89.61	8.23	91.77	96.97	3.03	97.84	2.16
13^{33}	23.32	76.68	19.17	80.83**	8.29	91.71	8.81	91.19*	93.01	6.99	97.15	2.85
14^{34}	15.66	84.34**	20.71	79.29**	11.11	88.89	9.60	90.40	91.92	8.08	97.47	2.53
16^{35}	13.31	86.69	27.42	72.58	12.10	87.90	11.29	88.71	89.92	10.08	96.77	3.23
18^{36}	18.13	81.87*	19.55	80.45	8.50	91.50	8.78	91.22	92.92	7.08	97.73	2.27
19^{37}	20.05	79.95	20.30	79.70***	8.27	91.73	9.02	90.98	92.98	7.02	97.74	2.26
21^{38}	20.45	79.55*	20.20	79.80**	8.33	91.67**	7.32	92.68**	93.43	6.57	97.47	2.53
22^{39}	25.45	74.55	20.00	80.00	10.91	89.09	9.09	90.91	94.55	5.45	96.36	3.64
23^{40}	15.56	84.44	18.29	81.71**	8.95	91.05	8.17	91.83	93.77	6.23	98.05	1.95

Table 5.4: Motivation for attendance results: A comparison between classifications

Note: Figure shows the proportion of the spectators who indicated either 'strongly agree' or 'agree' for the respective questions. $P<0.01^{***}$, $p<0.05^{**}$, $p<0.10^{*}$ (indicating significant differences in motivation to attend between omnivores and univores).

²⁶ Question 1: "I consider myself a fan of South African cricket"
²⁷ Question 3: "I play/have played cricket competitively before (At school/provincially/internationally)"
²⁸ Question 4: "I consider myself a fan of cricket as a whole and not just of a particular team"

²⁹ Question 5: "I attend cricket matches mainly because my friends or family are attending"

³⁰ Question 8: "The social experience (e.g. socialising at the game, dancing, beer tent, competitions, etc.) is just as important as the cricket match"

³¹ Question 10: "I attend cricket matches because I am specifically a fan of one/some of the players"

³² Question 11: "I enjoy attending the before and after game entertainment"

³³ Question 13: "I prefer close competition games where the outcome is uncertain"

³⁴ Question 14: "I attend the cricket matches a way of meeting new people"

³⁵ Question 16: "Cricket is my favourite sport to watch live"

³⁶ Question 18: "Watching cricket allows me to feel part of the community"

³⁷ Question 19: "Watching South African cricket makes me feel proudly South African"

³⁸ Question 21: "I believe playing sport at school developed my love for sport"

³⁹ Question 22: "If a certain player/some players was/were not playing I would not attend"

⁴⁰ Question 23: "Some of my family members play/have played cricket competitively"

In terms of the conventional definitions, it appears that motivation for attendance was largely the same between omnivores and univores, with the exception of one motive for each conventional definition respectively. Statistically significant differences in the motivation for attendance under the Conventional definition-live pertained to whether individuals considered themselves fans of South African cricket. For this question, 93.27% of the individuals who considered themselves fans of South African cricket were univores, whereas 6.73% were omnivores. This could suggest that considering oneself as a fan of a specific sport may be indicative of an exclusive taste spectrum. As argued by Simmons (2011), if you are a fan of a specific sport (in this case South African cricket) it requires one to devote time, energy and financial resources into that particular sport. Given that the resources devoted to such activities may not simultaneously be used to engage in other sports/luxury activities (Goode, 1960), it may mean that individuals are choosing to be devoted fans of a specific sport, to the exclusion of others. To encourage omnivorous behaviour within South African sport, it could therefore be recommended that stakeholders (for example, the Department of Sport and Recreation) refrain from focusing on the promotion/marketing of a single sport and rather encourage consumption of a range of sport. Possible ways in which this could be done could be to distribute complementary tickets or fan apparel to individuals in hope of igniting an interest in consuming a wider range of sport than what they would ordinarily choose to consume. This could encourage individuals to navigate away from univorous fandom to omnivority, which could in turn have positive implications on social tolerance.

Under the Conventional Definition-TV, the only significant difference in motivation between omnivores and univores pertained to the nature of their cricket consumption preferences. About 97.16% of those who considered themselves to be fans of cricket as a whole as opposed to being a fan of a particular team were univores. This could suggest that individuals who display inclusive taste spectrums in terms of television consumption are generally more concerned about the sport that is screened as opposed to the teams that are playing. A possible explanation for this could be attributed to the nature of sports television broadcasting in South Africa. According to Borland (2013:1), cricket broadcast in South Africa receives "limited coverage". Therefore, given the limited coverage of cricket on television, fans may not have the opportunity to watch one specific team throughout a series, and as a result thereof may be more inclined to appreciate the sport as a whole rather than a particular team. Conversely, given that omnivores watch a wide array of sport on television, their consumption behaviour could be as a result of them choosing to only watch cricket mainly when a specific team is playing (otherwise they substitute their television consumption for other televised sport).

As a general observation, when individuals are classified as under the conventional definitions, the biggest proportion of individuals who agreed to the different questions were univores; whereas the opposite was true in the case of Definitions 1 to 4. This finding suggests that a significant difference lies within motivation for attendance and classifications. More specifically this suggests that, when individuals are classified in terms of the exclusivity of their taste spectrum because of the nature/eliteness of their consumption choices, there are significant differences in their motivation to attend as opposed to when the breadth of consumption is used as the basis of the classification. A possible reason for such differences could stem from the way in which sports were classified as elite/non-elite. It is, however, not apparent why motivation would be vastly different under the conventional definitions as opposed to other understandings of consumption behaviour. This would be a fruitful area of future research in order to ascertain why such significant differences exist. Nonetheless, it highlights the importance of considering more than just the conventional definitions in order to gain a holistic understanding of motivation for attendance.

Under Definition 1, statistically significant differences in motivation between omnivores and univores pertained to the social experience of the game. More specifically, under Definition 1, a strong social motive for attendance is evident amongst omnivores. That is, the majority of the individuals who said yes to Question 8, 14 and 18 (all of which are underpinned by a social motive) were omnivores. In order to encourage omnivourous consumption behaviour (which could in turn be interpreted as encouraging social tolerance) it could therefore be recommended that hosts of sports events invest in social elements of the experience (for example, providing complementary entertainment at the grounds throughout the event).

Across all four alternative definitions, significant differences in response to Question 21 were found. The majority of the omnivores under Definition 1 to 4 agreed that playing sport at school developed their love for sport. This could signal to stakeholders that, in order to encourage omnivorous behaviour, it is important to encourage school-goers to participate in sport. Therefore, it may be worthwhile to invest in school sport and developmental sports programs that have the potential to develop a love for sport at a young age. In line with the arguments by White and Wilson (1999), Wilson (2002), Thrane (2001) and Mehus (2005), encouraging participation of sport at school could imply that there be positive developmental implications for other forms of sports consumption too (e.g. in turn encouraging omnivorous

television and live sports consumption). This recommendation could in turn essentially assist South Africa's Department of Sport and Recreation (2014) in achieving its mission⁴¹.

Finally, from the lack of significant differences across univores and omnivores for Questions 10 and 22 across all definitions, it could be suggested that consumption of sport in South Africa is not significantly dependent on the presence of certain players. That is, motives for consumption do not seem to be underpinned by the presence of specific players. This could in turn imply that marketers of cricket events should not focus on using specific players to market the event but structure the marketing focus around the sport as a whole.

5.5 Summary and Conclusion

This chapter presented the main empirical findings of the research. The probit regression results predicting the probability of a univore classification under each definition were presented and interpreted. From these results, it was concluded that sports consumption behaviour in the Eastern Cape displays elements of social connotations, with significant differences in consumption behaviour of sport found on the grounds of race and education. The chapter then presented the bivariate cross-tabulation results that suggested that omnivores and univores differ in terms of their motives for attendance. Throughout the chapter, a discussion of the findings and recommendations stemming therefrom were presented. The following chapter will summarise the research, discuss the limitations of the study and provide suggestions for future research.

⁴¹ As per their mission statement the Department of Sport and Recreation's (2014:1) mission is "to transform the delivery of sport and recreation by ensuring equitable access, development and excellence at all levels of participation and to harness the socio-economic contributions that can create a better life for all South Africans".

CHAPTER 6: SUMMARY AND CONCLUSION

6.1 Introduction

The primary goal of this research was to investigate whether South African cricket spectators are sporting omnivores or univores; with a secondary goal of identifying motives for consumers making specific sport consumption decisions, and ascertaining whether certain characteristics could be attributed to these consumption decisions. Chapters 2, 3, 4 and 5 presented the theoretical background, core empirical literature, data and methods, and empirical findings and recommendations pertaining to this study. An executive summary of each chapter is presented in Section 6.2, followed by limitations of the study and suggestions for future research in Section 6.3.

6.2 Executive summary of the research

Chapter 2 considers the theoretical background of the study. The chapter presents a broad review of some of the theoretical underpinnings of consumption behaviour in general, the consumption behaviour of leisure, and the consumption behaviour of sport as a form of leisure. Since its inception, consumption theory has evolved and become anthropocentric in nature, with people and societal influence at the forefront of the theory, and as an important source of understanding consumption behaviour. From a review of the determining factors of leisure consumption and particularly sport as form of leisure consumption, it is concluded that social comparison plays a significant role as a determining factor of both leisure and sports consumption.

Given the anthropocentric nature of consumption theory and the theoretical proposition that social comparison/influence is an important determinant of consumption behaviour (Veblen, 1899; Wakefield, 1995; Cornea and Jeanne, 1997 and Choi *et al.*, 2009), Chapter 3 reviews empirical literature investigating social comparison and consumption behaviour of leisure activities. The idea of social comparison in leisure consumption was initially investigated for cultural activities (for example, Bourdieu, 1984; Wilensky, 1964; Peterson, 1992; Lamont, 1992) and has since developed to include the consumption of sport (for example, Miles and Sullivan, 2010; Lefvre and Ohl, 2011; Warde, 2006; Widdop and Cutts, 2013). The idea of social stratification of the consumption of culture (as a form of leisure consumption) was

initially coined by Bourdieu (1984), and first contested by Peterson (1992). Since the origin of the omnivore/univore theory, writers have proposed extensions and developments to the theory, which are reviewed. From these extensions and developments, it is evident that there is value in studies that consider definitions other than the original or conventional definitions of omnivore and univore to move away from the binary understanding of consumption behaviour, towards a more comprehensive and multidimensional understanding of the omnivore/univore theory for a given context.

To date the omnivore/univore hypothesis has been tested in a number of different contexts. Studies testing this hypothesis in the cultural context are reviewed first. The general finding for these studies is that an omnivore/univore classification (or an extension thereof) was possible for patrons who consumed cultural products. Generally, the studies support the 'contested hypothesis' by Peterson (1992). However, there are cases (for example, in Suominen, 2007) where support of Bourdieu's (1984) hypothesis is evident. A common finding of the studies is that cultural tastes, in many cases, can be differentiated along socio-demographic lines; where in most cases reviewed, consumption behaviour differed in accordance with age, gender, education and income. Overall, it is concluded that consumption behaviour is largely still bound up by social hierarchies.

In a review of the application of the omnivore/univore hypothesis in the entertainment industry, the results are very much in line with that of the cultural context. Entertainment consumption behaviour was classified in terms of omnivorous or univourous characteristics and consumption patterns were identifiable on the grounds of socio-demographic variables. An overarching conclusion from the review of these studies, as noted by Chan and Goldthorpe (2005), is that consumption behaviour of entertainment forms can be socially stratified.

A review of the omnivore/univore hypothesis in sport (for the variety of different consumption forms) similarly suggests that the general findings of consumption behaviour of cultural products can largely be used to understand consumption behaviour of sport. Sports consumers can be classified as omnivores or univores and consumption behaviour can be differentiated in terms of socio-demographic variables. A striking difference between the studies in the cultural context and those in the sports context pertains to gender. A number of the cultural context studies generally found that females were more likely to be classified as omnivores than univores (Snowball *et al.*, 2009; Antrobus and Snowball, 2010; Suominen,

2007). The opposite is evident in the case of sports consumption (Lefevre and Ohl, 2011; Warde, 2006; James and Ridinger, 2002). As an overarching conclusion from the review of the sports consumption studies it is concluded that "there are certainly some hints that different sports carry connotations of social position, limitedly in relation to spectating, a little more in relation to participation" (Warde, 2006:119).

The final section of Chapter 3 reviews empirical findings on the motivation for attendance of entertainment products and sports products. As noted by Snowball *et al.* (2009), this is a topic closely related to the omnivore/univore hypothesis. The studies identified numerous influences as factors that motivated attendance. In the cultural context, these factors included recreation, social inclusion, escapism, self-esteem enhancement (which was taken as a proxy for status), and aesthetics. Motives for attendance at cultural events were found to differ across socio-demographic variables. Other motives for attendance included escapism, entertainment, socializing, artistic and status motive. Snowball *et al.* (2009), in support of Swanson *et al.* (2008), found motives to be significantly different across socio-demographic variables (particularly age, occupation and education). Overall, it is concluded from this section that "motivations were found to be as important in determining consumption patterns as socio-demographic variables" (Snowball *et al.*, 2009:482).

Similarly, in the case of sport, motivation for attendance differs across socio-demographic variables. Specifically, motivation for attendance is found to be significantly different between males and females. James and Ridinger (2002) concluded that sport provides an important identity for males that is not shared for females. The motivation for attendance is found to produce interesting implications for marketers and producers of such events, an understanding of motives of attendance can be beneficial to producers and marketers and even result in the enhancement of the product that the consumers are offered. James and Ridinger (2002) further provide an interesting suggestion for future research; that future studies on motivation for attendance should be conducted relative to specific teams, in order to gain better insight into the motives that influence individual team support and the consumption behaviour of sports fans. Snowball *et al.* (2009:482) further support this in the cultural context where it was suggested that "(t)his (motivation for attendance) is a potentially fruitful area for further research".

Finally, the consumption model, proposed by Trail and James (2011), shows that the consumption process is a complex process and is influenced by a wide range of internal and

external factors. From this, it was argued that, owing to the subjective nature and individuality of the decisions governing consumption behaviour, it is necessary for each individual context to undertake research on motives for attendance for a specific group of individuals in order to comprehensively understand their consumption decisions as well as pattern of consumption behaviour. Warde (2006) further noted that sport is a domain of extensive ramifications for a country's economy, society and culture and yet it is often treated as inferior to many of the minority forms of art. Owing to these and other ramifications (as were highlighted in Section 1.1), and the fact that research on sport consumption remains largely underdeveloped (Widdop, 2010), a need to further research and develop the application of consumption behaviour theory to the sport context is concluded.

Chapter 4 presents the data and empirical methods of the study. The chapter provides a description of the different ways in which cricket spectators could be classified into omnivores and univores. Based on previous empirical research the study investigates the conventional classification; in which spectators are classified as omnivores or univores according to the exclusivity of the sports consumed. This is largely informed by Bourdieu's (1984) theory. The adaptation of this definition for the sporting context is done for the two main forms of sports consumption; watching in attendance/live and watching the televised version of the sport. Four alternate omnivore/univore sub-classifications are then proposed for this study.

The first alternate definition focuses on the level at which sport is played. For this definition, spectators are classified as omnivores if they chose to consume professional and amateur sport; or univores if they chose to consume only professional sport. The second alternate definition is based on Peterson's (1992) argument on the importance of considering the array of consumption. In this case, individuals are classified as univores if they only chose to watch cricket live (to the exclusion of other sports), while omnivores are those that watched cricket and other sports live. The third alternate definition is largely the same as the second alternative definition but considers tastes in terms of consumption of televised sport as opposed to consumption in the form of attendance. The final alternate definition is similar to the second alternate definition (and therefore similarly in line with Peterson's (1992) recognition of a consumption. For this definition, univores are spectators who only considered themselves cricket fans, while omnivores considered themselves to be general sports fans.

The chapter then provides a description of the questions from the questionnaire that elicited responses that are true to the omnivore/univore classification for each of the definitions. A description of the questions underpinning the various motives for attendance are also presented. To determine the effect of individual characteristics in predicting the likelihood of being a univore, binary probit models were estimated for the conventional definitions as well as for all four alternate omnivore/univore classifications. The degree to which motivation for attendance differed in terms of omnivore/univore status was also examined.

In Chapter 5, the core empirical results of the study are presented. The results indicate that there are differences in the social-demographic determinants of univore status under each definition. These determinants and the implications thereof are discussed for each of the six definitions. Recommendations pertaining to encouraging omnivorous consumption behaviour (with the aim of promoting social tolerance) are proposed based on the findings.

6.3 Conclusions of the research

From an overall analysis of the findings on univore status under all the definitions, it is concluded that consumption behaviour of sport displays elements of social connotations. It is established that sports consumption behaviour is linked to educational hierarchies, but not significantly to levels of income. Given that education and income have generally been used as a proxy for social position, the findings of this study therefore do not support any previous empirical studies in its entirety. From an overall analysis of the findings, it is also showed that there are elements of racial differences in sports consumption behaviour in South Africa, suggesting that race plays a role in underpinning sports consumption behaviour. This finding is particularly pertinent given South Africa's history of racial divides. Considering the social tolerance implications that could stem from univorous consumption behaviour, it is suggested that stakeholders consider strategies to eradicate or minimise racial differences in consumption behaviour of sport. Suggestions of such strategies are, to ensure that marketing is targeted at all racial groups (for example, using professional players of different races in marketing media or advertising of sport events in different South African languages). Based on the arguments of Bryson (1996) and Van Eijck and Leivens (2008), such strategies could encourage relatively more inclusive taste spectrums and in turn lead to higher levels of social tolerance levels in South Africa, particularly amongst racial groups.

It appears that sports consumption behaviour of Eastern Cape cricket spectators is particularly unique when compared to consumption behaviour of cultural consumers in South Africa and consumption behaviour of sport internationally. The findings of the study differ from both the finding on sports consumption behaviour in other countries and from the findings of other forms of luxury activities consumption behaviour within South Africa.

Finally, in terms of motivation for attendance, it is concluded that there are differences in motivation for attendance between omnivores and univores. Considering the differences in motivation between omnivores and univores, proved useful in putting forward practical recommendations to stakeholders in order to promote specific consumption behaviour that could in turn have positive social tolerance ramifications.

6.4 Limitations of study and suggestions for future research

The first limitation of the study pertains to the cross-sectional nature of the research design. Given that consumption behaviour is dynamic as opposed to static in nature, making definite causal inferences from cross-sectional data should only be attempted with caution. It is suggested that future research should consider the use of longitudinal data in order to make accurate causal inferences and to account for the dynamic nature of sports consumption.

The second possible limitation is that even though the results may be representative of cricket spectators in the Eastern Cape, generalization of the findings to consumers of the wider context is limited. The results of the study only pertain to cricket spectators and therefore one cannot infer the findings to all sports spectators. The generalization of the results is further limited in that data collection was conducted at the same ground and same catchment area. Furthermore, and related to this, is the limitation of the relatively small sample size of the study. Thus, it is suggested that, in order to attain results that can be generalized to South African sports consumers, future research should be conducted at a range of sporting events, hosted at a range of stadia throughout the country.

The final shortcoming of the study could be the presence of common-method variance. Given that data collection is performed by means of a questionnaire only, there may be spurious correlation between the variables in the study, which may have in turn caused the relationships between variables to be under- or over-stated. Thus, it is suggested that future research in this area collect data by means of triangulation.

REFERENCES

- ABEL, A.B. (1990). Asset prices under habit formation and catching up with the Joneses. *American Economic Review Papers and Proceedings*. 80 (2): 38-42.
- ANDEASON, A.R. (1965). Attitudes and Consumer Behaviour: A Decision Model in New Research in Marketing. Institute of Business and Economic Research, University of California, Berkeley, 1-16.
- ALDERSON, A., JUNISBAI, A. and HEACOCK, I. (2007). Social status and cultural consumption in the United States. *Poetics*. 35: 191-212.
- ANTROBUS, R. and SNOWBALL, J. (2010). Physical Theatre consumption and the advent of the 'Festivore': a case study of audience attendance at the South African National Arts Festival. *South African Theatre Journal*. 24: 326-343.
- ASHER, R. (2011). *Cricket on the Back Foot*. Sports Illustrated. [Online]. Available: <u>http://showme.co.za/lifestyle/cricket-on-the-back-foot/</u> [Accessed 29 October 2012].
- BAIMBRIDGE, M., CAMERON, S. and DAWSON, P. (1996). Satellite television and the demand for football: A whole new ball game? *Scottish Journal of Political Economy*. 43: 317-333.
- BANKS, K. (2013). Kicking Conflict into Touch: How Sport and Technology Unite Community and Conversation in Kenya. National Geographic News Watch. [Online]. Available: <u>http://newswatch.nationalgeographic.com/2013/08/15/kicking-conflict-into</u> <u>-touch-how-sport-and-technology-unite-community-and-conservation/</u> [Accessed 11 March 2014].
- BASMANN, R., MOLINA, D. and SLOTTJE, D. (1988). A note on measuring Veblen's theory of conspicuous consumption. *Review of Economics and Statistics*. 70 (3): 531-535.
- BATRA, K.S. and KAZMI, S.H.H. (2008). *Consumer Behaviour: Texts and Cases*. 2e. India: Excel Books.
- BATTAGLIA, M.P. (2008). *Purposive Sampling*. [Online]. Available: <u>http://srmo</u> <u>.sagepub.com/view/encyclopedia-of-survey-research-methods/n419.xml</u> [Accessed: 12 October 2013].
- BEARDEN, W.O. and ETZEL, M.J. (1982). Reference Group Influence on Product and Brand Purchase Decisions. *Journal of Consumer Research*, 9 (2): 183-194.
- BECKER, G.S. and MURPHY, K.M. (1988). A Theory of Rational Addiction. *Journal of Political Economy*. 96 (4): 675-700.
- BECKETT, S. (2014). *Opinion: The Commercial Challenges of Pro-Cycling*. [Online]. Available: <u>http://www.cyclingnews.com/features/opinion-the-commercial-challenges-for-pro-cycling</u> [Accessed: 19 August 2014].
- BERNHEIM, B.D. (1994). A theory of conformity. *Journal of Political Economy*. 102 (5): 841-877.

- BERNSTEIN, D.A., PENNER, L.A., CLARKE-STEWARD, A. and ROY, E.J. (2006). *Psychology*. (7th ed). New York: Houghton Mifflin Company.
- BEST, S. (2010). *Leisure Studies: Themes and Perspectives*. Chapter 2. Socio-economic Approaches to Leisure. London: SAGE Publications Ltd.
- BLACKWELL, R.D., MINIARD, P.W. and ENGEL, F.J. (2001). *Consumer Behavior*. (9th ed). Harcourt College Publishers, Fort Worth, TX.
- BLUNDELL, R. (1988). Consumer Behaviour: Theory and Empirical Evidence A survey. *Economic Journal*. 98 (389): 16-65.
- BORLAND, J. (1987). The Demand for Australian Rules Football. *Economic Record*. 63 (182): 220-230.
- BORLAND, K. (2013). *R20M Loss for Cricket SA as SABC Cut Back*. [Online]. Accessed: <u>http://kenborland.com/2013/01/r20m-loss-for-cricket-sa-as-sabc-cut-back/</u> [Accessed: 22 October 2014].
- BORLAND, J. and MACDONALD, R. (2003). Demand for Sport. Oxford Review of *Economic Policy*. 19 (4): 478-502.
- BOURDIEU, P. (1984). *Distinction: A social critique of the judgment of taste*. London, England: Routledge.
- BRAND SOUTH AFRICA. (2012). Sport in South Africa. [Online]. Available: <u>http://www.southafrica.info/about/sport/sportsa.htm#.UvprvFdDHIW</u> [Accessed: 11 February 2014].
- BRAY, J. (2008). Consumer Behaviour Theory: Approaches and Models. [Online]. Available: <u>http://eprints.bournemouth.ac.uk/10107/1/Consumer_Behaviour_Theory-</u><u>Approaches%26_Models.pdf</u> [Accessed 12 March 2013].
- BREMBER, A. (2009). Understanding Consumer Engagement with Test Match Cricket Multi-market study. TNS Sport. [Online]. Available: <u>http://www.lords.org/data</u> /files/mcc-mulitmarket-study-declining-test-match-attendance-2009-1-10342.pdf [Accessed 14 March 2013].
- BRYSON, B. (1996). Anything but heavy metal: Symbolic exclusion and musical dislikes. *American Sociological Review*. 61 (5): 884-899.
- BURIAMO, B. (2006). Chapter 10. *The demand for sports broadcasting*. Handbook on the Economics of Sport. Northampton: Edward Elgar Publishing Inc.
- CAIRNS, J. (1990). The demand for professional sports teams. *British Review of Economic Issues.* 12 (28): 1 20.
- CAMPBELL, J.Y.and COCHRANE, J.H. (1999). By force of habit: A consumption-based explanation of aggregate stock market behavior. *Journal of Political Economy*. 107 (2): 205-249.

- CARMICHAEL, F., MILLINGTON, J. and SIMMONS, R. (1999). Elasticity of demand for rugby attendance and the impact of BskyB. *Applied Economics Letters*. 6: 797-800.
- CARRASCO, R., LABEAGA, J.M. and LµOPEZ-SALIDO, J.D. (2005). Consumption and habits: Evidence from panel data. *The Economic Journal*. 115 (500): 144-165.
- CHAN, T.W. and GOLDTHORPE, J.H. (2005). The Social Stratification of Theatre, Dance and Cinema Attendance. *Cultural Trends*. 14 (3): 193-212.
- CHAN, T.W. and GOLDTHORPE, J.H. (2007). Social Stratification and Cultural Consumption: Music in England. *European Sociological Review*. 23: 1-19.
- CHEVALIER, S. (2014). The Black Diamonds: a South African phantasmagoria. Economic and Social research council. [Online]. Available: <u>https://www.esrc.ac.uk/.../54a52031-5996-4cc1-a341-6f4c682ca265</u> [Accessed: 10 February 2014].
- CHOI, Y.S., MARTIN, J.J., PARK, M. and YOH, T. (2009). Motivational Factors Influencing Sport Spectator Involvement at NCAA Division II Basketball Games. *Journal for the study of Sports and Athletes in Education*. 3 (3): 265-284.
- COLLINS, R. (1992). Women and the production of status cultures. In: Lamont, M., Fournier, M. (Eds.), Cultivating Differences. *University of Chicago Press*, Chicago: 213–231.
- CONSTANTINIDES, G.M. (1990). Habit formation: A resolution of the equity premium puzzle. *Journal of Political Economy*. 98(3): 519-543.
- CORNEO, G.and JEANNE, O. (1997). Snobs, bandwagons, and the origin of social customs in consumer behavior. *Journal of Economic Behavior & Organization*. 32: 333-347.
- COULANGEON, P. (2005). Changing perspectives on the social differentiation of cultural practices in contemporary societies. Available: <u>http://www.unicatt.it/unicattolica/</u>centriricerca/modacult/Download/Coulangeon.pdf [Accessed: 7 November 2013].
- CROMPTON, J. L. and MCKAY, S. L. (1997). Motives of Visitors Attending Festival Events. *Annals of Tourism Research*. 24 (2): 425-439.
- CUNNINGHAM, G.B. and KWON, H. (2003). The Theory of Planned Behaviour and Intentions to Attend a Sport Event. *Sport Management Review*. 6: 127-145.
- DATE, K. (2012). *T20 (and ODI) is a separate sport and should be seen as such. A Cricketing View.* [Online]. Available: <u>http://cricketingview.blogspot.com/2012/09/t20-and-odi-is-separate-sport-and-needs.html</u> [Accessed: 16 September 2014].
- DEATON, A. and MUELLBAUER, J. (1980). *Economics and Consumer Behaviour*. Cambridge: Cambridge University Press.
- DE MOOIJ, M. (2004). Consumer Behavior and Culture: Consequences for Global marketing and advertising. Thousand Oaks, CA: Sage.
- DEPARTMENT OF SPORT AND RECREATION. (2014). *Mission and Values*. [Online]. Available: <u>http://www.srsa.gov.za/pebble.asp?relid=30</u> [Accessed: 26 October 2014].

- DEVINE, C.C., JASTRAN, M., JABS, J., WETHINGTON, E., FARELL, T.J. and BISOGNI, C.A. (2006). "A lot of sacrifices:" Work–family spillover and the food choice coping strategies of low-wage employed parents. *Social Science and Medicine*. 63 (10): 2591-2603.
- DSTV. (2014). *Brands and Products*. [Online]. Available: <u>http://media.multichoice.co.za</u> /<u>media/brands-and-products/packages/</u> [Accessed 7 March 2014].
- ECONOMIC GLOSSARY. (2012). *Consumer behaviour*. [Online]. Available: <u>http://glossary.</u> <u>econguru.com/economic-term/consumer+behavior</u> [Accessed: 29 July 2012].
- ESCHENFELDER, M.J. and Li, M. (2007). *Economics of Sport*. Morgantown, West Virginia: Fitness Information Technology.
- ESPN. (2013). *All cricket records*. [Online]. Available: <u>http://stats.espncricinfo.com/ci/</u> <u>content/records/283309.html</u> [Accessed 14 March 2013].
- ETV. (2014). *Sports*. [Online]. Available: <u>http://www.etv.co.za/genre/sports</u> [Accessed: 19 August 2014].
- FEEHAN, P. (2006). Attendance at sports events. In: Andreff, W. and Szymanski, S. (eds) Handbook on the Economics of Sport. Northampton: Edward Elgar Publishing Inc. pp 90-99.
- FIZEL, J.L. and BENNETT, R.W. (1989). The impact of college football telecasts on college football attendance. *Social Science Quarterly*. 70 (4): 980-988.
- FOREST, D., SIMMONS, R. and FEEHAN, P. (2002). A spatial cross-sectional analysis of the elasticity of demand for soccer. *Scottish Journal of Political Economy*. 49 (3): 336-355.
- FORT, R. (2000). European and Northern American Sport Differences. *Scottish Journal of Political Economy*. 47 (4): 431-455.
- FRANK, R.H. (1999). *Luxury Fever: Money and Happiness in an Era of Excess*. New York: Free Press.
- FREY, B.S. and BENESCH, C. (2008). TV, Time and Happiness. *Homo Oeconomicus*. 25 (4): 413-424.
- FUNK, D. (2008). Consumer Behaviour in Sport and Events. Oxford: Elsevier Ltd.
- GARCIA, J. and RODRIGGUEZ, P. (2002). The Determinants of Football Match Attendance Revisited: Empirical Evidence from the Spanish Football League. *Journal of Sports Economics.* 3: 18-38.
- GERBER, C. and TERBLANCHE, N.S. (2012). Team Performance and Sport Attendance of South African Super Rugby and Currie Cup Rugby Fans. *South African Journal for Research in Sport, Physical Education and Recreation.* 34: 75-91.

- GILMORE, T.L., MORGAN, E.T., and OSBORNE, SB. (2011). *Annual Industry Accounts*. [Online]. Available: <u>http://www.bea.gov/scb/pdf/2011/05%20May/0511 indyaccts.pdf</u> [Accessed 30 July 2012].
- GLENCOE. (2013). Sports and Entertainment Marketing. [Online]. Available: <u>http://staff</u> web.srk12.org/simons m/presentations/Ch_04.pdf [Accessed 12 March 2013].
- GOODE, W.J. (1960). A theory of role strain. American Sociological Review. 24 (4): 483-496.
- GREEN PAPER FOR FAMILIES. (2012). Green Paper for Families: Promoting Family Life and Strengthening Families in South Africa. [Online]. Available: <u>http://www.</u> <u>thepresidency.gov.za/MediaLib/Downloads/Home/Profiles/PresidentJacobZuma/Press</u> <u>Statements/201212/Green%20Paper%20on%20Families.pdf</u> [Accessed 22 October 2014].
- HALL, J., O'MAHONY, B. and VIECELI, J. (2009). A Model of Predictors of Attendance at Major Sporting Events. Australian and New Zealand Marketing Academy Conference. Melbourne: ANZMAC.
- HANSEN, H. and GAUTHIER, R. (1989). Factors Affecting the Attendance at Professional Sporting Events. *Journal of Sports Management*. 3: 15-32.
- HOGG, M.A. and VAUGHAN, G.M. (2002). Social psychology. Harlow: Prentice Hall.
- JAMES, J.D. and RIDINGER, L.L. (2002). Female and Male Sport Fans: A comparison of Sport Consumption Motives. *Journal of Sport Behaviour*. 25 (3): 260-280.
- JAMES, J.D. AND TRAIL, G.T. (2001). The Motivation Scale for Sport Consumption: Assessment of the Scale's Psychometric properties. *Journal of Sports Behaviour*. 24 (1): 108-128.
- JENSEN, R. (2009). In South Africa, Apartheid is Dead, But White Supremacy Lingers On. [Online]. Available: <u>http://www.counterpunch.org/2009/06/09/in-south-africa-apart</u> <u>heid-is-dead-but-white-supremacy-lingers-on/</u> [Accessed 3 March 2014].
- JOLIBERT, A. and BAUMGARTNER, G. (1997). Values, motivations and personal goals: Revisited. *Psychology and Marketing*. 14 (7): 675-688.
- KAEMPFER, W. and PACEY, P. (1986). Televising college football: the complementarity of attendance and viewing. *Social Science Quarterly*. 67: 176-185.
- KARG, A.J. and MCDONALD, H. (2011). Fantasy sport participation as a complement to traditional sports consumption. *Sport Management Review*. 14 (4): 327-346.
- KATONA, G. (1968). Consumer Behaviour: Theory and Findings of Expectations and Aspirations. *The American Economic Review*. 58 (2): 19-30.
- KIM, Y. K. and TRAIL, G. T. (2010). Constraints and motivators: A new model to explain consumer behavior. *Journal of Sport Management*. 24: 190-210.

- KIM, Y. K. and TRAIL, G. T. (2011). A conceptual framework for understanding relationships between sport consumers and sport organizations: A relationship quality approach. *Journal of Sport Management*. 25: 57-69.
- KUMAR, R. (2008). *Consumption Behaviour*. Consumer Behaviour: MKT724 course notes. National Open University of Nigeria: Lagos.
- LAERD DISSERTATION. (2013). *Purposive Sampling*. [Online]. Available: <u>http://dissert</u> ation.laerd.com/purposive-sampling.php#adv-dis [Accessed 12 October 2013].
- LAESSO, J., ALJUNGDAHL, A.K., KASTBERG, P., NOE, E., ALROE, H.F., CHRISTENSEN, T., DUBGAARD, A., OLSEN, S.B. and KAERGARD, N. (2013). *Theories on motivation and their implications for supporting communication, learning and decision-making in relation to organic food systems*. [Online]. Available: http://www.ifsa2012.dk/downloads/WS6_1/Laessoe.pdf [Accessed 12 March 2013].
- LAMONT, M. (1992). Money, Morals and Manners: The Culture of the French and the American Upper- Middle Class. *University of Chicago Press*, Chicago.
- LANCASTER, K.J. (1966). A New Approach to Consumer Theory. Journal of Political Economy. 74 (2): 132-157.
- LEFEVRE, B. and OHL, F. (2011). Consuming sports: distinction, univorism and omnivorism, *Sport in Society: Cultures, Commerce, Media, Politics.* 15: 44-63.
- LOUDEN, D.L. and DELLA BITTA, A.J. (1993). Consumer Behaviour Concepts and Applications. 4th ed. McGraw Hill.
- LOUW, A. (2010). Sports Law in South Africa. Netherlands: Kluwer Law International.
- LUNENBURG, F.C. (2010). The Decision Making Process. *National Forum of Educational Administration and Supervision Journal*. 27 (4): 1-12.
- MALLON, S. (2012). Attendance at live sports events by adults in Northern Ireland. Department of Culture, Arts and Leisure. [Online]. Available: <u>http://www.dcalni.gov</u>.<u>uk/attendance_at_live_sports_events_by_adults_in_northern_ireland.pdf</u> [Accessed: 8 October 2014].
- MARBURGER, D.R. (1997). Gate revenue sharing and luxury taxes in professional sports. *Contemporary economic policy*. 15: 114-123.
- MARX, S. and VAN DER WALT, A. (1989). *Marketing Management*. Juta and Co, Ltd: Cape Town, South Africa.
- MCGINNIS, L., CHUN, S. and MCQUILLAN, J. (2003). A Review of Gendered Consumption in Sport and Leisure. *Academy of Marketing Science Review*. 5: 1-27.
- MEHUS, I. (2005). Distinction through Sport Consumption: Spectators of Soccer, Basketball and Ski-jumping. *International Review for the Sociology of Sport*. 40: 321-333.
- MESSINIS, G. (1999). Habit formation and the theory of addiction. *Journal of Economic Surveys*. 13(4): 417-442.

- MILANO, M. and CHELLADURAI, P. (2011). Gross Domestic Sports Product: The Size of the Sports Industry in the United States. *Journal of Sports Management*. 25: 24-35.
- MILES, A. and SULLIVAN, A. (2010). Understanding the relationship between taste and value in culture and sport. [Online]. Available: <u>https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/77966/DCMS_taste_and_value_document.pdf</u> [Accessed 12 November 2013].
- MILLER, D. AND L. A. GORDON. (1975). Conceptual Levels and The Design of Accounting Information Systems. *Decision Sciences*. April 1975: 259-269.
- MILNE, G. R. and MCDONALD, M. A. (1999). *Sport marketing: Managing the exchange process*. Sudbury, MA: Jones and Bartlett Publishers.
- MONTGOMERY, S. and ROBINSON, M (2008). *Take me out to the opera: Are Sports and Arts Compliments?* Unpublished paper. Department of Economics. Mount Holyolk College. MA.
- MORGAN, B. (2013). *Cricket in South Africa*. [Online]. Available: <u>http://www. Southafrica</u>. <u>.info/about/sport/cricket.html</u> [Accessed 14 March 2013].
- MULLER, A. (2014). *Analysis: Who watches cricket in South Africa?* Daily Maverick. [Online]. Available: <u>http://www.dailymaverick.co.za/article/2014-05-22-analysis-who-watches-cricket-in-south-africa/#.VDU4uk-KDGg</u> [Accessed: 8 October 2014].
- MURRAY, E.J. (1964). Motivation and Emotion, Englewood Cliffs, NJ: Prentice Hall.
- NATIONAL DEVELOPMENT PLAN. (2014). *Transforming society and uniting the country*. [Online]. Available: <u>http://www.npconline.co.za/MediaLib/Downloads/Home/Tabs/</u><u>NDP%202030-CH15transforming%20society%20and%20uniting%20the%20country</u>.<u>pdf</u> [Accessed 5 March 2014].
- NIR, A. (2004). A behavioral model of consumption patterns: The effects of cognitive dissonance and conformity. Discussion paper No. 2004-48, Tilburg University.
- NIXON, R.D. (1992). Apartheid on the Run— The South African Sports Boycott. *Transition*. 58: 68-88.
- NJANJI, S. (2013). Tackling race issues in white-dominated cricket, rugby. Mail and Guardian. [Online]. Available: <u>http://mg.co.za/article/2013-10-29-tackling-quota-issues-in-white-dominated-cricket-rugby</u> [Accessed 11 February 2014].
- NOLL, R.G. (1974). Attendance and Price Setting in R.G. Noll *Govenernment and the Sports Business.* pp. 115-157. Washington, D.C.: The Brookings Institute.
- OLIVER, P. (2006). *Purposive Sampling*. The SAGE Dictionary of Social Research Methods. [Online]. Available: <u>http://srmo.sagepub.com/view/the-sage-dictionary-of-social-res</u> <u>earch-methods/n162.xml</u> [Accessed 11 March 2013].
- OLYMPIC.ORG. (2013). Sports and Disciplines. [Online]. Available: <u>http://www.olympic.org/sports</u> [Accessed 23 October 2013].

- PAWLOWSKI, T. and BREUER, C. (2011). The Demand for Sports and Recreational Services: Empirical Evidence from Germany. *European Sports Management Quarterly*. 11: 5-34.
- PERKINS, A. (2013). Exploring Motives for Popular Music Concert Attendance. [Online]. Available: <u>http://anzmac.org/conference/2012/papers/317ANZMACFINAL.pdf</u> [Accessed 12 November 2013].
- PETERSON, R.A. (1992). Understanding Audience Segmentation: From elite and mass to omnivore and univore. *Poetics*. 21 (4): 243-258.
- PETERSON, R.A. (2005). Problems in comparative research: The example of omnivourousness. *Poetics*. 33: 257-282.
- PETERSON, R.A., and KERN, R.M. (1996). Changing highbrow taste: from snob to omnivore. *American Sociological Review*. 61: 900-907.
- RAO, S.R. (2012). *Gross Domestic Product (GDP)*. [Online]. Available: <u>http://www.citeman.com/20365-gross-domestic-product-gdp.html</u> [Accessed 20 February 2013].
- RATNESHWAR, S., MICK, D.G. and HUFFMAN, C. (2000). The why of consumption: Contemporary perspectives of consumer motives, goals and desires. New York: Routledge.
- REID, J. (2012). *The digital disaster that we know nothing about. Daily Maverick*. [Online]. Available: <u>http://www.dailymaverick.co.za/opinionista/2012-12-03-the-digital-disaster</u> <u>-that-we-know-nothing-about/</u> [Accessed 7 March 2014].
- ROBERTS, K. (2012). The leisure of young people in a contemporary society. *ARBOR Ciecia*, 754: 327-337.
- ROKEACH, M. (1973). The Nature of Human Values. New York: The Free Press.
- ROSENBURG, M. (1979). Conceiving the Self. New York: Basic Books.
- ROSSMAN, G.H. and PETERSON, R.A. (2005). The instability of omnivorous cultural taste over time. In Philadelphia: Annual meetings of the American Sociological Association, Philadelphia, PA.
- RUTENBERG, H. (2003). *Consumer Behaviour*. [Online]. Available: <u>https://ujdigispace</u> <u>.uj.ac.za/bitstream/handle/10210/302/DChapter2.pdf?sequence=10</u> [Accessed: 12 August 2014].
- SABC SPORT. (2014). Sport. [Online]. Available: <u>http://www.sabc.co.za/wps/wcm/connect</u> /71fb25804525636d99da99a5ad025b24/SportSchedule+15+August+2014.pdf?MOD= <u>AJPERES&CACHEID=71fb25804525636d99da99a5ad025b24</u> [Accessed: 19 August 2014].
- SAHNEY, S. (2013). *Consumer Behavior*. [Online]. Available: <u>http://nptel.ac.in/courses/110</u> <u>105029/pdf%20sahany/Module,module-2-3.pdf</u> [Accessed: 3 June 2014].
- SAHO. (2014). The full citizenship of Coloured people is called for. [Online]. Available: <u>http://www.sahistory.org.za/dated-event/full-citizenship-coloured-people-called</u> [Accessed: 24 September 2014].
- SAMUELSON, L. (2004). Information-based relative consumption effects. *Econometrica*. 72: 93-118.
- SCHEETZ, T.K., DUBIN, R.A. and GARBARINO, E.C. (2014). A Modern Investigation of Status Consumption. [Online]. Available: <u>https://case.edu/artsci/dean /elf/documents</u> /scheetzreport.pdf [Accessed: 25 June 2014].
- SCHIFFMAN, L. G. and KANUK, L. L. (2007). *Consumer Behavior*, 9th ed. New Jersey, Prentice Hall.
- SCHMIDT, M.B. and BERRI, D. (2001). Competitive balance and attendance: The case of major league baseball. *Journal of Sports Economics*. 3: 204-228.
- SENTLETSE, D. (2010). In defense of conspicuous consumption. Mail and Guardian. 3 December. [Online]. Available: <u>http://www.thoughtleader.co.za/sentletsediakanyo/</u>2010/12/03/in-defence-of-conspicuous-consumption/ [Accessed 10 February 2014].
- SHAH, A. (2010). *Factors Affecting Consumer Behaviour*. [Online]. Available: <u>http://www.aipmm.com/html/newsletter/archives/000434.php</u> [Accessed 12 March 2013].
- SIEGFRIED, J.J. and EISENBERG, J.D. (1980). The demand for minor league baseball. *Atlantic Economic Journal*. 8: 59 69.
- SIMMONS, M.J. (2011). Fan-family conflict: An examination of inter-role conflict between sport fan and family roles. [Online]. Available: <u>http://digital.library.louisville.edu</u>/utils/getfile/collection/etd/id/2151/filename/4948.pdf [Accessed:22 October 2014].
- SIMMONS, R. (1996). The Demand for English League Football: A Club Level Analysis. *Applied Economics*. 28: 139-155.
- SIMMONS, R. (2006). The demand for spectator sports. In Andreff, W. and Szymanski, S. (Eds). *Handbook on the Economics of Sport*. Northampton: Edward Elgar Publishing Inc. pp 77-89.
- SIMON, H.A. (1997). Models of Bounded Rationality: Empirically Grounded Economic Reason. Cambridge, MA. MIT Press.
- SNOWBALL, J.D., JAMAL, M. and WILLIS, K.G. (2009). Cultural Consumption Patterns in South Africa: An investigation of the Theory of Cultural Omnivores. *Social Indicators Research*. 93 (3): 467-483.
- SOUTH AFRICAN WEB. (2013). Sport and Recreation in South Africa. [Online]. Available: <u>http://www.southafricaweb.co.za/page/sport-and-recreation-south-africa</u> [Accessed: 26 July 2013].
- SOUTHERTON, D. (2003). 'Squeezing Time': Allocating Practices, Coordinating Networks and Scheduling Society. *Time Society*. 12 (5): 5-25.

- SRSA. (2012). *Mission and Values*. [Online]. Available: <u>http://www.srsa.gov.za/ pebble.</u> <u>asp?relid=96</u> [Accessed 7 June 2012].
- STANDER, S. (2009). The Commodity. Political Economy. 25: 71-91.
- STATISTICS SOUTH AFRICA. (2012). *National Accounts*. [Online]. Available: <u>http://www.statssa.gov.za/publications/Report-04-05-07/Report-04-05-072010.pdf</u> [Accessed: 26 July 2013].
- SULLIVAN, O. and KATZ-GERRO, T. (2007). The Omnivore Thesis Revisited: Voracious Cultural Consumers. *European Sociological Review*. 23: 123-137.
- SUOMINEN, S. (2007). Are the spectators of performing arts and the spectators of movies the same? [Online]. Available: <u>http://www.jace.gr.jp/ACEI2012/usb_program/pdf/</u>7.2.2.pdf [Accessed 10 November 2013].
- SUPERSPORT. (2014). *Action Attractions*. [Online]. Available: <u>http://www.supersport.com/</u> <u>tv-guide</u> [Accessed: 19 August 2014].
- SUTTON, W.A., MCDONALD, M.A., MILNE, G.R. and CIMPERMAN, J. (1997). Creating and fostering fan identification in professional sports. *Sport Marketing Quarterly*. 6: 15-22.
- SWANSON, S.R., SCOTT, R., DAVIS, C. ZHAO, Y. (2008). Art for Arts' Sake? An Examination for Motives for Arts Performance Attendance. *Nonprofit and Voluntary Sector Quarterly*. 37: 300-323.
- SZYMANSKI, S. and KUYPERS, T. (1999). The economic design of sporting contests. Journal of Economic Literature. 41: 1137-1187.
- SZYMANSKY, S. (2003). The Economic Design of Sporting Contests. *Journal of Economic Literature*. 41 (4): 1137-1187.
- TAMARA, R. (2011). A Snapshot of Consumer Behaviour. [Online]. Available: <u>http://www.livemint.com/2011/02/17203310/A-snapshot-of-consumer-behavio.html</u> [Accessed: 5 July 2012].
- TAMPUBOLON, G. (2008). Revisiting omnivores in America circa 1990s: The exclusiveness of omnivores? *Poetics*. 36: 243-264.
- TANNER, J.F. and RAYMOND, M.A. (2012). Principles of Marketing. 2 ed. Flat World Knowledge. [Online]. Available: <u>http://2012books.lardbucket.org/books/marketing-principles-v2.0/s01-about-the-authors.html</u> [Accessed 22 May 2014].
- THE ECONOMIST. (2014). Equestrian eventing at the Olympics: Elitist of egalitarian? [Online]. Available: <u>http://www.economist.com/blogs/gametheory/2012/08/equestrian</u> <u>-eventing-olympics</u> [Accessed: 12 August 2014].
- THE WORLD YOUTH REPORT. (2003). *Expanding Opportunities for Young People and Communities*. [Online]. Available: <u>http://www.un.org/esa/socdev/unyin/documents/</u> <u>ch08.pdf</u> [Accessed: 14 October 2014].

- THRANE, C. (2001). Sport Spectatorship in Scandinavia: A Class Phenomenon? *International Review for the Sociology of Sport.* 36: 149–163.
- TOWLER, M. (2010). Rational Decision Making: An Introduction. New York: Wiley.
- TRAIL, G. T., ANDERSON, D. F., and FINK, J. S. (2000). A theoretical model of sport spectator consumption behavior. *International Journal of Sport Management*. 1: 154-180.
- TRAIL, G. T., ANDERSON, D. F. and FINK, J. S. (2005). Consumer satisfaction and identity theory: A model of sport spectator conative loyalty. *Sport Marketing Quarterly*, 14: 98-112.
- TRAIL, G. T. and JAMES, J. D. (2010). Theories of consumer behavior. In TRAIL, G.T. and JAMES, J.D. (Eds). Sport Consumer Behavior. Seattle, WA: Sport Consumer Research Consultants Internet Publishing Co. pp 33-63.
- TRAIL, G.T. and JAMES, J. D. (2001). The Motivation Scale for Sport Consumption: A comparison of psychometric properties with other sport motivation scales. *Journal of Sport Behavior*. 24: 108-127.
- TRAIL, G.T. and JAMES, J.D. (2011). Model of Spectator Sports Consumption. AMA Summer Educators' Conference Proceedings. 22: 1-7.
- TRAIL, G. T, ROBINSON, M., & KIM, Y. (2008). Sport consumer behavior: A test for group differences on structural constraints. *Sport Marketing Quarterly*. 17: 190-200.
- VAN EIJCK, K., and LEIVENS, J. (2008). Cultural omnivorousness as a combination of highbrow, pop and folk elements: The relation between taste patterns and attitudes concerning social integration. *Poetics*. 36: 217-242.
- VEBLEN, T. (1899). *The Theory of the Leisure Class: An Economic Study of Institutions*. New York: The Macmillan Company.
- VEBLEN, T. (1994). The Theory of the Leisure Class. New York: Dover Publications.
- VW STAFF. (2012). Who Has More Time: High or Low Income Earners? [Online]. Available: <u>http://www.valuewalk.com/2012/04/who-has-more-leisure-time-high-or-low-income-earners/</u> [Accessed 7 March 2014].
- WAKEFIELD, K.L. (1995). The Pervasive Effects of Social Influence on Sporting Event Attendance. *Journal of Sport and Social Issues*. 19 (4): 335-351.
- WALSH, R.G. (1986). Recreation Economic Decisions: Comparing Benefits and Costs. Venture Publishing Inc. State College. Minnesota.
- WARDE, A. (2006). Cultural capital and the place of sport. *Cultural Trends*. 15 (2–3): 107–123.
- WESTBROOK, R.A. and OLIVER, R.L. (1991). The Dimensionality of Consumption Emotion Patterns and Consumer Satisfaction. *Journal of Consumer Research*. 18: 84-91.

- WHITE, P. and WILSON, B. (1999). Distinction in the Stands: An investigation of Bourdieu's "Habitus" Socioeconomic Status and Sport Spectatorship in Canada. *International Review for the Sociology of Sport*. 34: 245-264.
- WHITNEY, J. (1988). Winning Games versus Winning Championships: The Economics of Fan Interest and Team Performance. *Economic Inquiry*. 26: 703-724.
- WIDDOP, P. (2010). Sport, Capital and Consumption. ISC Working Paper. [Online]. Available:<u>http://www.humanities.manchester.ac.uk/socialchange/publications/working</u>/documents/SportCapitalandConsumption.pdf [Accessed: 16 October 2010].
- WIDDOP, P. and CUTTS, D. (2013). Social Stratification and Sports' Participation in England. *Leisure Sciences*. 35: 107-128.
- WILENSKY, H. L. (1964). Mass Society and Mass Culture: Interdependence or Independence? *American Sociological Review*. 29 (2): 173-197.
- WILLIAMS, M.F. (2013). *Consumer Behavior and Utility-maximization*. [Online]. Available: <u>http://getyourecon.com/prin-gen/prin-utility.pdf</u> [Accessed: 3 June 2014].
- WILSON, T.C. (2002). The Paradox of Social Class and Sports Involvement: The Roles of Cultural and Economic Capital. *International Review for the Sociology of Sport.* 37: 5–16.
- WILSON, P. and SIM, W.B. (1995). The demand for semi-pro league football in Malaysia 1989-91: A panal-data approach. *Applied Economics*. 27: 131-138.
- ZAICHKOWSKY, J.L. (1991). Consumer Behaviour: Yesterday, today and tomorrow. *Business Horizons*. 34 (3): 51-58.
- ZHENG, Y. (2014). Analysis of the Influence of Cultural and Social Psychological Factors on Sports Consumption. 2014 International Conference on Education Reform and Modern Management (ERMM-14). Atlantis Press, 2014

APPENDIX A: QUESTIONNAIRE

Appendix A is a copy of the questionnaire that was developed and administered to ascertain motives for consumption behaviour and possible responses revealing tendencies of sporting omnivores and univores for various definitions (referred to throughout Chapter Three). The first page was given to each participant to read prior to the commencement of their interview. The subsequent pages of this appendix are the questionnaire itself. The text stated in square brackets are instructions to the interviewers, prompting them to either do or say specific things at different stages of the interview.



Dear Participant

The purpose of this research is to investigate consumption behaviour of cricket in South Africa. The research aims to identify the motives for match attendance with the ultimate goal of adding value to the game from a developmental and informed marketing perspective. The research is commissioned by both Rhodes University and the Chevrolet Warriors, represented by Dave Emslie and Leigh Deyzel.

I kindly request that you share your motives for match attendance by completing the questionnaire. The questionnaire will take approximately 10 minutes to complete. There are no correct or incorrect answers and your opinion is highly valued. You may withdraw from completing the survey at any time.

Please note that your participation in this survey is voluntary and anonymous. All information provided will be used for research purposes only. Privacy and confidentiality will be respected. The data obtained will be housed in private storage upon completion of the analysis. The research study has been approved by the Departmental Research Ethics Committee as well as by the Faculty of Commerce Higher Degrees Committee.

Thank you for taking the time to complete the survey and help us ensure that the legacy of the game be maintained and developed.

Ms Kelcey Brock and Prof GCG Fraser Department of Economics and Economic History Rhodes University Grahamstown [The first part of the survey is to find out what your opinions are on watching cricket and other sport. I will read out a statement, and you will then please tell me if you strongly agree, agree, disagree or strongly disagree with it. OK? If you would like me to repeat a statement, or remind you of the categories as we go along, just let me know.]

STRONGLY AGREE (SA), AGREE (A), DISAGREE (D),	SA	A	D	SD
STRONGLY DISAGREE (SD)				
1. I consider myself a fan of South African cricket				
2. Cricket is the <u>only</u> sport I choose to watch live (in attendance)				
3. I play/ have played cricket competitively before (At school/provincially/internationally)				
4. I consider myself a fan of cricket as a whole and not just of a particular team				
5. I attend cricket matches mainly because my friends or family are attending				
6. From time to time I go to other professional sports matches				
7. I prefer watching sport live as opposed to on TV.				
8. The social experience (e.g. socialising at the game, dancing, beer tent, competitions etc) is just as important as the cricket match				
9. I would consider myself as a sports fan rather than just a cricket fan				
10. I attend cricket matches because I am specifically a fan of one/some of the players				
11. I enjoy attending the before and after game entertainment				
12. I would like to attend more live cricket matches				
13. I prefer close competition games where the outcome is uncertain				
14. I attend the cricket matches as a way of meeting new people				
15. I watch sports channels on TV				
16. Cricket is my favourite sport to watch live				
17. I feel the matches are advertised well				
18. Watching cricket allows me to feel part of the community				
19. Watching South African cricket makes me feel proudly South African				
20. I believe that the ticket price is worth the experience				
21. I believe playing sport at school developed my love for sport				
22. If a certain player/some players was/were not playing I would not attend				

23. Some of my family members play/have played cricket competitively		
24. I watch both professional and amateur cricket		
25. Cricket is the only sport I choose to watch on TV		

[Please ask the participants the following open-ended questions]

- a) Approximately how many live cricket matches do you watch in a 12 month period?
- b) Do you watch cricket on TV?

- c) During the cricket season, about how many games would you say you watch on TV per month?
- d) What other sports do you go and watch live?

Golf	
Tennis	
Soccer	
Rugby	
Horse Racing	

Other (Please specify).....

e) What sport do you watch on TV?

Golf	
Tennis	
Soccer	
Rugby	
Horse Racing	

Other (Please Specify).....

f) What influences whether you watch a cricket match live or on TV?

[Thanks very much for your help so far. The last part of the questionnaire is about you. While we support non-discrimination, this information will be useful in analysing the results. Your name will not be attached to the questionnaire at all, and you can refuse to answer a particular question if you like. OK?]

g) How old are you?

h)	Please record this information but DO NOT ask the participants
Race: .	
Gender	

[Provide participants with the table and ask them to please state what their highest level of education is – record their response on the table below]

i) Highest level of education:

Primary education	rimary education Secondary education but no matric		Tertiary education	Other (please specify)	

j) Personal Monthly Net (after tax) Income Bracket

[Provide interviewee with income bracket table and ask them to please state which bracket best describes their personal monthly income after tax – record their response on the table below]

R0 –	R10 000 –	R20 000 –	R30 000 –	R 50 000 –	R60 000 –	R70 000 +
R10 000	R20 000	R30 000	R40 000	R60 000	R70 000	

[Thank you for taking the time to complete the questionnaire.]

Interviewer's initials:

Interviewer, please state whether you believe that the data collected is reliable (e.g. the participant understood all questions asked, the participant was not under the influence of alcohol or not affected by any other factors that may skew their responses): Y/N

APPENDIX B:





Source: Trail and James (2011)

APPENDIX C:

POST-ESTIMATION F-TEST RESULTS OF SIGNIFICANCE BETWEEN EXPLANATORY VARIABLES (OTHER THAN BETWEEN THE SPECIFIED BASE GROUPS AS TESTED FOR IN THE INDIVIDUAL PROBIT MODELS).

Test for significant difference between:	Definition 1 ⁴²		Definition 2 ⁴³		Definition 3 ⁴⁴		Definition 4 ⁴⁵		Conventional Live		Conventional TV	
	χ^2	$P\!\!>\!\chi^2$	χ^2	$P\!\!>\!\chi^2$	χ^2	$P > \chi^2$	χ^2	$P\!\!>\!\chi^2$	χ^2	$P\!\!>\!\chi^2$	χ^2	$P\!\!>\!\chi^2$
Race:												
White and Coloured	1.78	0.1823	3.58	0.0585	0.19	0.6669	5.81	0.0159	0.07	0.7970	1.35	0.2459
White and Indian	6.80	0.0091	1.81	0.1783	1.85	0.1738	9.69	0.0019	0.32	0.5739	0.90	0.3429
Coloured and Indian	3.47	0.0624	0.01	0.9080	1.21	0.2722	1.74	0.1871	0.14	0.7120	0.02	0.8800
Age:												
21-30 and 31-40	0.01	0.9414	0.28	0.5969	0.17	0.6825	0.53	0.4665	1.70	0.1927	0.06	0.8040
21-30 and 41-50	1.41	0.2349	0.06	0.8139	0.03	0.8673	2.00	0.1572	1.47	0.2258	0.11	0.7426
21-30 and 51+	1.85	0.1733	0.39	0.5332	0.47	0.4911	3.99	0.0457	1.28	0.2572	0.86	0.3536
31-40 and 41-50	0.93	0.3339	0.45	0.5002	0.04	0.8396	0.31	0.5764	0.05	0.8273	0.26	0.6084
31-40 and 51+	1.22	0.2688	0.01	0.9261	0.77	0.3801	1.48	0.2240	0.07	0.7970	1.06	0.3041
41-50 and 51+	0.01	0.9385	0.63	0.4291	0.50	0.4797	0.48	0.4893	0.00	0.9662	0.07	0.7966
Education:												
Secondary (no matric) and matric	0.18	0.6738	5.41	0.0200	0.00	0.9531	2.40	0.1216	8.66	0.0032	2.49	0.1147
Secondary (no matric) and post- secondary	1.09	0.2967	2.21	0.1372	0.10	0.7525	8.19	0.0042	5.39	0.0203	0.53	0.4657
Matric and post-secondary	1.00	0.3173	0.97	0.3236	0.41	0.5236	4.16	0.0318	0.24	0.6262	0.85	0.3559
Income:												
R20001- R40 001	0.65	0.4210	0.11	0.7357	0.62	0.4311	0.83	0.3636	0.25	0.6148		

Table C1: Post-estimation F-test results showing significance between explanatory variables

Worth noting is that 'Income' could not be included in the regression for the Conventional Definition-TV. This is because only 12 individuals were classified as univores, and after a cross-tabulation, it became evident that of these 12 individuals, none fell into the top two income categories; hence, no comparison groups were available for the omnivore status.

⁴² For Definition 1 participants were classified as univores if they watched only professional cricket; or omnivores if they watched professional and amateur cricket.

⁴³ For Definition 2 participants were classified as univores if they watched only cricket live; or omnivores if they watched cricket and other sport live.

⁴⁴ For Definition 3 participants were classified as univores if they watched only cricket on television; or omnivores if they watched cricket and other sport on television.

⁴⁵ For Definition 4 participants were classified as univores if they considered themselves to be only cricket fans; or omnivores if they considered themselves general sports fans.